PROJECT DRAWDOWN.



DRAWDOWN LIFT

CLIMATE—POVERTY CONNECTIONS: OPPORTUNITIES FOR SYNERGISTIC SOLUTIONS AT THE INTERSECTION OF PLANETARY AND HUMAN WELL-BEING

By Yusuf Jameel, Carissa M. Patrone, Kristen P. Patterson, and Paul C. West

Copyright info

© 2022 by Project Drawdown

Suggested citation

Jameel, Y., Patrone, C. M., Patterson, K. P., & West, P. C. (2022). Climate–poverty connections: Opportunities for synergistic solutions at the intersection of planetary and human well-being. Project Drawdown. https://doi.org/10.55789/y2c0k2p2

FOREWORD

Climate change is one of the most important development challenges of our time. It intensifies the effects of poverty, inequality, population growth, rapid urbanization, and environmental degradation, and disrupts national economies and their long-term growth potential. No country is immune from the impacts of climate change. Reducing emissions and becoming more resilient are vital, and urgently require countries to make major social, economic, and technological changes.

This report shows that climate mitigation and socioeconomic development can be achieved simultaneously. It summarizes evidence of the human well-being co-benefits of proven climate mitigation solutions in the context of rural communities in sub-Saharan Africa and South Asia. The report complements global initiatives such as the Climate Action Pathways, a set of comprehensive road maps to fully implementing the Paris Agreement across all sectors.

The authors make a powerful case for strengthening the evidence of impact of climate solutions on human well-being. They also point out that the efficacy of several solutions that mitigate greenhouse gas emissions and improve human well-being either has not been tested in realworld settings or is rare. Thus, it would be beneficial to evaluate the impact of such climate solutions. The report also underscores the need to assess their unintended consequences and differential impacts by important dimensions such as gender and income.

Transformation of the global economy to net-zero emissions by 2050 would require significant expansion of annual investments. The good news is that developing countries do not need to develop first in a carbon-intensive pathway and then clean up and decarbonize. This report can serve as an important guide for policymakers, the private sector, donors, and decision makers at national and global levels as they plan future investments in programs and research that alleviate both poverty and carbon emissions.

The task of bending the emissions curve and improving livelihoods is extremely urgent. The gravity and scale of the challenge requires countries to learn quickly from each other, adapt to their own special circumstances, and be audacious in implementing effective climate policies. Properly designed and implemented policies for low-carbon and resilient growth can also help address poverty and inequality, enabling millions to live safer, more prosperous, more inclusive, and more sustainable lives.

This report presents the benefits of climate mitigation for human well-being in a scientifically coherent manner. We count it a great privilege to introduce it at this crucial moment of human history.

Ademola Braimoh, World Bank, and Monica Jain, 3ie

EXECUTIVE SUMMARY

Addressing climate change and improving the well-being of people experiencing extreme poverty—two grand challenges of the 21st century—must occur simultaneously and, ideally, synergistically. These two inextricably intertwined issues will shape the well-being of humanity for the rest of the century and beyond. People living in extreme poverty are more vulnerable to natural disasters and commonly live in regions that have been and are projected to be most impacted by climate change. Unfortunately, the benefits of addressing climate change in ways that simultaneously improve the human well-being (HWB) of people living in extreme poverty are commonly overlooked—despite their promise to yield substantial socioeconomic, health, equity, ecological, and biodiversity gains.

This report highlights evidence of the co-benefits to HWB of proven climate mitigation solutions. It showcases areas of greatest need for funders, policymakers, nongovernmental organizations (NGOs), and other decision-makers to act and drive approaches to meet climate, development, and HWB needs while boosting prosperity for rural communities, especially in sub-Saharan Africa (SSA) and South Asia. The world should not have to choose among sustainable development, poverty eradication, climate mitigation, and climate adaptation; win-win solutions are at hand.

We summarize evidence of the HWB co-benefits of 28 climate mitigation solutions previously analyzed by Project Drawdown. To explore linkages among these solutions and their potential to improve lives, the report builds upon the Doughnut Economics model to develop a Drawdown Lift Human Well-Being Index (HWI), which serves as a framework for assessing 12 health, socioeconomic, and societal dimensions of HWB and advances evidence of the nexus of climate mitigation solutions, poverty alleviation, and HWB.



Women and children in low- and middle-income countries are disproportionately harmed by climate change. Through collective and urgent action, we can shape a healthy and sustainable planet for future generations. *Credit: Ollivier Girard for Center for International Forestry Research (CIFOR)*

Investments in low-carbon development pathways, as well as climate adaptation, must prioritize countries that are first and worst impacted by climate change—particularly low- and middle-income countries (LMICs). The costs are high, yet the costs of inaction are far higher—and leaves those least responsible for the climate crisis without the resources for a better, more resilient future as their economies bend toward prosperity.

Responses to climate change must also directly address rural communities' economic, nutritional, and energy needs—particularly those of women. Many groups of climate mitigation solutions, including those that focus on *Improving Agriculture and Agroforestry, Protecting and Restoring Ecosystems, Adopting Clean Cooking, Providing Clean Electricity,* and *Fostering Equality,* depend on women's labor and shape women's livelihoods. Climate interventions can and should contribute to improved education, health, and gender equality.

The world should not have to choose among human development, poverty eradication, climate mitigation, and climate adaptation; win-win solutions are at hand.

The opportunity to address climate change and improve HWB, particularly in LMICs, is enormous. Implementing the climate solutions presented in this report could make a dramatic difference for boosting HWB—especially when the solutions are implemented synergistically with cross-sectoral collaboration. This report raises awareness of the socioeconomic co-benefits of climate solutions, promotes cross-sectoral alliances to mobilize ambitious solutions, and seeks to channel funding for synergistic solutions from public, philanthropic, and private sectors.

We find that several actions can contribute to mobilizing win-win solutions to climate and poverty. Key recommendations include:

- Implement the 28 climate solutions highlighted in the report in synergistic ways to maximize co-benefits for boosting human well-being.
- Listen to, engage with, and work alongside diverse stakeholders—from local practitioners to senior policymakers—to build capacity and plan for synergistic climate-development actions.
- Include HWB co-benefits in multisector climate finance.
- Accelerate the development of innovative finance for climate solutions that include HWB co-benefits.
- Assemble additional evidence of HWB co-benefits to further drive a more holistic approach to climate solutions and poverty alleviation.

This report serves as a guide for decision-makers, policymakers, and donors working to advance low-carbon pathways to prosperity for under-resourced and socially disadvantaged groups in rural communities in LMICs. It seeks to support leaders and stakeholders in achieving the dual goals of alleviating poverty while limiting increases in global temperatures.

CONTENTS

| FOI | REW | ORD | Ш |
|-----|------|---|------|
| EXE | ECU | TIVE SUMMARY | IV |
| LIS | t of | FIGURES, FEATURE BOXES, AND TABLES | VIII |
| ACI | RON | YMS AND ABBREVIATIONS | IX |
| ACI | KNC | WLEDGMENTS | Х |
| 1 | INT | RODUCTION | 1 |
| | 1.1 | Addressing poverty and climate change: dual challenges for the 21st century | |
| | 1.2 | Climate impacts on vulnerable populations | 3 |
| | 1.3 | Climate-smart development, the SDGs, and the Paris Agreement | 6 |
| | 1.4 | Financing climate action | 9 |
| | 1.5 | Report road map | 11 |
| 2 | | DJECT DRAWDOWN CLIMATE SOLUTIONS AND -BENEFITS FOR HUMAN WELL-BEING | 12 |
| | 2.1 | The Drawdown Framework for Climate Solutions | 12 |
| | 2.2 | Climate solutions co-benefits for poverty alleviation and well-being | 13 |
| 3 | FR/ | AMEWORK FOR EVALUATING HUMAN WELL-BEING | 15 |
| | 3.1 | Introduction | 15 |
| | 3.2 | Drawdown Lift Human Well-Being Index | 16 |
| | 3.3 | Global patterns of human well-being | 18 |
| 4 | LIN | KING CLIMATE SOLUTIONS AND HUMAN WELL-BEING | 20 |
| | 4.1 | Climate solutions for improving agriculture and agroforestry | 22 |
| | 4.2 | Climate solutions for protecting and restoring ecosystems | 28 |
| | 4.3 | Climate solutions for adopting clean cooking | 35 |
| | 4.4 | Climate solutions for providing clean electricity | 40 |
| | 4.5 | Climate solutions for fostering equality | 49 |
| | 4.6 | Additional indirect co-benefits and ripple effects on human well-being dimensions of Project Drawdown climate solutions | 56 |
| | 4.7 | Summary and conclusion | 60 |
| 5 | МО | BILIZING AROUND THE CLIMATE | |
| | AN | D POVERTY NEXUS: CRITICAL ACTIONS | 62 |

| APPENDIX A Descriptions of the 28 Project Drawdown solutions | 68 |
|---|----|
| APPENDIX B Summary of common human well-being indices and description of indicators used in this report | 70 |
| APPENDIX C Calculating the Drawdown Lift Human Well-Being Index | 73 |
| REFERENCES | 76 |

LIST OF FIGURES, FEATURE BOXES, AND TABLES

Figure 1.1 Number of people living in extreme poverty

Figure 1.2 Poverty and vulnerability to climate change

Feature Box | Hariyo Ban, Nepal: Centering climate mitigation, adaptation, and livelihoods

Figure 1.3 Current finance and future investment needed to limit global temperatures to 1.5°C above preindustrial levels

Figure 2.1 The Drawdown Framework for Climate Solutions

Figure 2.2 Project Drawdown solutions with human well-being co-benefits

Figure 3.1 Doughnut Economics framework

Figure 3.2 Drawdown Lift Human Well-Being Index dimensions and related SDGs

Figure 3.3 Drawdown Lift Human Well-Being Index

 Table 4.1 Climate solutions groups and human well-being dimensions

Figure 4.1 Human well-being benefits of climate solutions for Improving Agriculture and Agroforestry

Feature Box | Climate-Smart Village project, India: Improving food security, income, and education

Figure 4.2 Securing land tenure for women leads to

Figure 4.3 Human well-being benefits of climate solutions for Protecting and Restoring Ecosystems

Feature Box I Gorongosa National Park, Mozambique: Lifting people out of poverty through conservation

Feature Box | Time spent by women and girls every day collecting fuel in several sub-Saharan African countries **Figure 4.4** Human well-being benefits of climate solutions for Adopting Clean Cooking

Figure 4.5 Human well-being levels in Food, Water and Sanitation, and Health for countries with low access to electricity

Figure 4.6 Competitiveness of solar- and diesel-based mini-grids

Figure 4.7 Human well-being benefits of climate solutions for Providing Clean Electricity

Feature Box | Solar Sister: Improving gender equality, income, health, and education by reducing energy poverty

Figure 4.8 Challenges faced by women and children

Figure 4.9 Human well-being benefits of climate solutions for Fostering Equality

Feature Box | Tuungane Project, Tanzania: Creating healthier and more resilient families, fisheries, and forests

Figure 5.1 Climate solutions' co-benefits for human well-being

Figure 5.2 Key actions to foster synergistic solutions

Table B.1Summary of commonlyused socioeconomic and humanwell-being indices

Table B.2Brief description and data sourcesfor the indicators included in the DrawdownLift Human Well-Being Index

Figure C.1 Drawdown Lift Human Well-Being Index provides an indication of relative human well-being for A: Income and Work, B: Water and Sanitation, C: Food, D: Health, E: Gender Equality, and F: Energy

Figure C.2 Drawdown Lift Human Well-Being Index provides an indication of relative national human well-being for A: Education, B: Networks, C: Housing, D: Peace and Justice, E: Political Voice, and F: Social Equity

ACRONYMS AND ABBREVIATIONS

| CO ₂ | Carbon Dioxide | NGO | Non-governmental Organization |
|---------------------|--------------------------------------|--------|---|
| CO_2 | Carbon Dioxide | NGO | Non-governmental Organization |
| CO ₂ -eq | Carbon Dioxide Equivalent | NTFP | Non-timber Forest Product |
| DE | Doughnut Economics | N_2O | Nitrous Oxide |
| FAO | Food and Agriculture Organization | OECD | Organisation for Economic Co-operation and Development |
| GBV | Gender-Based Violence | SDGs | Sustainable Development Goals |
| GDP | Gross Domestic Product | SPI | Social Progress Index |
| GHG | Greenhouse Gas | SPV | Solar Photovoltaics |
| Gt | Gigatons | SSA | Sub-Saharan Africa |
| HAP | Household Air Pollution | UN | United Nations |
| HWB | Human Well-Being | UNEP | United Nations Environment |
| HWI | Drawdown Lift Human | | Programme |
| | Well-Being Index | UNFCCC | United Nations Framework |
| IRENA | International Renewable Energy | | Convention on Climate Change |
| | Agency | USAID | United States Agency for |
| IPCC | Intergovernmental Panel on | | International Development |
| | Climate Change | WASH | Water, Sanitation, and Hygiene |
| LMICs | Low- and Middle-Income Countries | WHO | World Health Organization |
| MPI | Multidimensional Poverty Index | | |

ACKNOWLEDGMENTS

Thank you to external reviewers Debbie Aung Din (Proximity Designs), Mohamed Imam Bakarr (Global Environment Facility), Jill Baumgartner (McGill University), Ademola Braimoh (World Bank), Jenny Frankel-Reed (Bill & Melinda Gates Foundation), Rajib Ghosal (Save the Children), Monica Jain (3ie), Christina Kwauk (Education Consultant), Clive Mutunga (USAID),ⁱ Yolande Wright (Save the Children), and Hisham Zerriffi (University of British Columbia) for their valuable insights, which made the report much stronger. We would also like to thank Ademola and Monica for crafting a thoughtful foreword. Thanks also to Margaret E. Greene for her review with an eye toward gender. Any errors in this report are the responsibility of the co-authors. We appreciate colleagues from Gorongosa National Park, The Nature Conservancy, and WWF for their review of the case studies and sharing photos. We recognize and thank the inaugural Drawdown Lift Advisory Council for their helpful feedback in the early stages of the report framing as well as key inputs in the final stages. We are grateful to Jonathan Foley (Project Drawdown) for his guidance and to our collaborative Project Drawdown colleagues. We thank King Philanthropies for their generous support of Drawdown Lift.

Finally, we acknowledge and deeply appreciate the many friends and colleagues in countries around the world—especially those in Brazil, Burkina Faso, Colombia, Ghana, India, Kenya, Madagascar, Nicaragua, Niger, Palau, South Africa, Tanzania, Uganda, Zambia, and Zimbabwe—whose knowledge and wisdom have deeply enriched our personal and professional lives and continue to inform our work.

i Under a Global Health Support Initiative contract

1.1 Addressing poverty and climate change: dual challenges for the 21st century

Eradicating extreme poverty and improving HWBⁱⁱ are critical needs for humanity. In recent decades, LMICs have seen rapid economic growth and a steady decline in extreme poverty, alongside many improvements in HWB. Until 2020, extreme absolute poverty—defined as the condition of living on less than US\$1.90 per day—had been declining for almost 25 years— alongside improvements in life expectancy and adult literacy, reduced child and maternal mortality, and lowered malnourishment.³ However, much more work remains, as progress has been geographically uneven. It is starkest in SSA, where 40 percent of the population still lives in extreme poverty and over half of the population is multidimensionally poor;ⁱⁱⁱ indeed, two-thirds of those experiencing multidimensional poverty live in Africa (Figure 1.1).⁴



COVID-19 and vaccine inequality have had major impacts on people living in low- and middle-income countries, many of whom are also burdened by the effects of climate change. *Credit: Matteo Guedia | Alamy Stock Photo*

ii HWB is multidimensional. It is defined as people's ability to access fundamental social, cultural, economic, and natural/ environmental resources that are critical for sustaining a decent living standard and living a life they value (see also^{1,2}).

iii Multidimensional poverty goes beyond income to include deprivations around health, education, living standards, quality of work, safety, and environmental health.



Figure 1.1 — Number of people living in extreme poverty

The number of people living in extreme poverty (defined as < \$1.90 per day) has decreased across all regions except sub-Saharan Africa since 1990. *Data source: Poverty and Shared Prosperity 2020 PovcalNet*

Eighteen of the 20 economies with the world's highest poverty rates are in SSA.⁴ What is most startling is that poverty rates in the region have not budged for over a generation (since 1990) while, during the same period, poverty levels dropped dramatically in countries in the Pacific and East Asia (and pre-pandemic, more recently in South Asia).⁴ Fragility, conflict, and, increasingly, climate change, all contribute to the continued high rate of poverty in SSA.^{4,5} Even where there was rapid economic growth—in countries such as Kenya and Ethiopia, which saw a decrease in the share of their populations living on less than US\$1.90 per day from 2005 to 2015⁴—it was countered by rapid population growth, so the number of people experiencing poverty actually increased during that period. Lifting people out of the poverty trap and building long-lasting resilience requires a complementary approach that includes increasing assets, building capacity, and addressing climate risks.⁶

Effective policy to reduce poverty requires multi-dimensional thinking. Responses to only one among the financial, human, natural, physical and social capital that could help a family lift themselves out of poverty might miss the root causes of persistent, extreme poverty in a complex, realworld setting. Simultaneously addressing assets, capacities and the risk of [climate] shocks will have the best chance of creating lasting change for poor and vulnerable households.

CHRISTOPHER BARRETT, MICHAEL CARTER, AND JEAN-PAUL CHAVAS⁶

Unfortunately, COVID-19 has set back years of progress and further deepened the poverty trap. Prior to the pandemic, roughly 65–75 million people globally were living in extreme poverty;⁷ the latest projections suggest that the number of people experiencing poverty could nearly double,⁸ with South Asia and SSA the most affected. One study estimates that 96 million more people will be pushed into poverty.⁸

Recent events also exacerbated the poverty gender gap. Pandemic lockdowns led to increased violence against women, while school closures and job losses increased rates of child, early, and forced marriage and unions. All of these exert dire effects on women's and adolescents' mobility and well-being.⁹ In 2017, approximately 18 percent of children lived in poverty, compared with 8 percent of adults.¹⁰ Women ages 25–34 are particularly vulnerable to poverty due largely to persistent challenges around gender inequality.¹¹

As inequities deepen on our warming planet, the pandemic has made the economic, health, and planetary costs of inaction starker than ever.^{12, 13} Synergistically addressing climate and poverty is possible; a low-carbon development path helps reduce inequality while also boosting absolute well-being.

1.2 Climate impacts on vulnerable populations

Under-resourced households experience disproportionate impacts of climate change. Thus, prioritizing climate mitigation solutions that also alleviate poverty, improve HWB, enhance climate adaptation, and facilitate low-carbon pathways to prosperity will doubly benefit such households.

Climate vulnerability is also closely correlated with gender inequality. Improving gender equality lifts up vulnerable populations while generating multiplier effects for reducing climate vulnerability and achieving global development and climate goals.

Impacts of climate change vary enormously across countries and social groups. Some LMICs, particularly those in SSA, face a triple nexus of gender inequality, state fragility, and climate vulnerability, affecting both HWB and ecosystem health.¹⁴ Women, people experiencing poverty, Indigenous peoples, ethnic minorities, persons with disabilities, and those who are socially disadvantaged are often the most vulnerable to climate hazards because their exclusion from economic, educational, and social systems limits their access to vital resources and opportunities, leading to less capacity to adapt or build resilience.¹⁴ Women and girls in LMICs are disproportionately impacted by the effects of climate change,^{15, 16} including extreme weather and natural disasters.^{17, 18}

In many LMICs, climate change already threatens decades of progress across multiple areas of development. Changing climatic patterns oblige millions of vulnerable people to face more frequent and extreme weather events, climate variability and hazards, and threats to food security, water security, health, livelihoods, and cultural integrity. Rural populations and Indigenous peoples in LMICs—particularly in communities that depend on rain-fed and smallholder agriculture—rely extensively on natural ecosystems for their livelihoods, while many countries extract and export vital natural resources to generate economic growth. Climate change could shrink the gross domestic products (GDPs) of countries like Bangladesh, Madagascar, and Nigeria by more than 10 percent¹⁹ and could push an additional 100 million people into poverty by 2030.²⁰

Those experiencing poverty do not feel the effects of climate change equally. The most vulnerable and the least prepared countries also experience the highest levels of poverty (Figure 1.2).



Figure 1.2 — Poverty and vulnerability to climate change

Countries with high rates of poverty (percent of people earning < US\$1.90 per day) tend to be more vulnerable to, and less ready for, climate change than those with lower poverty rates. *Data source: World Bank 2019, ND-GAIN 2019*^{tv}

Just as people experience poverty in gender-specific ways, the adverse effects of climate change are also strongly gendered. Women and men have vastly divergent access to financial resources, natural resources, land tenure, education, health care, and decision-making power.²² Those differences increase women's vulnerability and can limit their ability to adapt to climate change.²³ Indeed, two-thirds of the more than 835 million people who experience multidimensional poverty reside in households in which no women have completed at least six years of schooling.²⁴ Climate-driven hazardous events disproportionately harm women and exacerbate gender disparities in education, reproductive health, and socioeconomic status.²⁵

iv The Notre Dame Global Adaptation Initiative (ND-GAIN) enhances understanding of climate adaptation through knowledge, products, and services that inform public and private actions and investments in vulnerable communities. The Country Index uses two decades of data across 45 indicators to rank 181 countries annually based on their level of vulnerability and their readiness to successfully implement adaptation solutions. Countries with a low ND-GAIN score are more vulnerable to climate change and less prepared to address it.²¹



Tackling climate change and achieving gender equality go hand in hand. Climate action must focus on the most vulnerable people and those who have been excluded from necessary and vital resources to better support themselves and their families. *Credit: Valerie Caldas | United States Agency for International Development Suaahara project*

It is crucial that women are represented as key stakeholders in all spheres of life and in climate negotiations and action; full, equal, and meaningful participation and leadership of women must take place in all aspects of United Nations Framework Convention on Climate Change (UNFCCC) processes and in local and national climate policy work to achieve shared climate goals²⁶ and advance more responsive, accountable governments.²⁷ According to a 2016 review of 162 submitted intended Nationally Determined Contributions, only 40 percent mention gender or women in their national priorities for reducing greenhouse gas (GHG) emissions.²⁸

Good governance and democracy require inclusive leadership and representation. While men and women are equally responsible for achieving gender equality, a larger number of women in office can influence gender responsive public policies and institutional practices. Women have a right to be equally represented and consulted in decision-making.

REPORT OF THE SECRETARY GENERAL, 65TH COMMISSION ON STATUS OF WOMEN.²⁷

The impact of climate change on children—especially those in LMICs—and future generations is particularly grim.²⁹ Babies born in 2020 in SSA are projected to experience around six times more extreme climate events over their lifetimes than those born in 1960.³⁰ Children experience adverse consequences in terms of nutrition, education, and health-care access after natural hazards and climate shocks.³¹ Globally, more than 90 percent of children are exposed to at least two climate and environmental hazards over the course of their childhood.³² Increases in temperature reduce food diversity and can undermine development interventions and impact children's health in regions experiencing long droughts and high temperatures.³³ All of these impacts can perpetuate intergenerational poverty.^{32, 34}

1.3 Climate-smart development, the SDGs, and the Paris Agreement

The immensity of these challenges requires that poverty and climate change be addressed simultaneously. To do this, the world needs to find pathways for development that lift people out of extreme poverty without further exacerbating the risks of climate change.



Implementing climate solutions such as Distributed Solar Photovoltaics, Farm Irrigation Efficiency, and Sustainable Intensification for Smallholders together can provide win-wins for climate, human well-being, and sustainable development. Credit: Joerg Boethling | Alamy Stock Photo

In 2015, all 193 United Nations (UN) member countries adopted the integrated 2030 Agenda for Sustainable Development with crosscutting objectives to improve HWB, act on climate change, end hunger and poverty, realize human rights, ensure gender equality, and enable communities to thrive peacefully while protecting the planet from degradation. Similarly, the UNFCCC's 2015 Paris Agreement aims to "promote the mitigation of GHG emissions while fostering sustainable development."³⁵ One principle of the UN Sustainable Development Goals (SDGs) is to invest in areas that yield multiple benefits across the goals and avoid unnecessary trade-offs, such as development strategies that undermine climate change mitigation and adaptation, and vice versa.³⁶ Climate-resilient development that intentionally reduces existing inequalities and facilitates a just transition to low-carbon economies in LMICs (while preventing and minimizing adverse outcomes) will yield both sustainable economic and human development benefits.^{37, 38}

As the world strives to achieve all 17 SDGs and keep global temperature increases well below 2°C, implementing climate solutions that facilitate a just transition to a low-carbon economy while fostering sustainable economic and human development should be a top priority.³⁹ Achieving the SDGs and upholding the aims of the Paris Agreement in a mutually reinforcing way is possible.^{39,40} Global leaders and policymakers have an opportunity and an obligation to advance climate solutions that foster a low-carbon transition based on equitable and sustainable social, economic, and environmental development across all stages of policymaking.³⁷ The needs are daunting. For example, half of Africa's population does not yet have access to electricity. The opportunity to develop clean energy sources at national and regional levels as Africa powers up is vast, yet the costs to support economic development and build low-carbon infrastructure will also be great.^{41,42}

LMICs are historically the lowest GHG emitters, yet they are disproportionately impacted by climate change; countries in the Global South (including Latin America, Africa, the Middle East, and Asia) contribute only 8 percent of cumulative carbon dioxide (CO₂) emissions.⁴³ High-emitting countries must substantially reduce their emissions to ensure LMICs can develop without exceeding climate targets.⁴⁴ In addition, equalizing trade so that high consumption in the Global North is no longer subsidized by natural resource extraction and labor in the Global South is key to addressing inequality.⁴³

The climate crisis affects some parts of the planet more than others. Historical and present-day injustices have both left Black, Indigenous and people-of-colour communities exposed to far greater environmental health hazards than white communities. Those most affected by climate change are black and poor communities. As a continent we are one of the hardest hit by the impacts of climate change and we are left behind as the world progresses toward a low-carbon economy. Without taking into account those most affected, climate solutions will turn into climate exclusion."

VERONICA MULENGA, ZAMBIAN CLIMATE ACTIVIST⁴⁵

As emerging economies strive to meet people's needs sustainably, climate-smart development can bring wins for the climate and HWB. Synthesis reviews of climate mitigation policy found substantial co-benefits^{46, 47} for air quality, health, and economic benefits. However, few papers have focused on the co-benefits for poverty alleviation, food security, and conflict resilience.⁴⁷ Furthermore, policymakers and decision-makers often fail to consider such co-benefits, resulting in suboptimal climate policies.⁴⁶ It is critical to center HWB co-benefits alongside climate-resilient development as countries experience rapid economic growth, urbanization, and industrialization along with high climate adaptation needs.

Reducing GHG emissions can benefit both climate and people in LMICs, for example through job creation (e.g., in clean energy production) that drives new opportunities for sustainable economic growth. Advances in formal job markets rely on boosting educational quality and learning outcomes—particularly around science, technology, engineering, and mathematics—subjects needed to support growth in jobs centered on low-carbon technology. This demonstrates how climate-smart development also depends on human capital development and raising climate literacy. Climate mitigation and adaptation are two sides of a coin that share the vital aim of diminishing climate change impacts in the immediate and long term.⁴⁸

Development solutions that avoid or reduce GHG emissions can lead to more sustained HWB benefits and reduce risks of conflict and political instability. Globally, ambitious climate actions could generate more than 65 million low-carbon jobs and economic gains of US\$26 trillion by 2030.⁴⁹

FEATURE BOX

Hariyo Ban, Nepal: Centering climate mitigation, adaptation, and livelihoods

Nepal is an ecological hot spot, but climate change threatens people's livelihoods and the country's biodiversity. The Hariyo Ban ("Green Forests")—a community-driven, United States Agency for International Development (USAID)-funded climate adaptation program that ran from 2011 to 2021—worked to conserve biodiversity, build community and ecosystem resilience, and mitigate climate change by reducing emissions from deforestation and forest degradation.⁵⁰ The program's crosscutting work improved gender equality and social equity, and diversified livelihoods; it benefited tremendously from collaborations among local non-governmental organizations (NGOs), government agencies, the private sector, and community organizations with expertise in conservation, governance, community building, livelihoods, gender empowerment, and social inclusion. Results included the sequestration of 3.7 million metric tons of CO_2 emissions and improved access to cleaner energy for 173,000 people.⁵¹ The program contributed to boosting the climate resilience of 400,000 people and provided skills training for 60,000 people in farming and off-farm activities. Hariyo Ban made a concerted effort to include women, youth, and marginalized communities. Hariyo Ban was implemented by a consortium of four NGOs: World Wildlife Fund, CARE Nepal, the Federation of Community Forest Users Nepal, and the National Trust for Nature Conservation; CARE provided the lead on governance and on gender and social inclusion.⁵²



Hariyo Ban provided many benefits to both climate and human well-being over a 10-year period. *Photo credit: Karine Aigner | World Wildlife Fund-United States*

1.4 Financing climate action

Climate action is not cheap, but the cost pales in comparison to the cost of inaction. Huge financial inputs are required to maintain the fossil-fuel-based status quo, and the costs of preventing, dealing with, and recovering from climate-catalyzed disasters, along with the costs of adapting to new environmental conditions in the absence of action, would be astronomical. According to Project Drawdown's financial analysis on the start-up and implementation costs of the suite of climate solutions over time, net operational savings exceed net implementation costs by four to five times (US\$22.5–28.4 trillion versus US\$95.1–145.5 trillion).⁵³ These costs do not include the monetary value of co-benefits (introduced in Chapter 2 of this report), which makes the financial case for immediate climate action even stronger. They also do not include the costs of avoided damages from climate change, which could be extreme.

Although global climate finance has steadily increased over the past decade, it is not keeping pace with needs (Figure 1.3). In 2019 and 2020, about US\$632 billion per year was dedicated to global climate finance, but this annual average is not sufficient to limit global warming to well below 2°C above preindustrial levels.⁵⁴ To meet international climate goals and objectives and avoid further catastrophic climate impacts, an increase of at least 590 percent (to US\$4.35 trillion) in annual climate finance is needed by 2030 from diverse sources, including high-income countries, foundations, private companies, bilateral and multilateral donors, and NGOs, as well as in national budgets of LMICs.^{54, 55}



Figure 1.3 — Current finance and future investment needed to limit global temperatures to 1.5°C above preindustrial levels

Although global climate finance steadily increased between 2011 and 2020, it is not keeping pace with future needs to limit global temperatures to 1.5°C. Source: Climate Policy Initiative

The vast majority of global climate finance (90 percent) goes to mitigation (US\$571 billion), while adaptation funding (US\$46 billion) makes up 7 percent, and dual uses account for the rest (US\$15 billion).⁵⁴ Almost all tracked adaptation finance stems from the public sector and is mainly focused on water, wastewater, and cross-sectoral projects (such as support for capacity building, biodiversity, land and marine conservation, and disaster risk management).⁵⁴ In Africa in 2017 and 2018, 62 percent of total adaptation finance (US\$6 billion) went to agriculture, forestry, land use and natural resource management, and water and wastewater management.⁵⁴

From 2019 to 2020, three-quarters of global climate investments were concentrated in East Asia and the Pacific, Western Europe, and North America, and more than 75 percent of climate investments were raised and spent in the same country (close to US\$479 billion).⁵⁴ During the same period, SSA received the largest share of international adaptation finance (around 25 percent).⁵⁴

These adaptation investments in SSA are much less than what is needed.⁵⁴ High-income countries have yet to fulfill the joint pledges they made in 2009 at COP15^{56, 57, 58} to mobilize US\$100 billion per year by 2020 to meet the adaptation and mitigation needs of the most vulnerable low-income countries (prioritizing small island developing states and Africa).⁵⁹ Recent analyses suggest that climate finance (for both mitigation and adaptation) for low-income countries reached US\$79.6 billion in 2019.⁵⁸ And yet, annual adaptation costs in low-income countries are projected to be US\$155–330 billion by 2030 and US\$310–555 billion by 2050.^{54, 58}

Adaptation ... presents major opportunities to achieve a larger development agenda and put Africa on a new "green" and resilient pathway to growth. Adaptation and development work hand in hand, creating powerful synergies that can increase the chances of meeting global SDGs and additional goals that African nations have set for improving agricultural productivity.⁵⁵

The Intergovernmental Panel on Climate Change (IPCC) estimates the need for US\$1.6–3.8 trillion annually to fund climate solutions;⁶⁰ however, current investments fall far below this amount. Private investments are crucial to address the full scope of climate change.⁶¹ At a recent World Bank conference, Making Climate Action Count: Turning Ambition Into Reality,⁶² Mark Carney, the special envoy on climate action and finance for the UN, stated that funding for climate adaptation and mitigation should come from a mix of the private sector, multilateral grants, and foundations. Fortunately, private sector engagement in, and contributions toward, meeting the Paris Agreement objectives is gaining momentum.⁶³ Investment firms are also engaged, and can catalyze climate action through their investment mandates; the Net Zero Asset Owner Alliance, launched in 2019, includes members with assets in the trillions of dollars that are willing to invest in climate mitigation and adaptation.⁶⁴

New finance mechanisms and innovative funds have great potential to shift climate action and bring about a global transformation that enables LMICs to meet their populations' self-identified needs, reduce risk, preserve natural resources, boost economic growth, and advance climate

goals.⁶⁵ Instead of financing status-quo fossil fuel technologies and practices, funding climatesmart development offers numerous benefits for people and the planet. The cost of inaction is astronomical; yet with intention and care, implementing climate solutions that contribute to a more equitable world is possible.

1.5 Report road map

This report relies primarily on peer-reviewed articles and reports and gray literature from government, NGOs, and donor institutions as the evidence base for examining linkages among 28 Project Drawdown climate solutions and 12 dimensions of HWB. The research cited here focuses on LMICs in South Asia and SSA unless otherwise noted.

Following an overview in Section 1, Section 2 explains the Drawdown Framework for Climate Solutions and highlights a subset of solutions with the potential to generate benefits for poverty alleviation and HWB in LMICs. Section 3 presents an HWB framework that explores linkages among Project Drawdown solutions and their potential co-benefits for HWB. Section 4 provides an in-depth analysis of evidence-based linkages between groups of Project Drawdown solutions and co-benefits for HWB. Section 5 focuses on the way forward for acting on synergistic climate–HWB solutions.

PROJECT DRAWDOWN CLIMATE SOLUTIONS AND CO-BENEFITS FOR HUMAN WELL-BEING

The most recent IPCC reports state that global warming will most likely reach 1.5°C within the next two decades unless GHG emissions are reduced quickly and dramatically, and that parts of the world are quickly becoming unlivable.^{13, 66} Considerable political, economic, and social forces stand in the way of achieving the dual goals of mitigating and adapting to climate change. Yet Project Drawdown's research suggests that it is technically possible, and financially viable, to limit warming safely and equitably to well below 2°C by rapidly scaling existing climate solutions.⁶⁷

2.1 The Drawdown Framework for Climate Solutions

Project Drawdown has identified more than 80 safe and equitable climate mitigation technologies and practices that reduce GHG levels in the atmosphere. These solutions are currently available and financially viable, have proven potential to reduce GHGs, and present sufficient data to model the amount of CO_2 equivalent (CO_2 -eq)^v sequestered or emitted. Project Drawdown published *Drawdown*, the first review and assessment of climate mitigation solutions that can lead to drawdown,^{vi} in 2017.⁶⁸

The *Drawdown Review*,⁵³ an update to *Drawdown*, organized the more than 80 climate solutions^{vii} into a three-pronged framework: 1) reducing *sources* of GHG emissions, 2) supporting nature's carbon *sinks*, and 3) transforming *society* by fostering equality. It defines sector-specific solutions for each group and estimates the impacts of reducing CO₂-eq emissions at the global level for the period 2020–2050 for each solution (Figure 2.1). Each solution's emissions reductions are compared with a baseline scenario whereby few or no new climate solutions

v CO₂ equivalents (CO₂-eq) are used to standardize calculations of climate impact across several GHGs: CO₂, methane, nitrous oxide, and fluorinated gases.

vi Drawdown is the future point in time when levels of GHGs in the atmosphere stop climbing and start to steadily decline, stopping catastrophic climate change.

vii The latest list of Project Drawdown solutions, which will include updates to existing solutions and some new solutions, will launch in 2022.

are adopted, otherwise known as "business as usual." In the baseline scenario, economic development will continue to use existing technologies instead of transitioning to clean ones. For example, with potential emissions reductions from *Distributed Solar Photovoltaics* are based on a comparison with using fossil fuel power plants for electricity generation.





The Drawdown Framework for Climate Solutions illustrates the potential greenhouse gas emissions impact of solution subgroups within each climate solutions sector globally.

2.2 Climate solutions co-benefits for poverty alleviation and well-being

Many of Project Drawdown's climate solutions have multiple co-benefits^{viii} for human development (including poverty alleviation) and climate adaptation, in alignment with the SDGs.^{39,70,71,72,73} For this report, we focused on a subset of Project Drawdown solutions; we excluded solutions that were 1) not applicable to rural under-resourced regions in LMICs and/or 2) did not provide clear co-benefits for the dimensions of HWB and geographic focus of

viii In this report, co-benefits of Project Drawdown climate solutions are defined as tangible and positive effects that a climate action aimed at climate mitigation may have on HWB in sectors such as climate adaptation, gender equality, food security, inclusive economic growth, access to clean energy, health, conservation of natural resources, and others. Co-benefits can be direct or indirect, or have ripple effects, where one co-benefit generates another.⁶⁹ For more information on climate co-benefits in this report, please refer to Section 4. See also⁴⁰ for a deeper discussion on climate change mitigation co-benefits.

this report. Solutions like *Refrigerant Management* and *High-Efficiency Heat Pumps* fall into the first category because they are most applicable to people living in urban areas or in high-income countries. Solutions like *Dynamic Glass* and *Smart Thermostats* fall into the latter category. The 28 Project Drawdown solutions included in this report are those that most clearly generate cobenefits for poverty reduction, human health, gender equality, and overall well-being for rural and underserved communities in LMICs in South Asia and SSA (Figure 2.2).

We recognize that some of the 28 Project Drawdown solutions are applicable in both LMICs and high-income countries. This is the case, for example, for *Temperate Forest Restoration*, *Health and Education*, *Reduced Food Waste*, and several solutions in the *Improving Agriculture and Agroforestry* group. Brief descriptions of each solution are provided in Appendix A (Figure 2.2).





Of the 80-plus solutions in the Project Drawdown framework, 28 have clear well-being co-benefits for rural populations in LMICs. See <u>Project Drawdown</u> for in-depth information and technical reports for each solution.

FRAMEWORK FOR EVALUATING HUMAN WELL-BEING



Climate solutions such as *Nutrient Management* that enhance yield for smallholder farms can, in turn, provide farmers with more money for their families' needs and children's education (an important dimension of human well-being). *Credit: Jonathan Torgovnik* | *Getty Images* | *Images of Empowerment*

3.1 Introduction

Advancing the understanding of HWB in a country and synthesizing information about cobenefits of climate solutions for poverty alleviation and HWB are critical to contextualize and prioritize solutions that address both issues. The Paris Agreement and the SDGs, both developed in 2015, are complementary. The SDGs recognize that people's lives are integrated with various dimensions of human development and the environment. They go beyond national averages (favored by the prior Millennium Development Goals) to articulate a mandate to leave no one behind. Several socioeconomic and well-being indices (Appendix B) are used to help assess progress toward attaining global agreements. Indices that provide a nuanced picture of the well-being of households, societies, or countries, such as the classic Human Development Index (HDI) developed in 1990 by the UN Development Programme and updated annually, have evolved over time.¹ The 2020 HDI included new metrics on planetary pressures such as climate change in recognition that planetary and social imbalances reinforce one another.⁷⁴ Reflecting the more expansive SDGs, measurements of socioeconomic development have thus shifted to be more multifaceted and inclusive visions of HWB. Multidimensional poverty indices, such as the World Bank's Multidimensional Poverty Measure (which go beyond income and consumption to capture deprivations in education, basic infrastructure, and other aspects of HWB) are essential to glean a deeper understanding of various facets of poverty.⁷⁵ The Social Progress Index (SPI) includes several gender-focused indicators, although it does not have a dimension exclusively focused on gender; other indices do not include a gender focus.

3.2 Drawdown Lift Human Well-Being Index

To systematically assess the co-benefits of climate solutions for HWB, this report adapts the Doughnut Economics (DE) framework to create a Drawdown Lift Human Well-Being Index (HWI).

The DE framework has 12 HWB dimensions. It was developed in 2011 and updated in 2017 to align with the SDGs. DE identifies an "environmentally safe and socially just space in which humanity can thrive"⁷⁶ (Figure 3.1). The HWI includes additional and/or slightly modified indicators of HWB, selected to support the evaluation of the co-benefits of Project Drawdown climate solutions for advancing the well-being of rural communities in SSA and South Asia that are experiencing poverty and are vulnerable to climate change.



The HWI is based on the DE framework for two main reasons: its explicit inclusion of gender equality and its adaptability. Climate change and poverty disproportionately affect women and girls (see Subsection 1.2, above). Including indicators of **Gender Equality** is thus critical to evaluate the overall potential HWB co-benefits of climate solutions. Therefore, in addition to the DE framework's "women's representation in government" indicator, the HWI includes three other **Gender Equality** indicators: intimate partner violence, national abortion policies, and demand satisfied for family planning (Figure 3.2).^{ix}

We also modified the indicators in select DE dimensions to address the needs of rural, underresourced, and politically vulnerable communities in LMICs. For instance, in the **Food** dimension, in addition to the DE indicator for percentage of people who are undernourished, we also added an indicator for the prevalence of child malnutrition (Figure 3.2). In **Networks** we added freedom of movement, in the **Social Equity** dimension we added civil liberties, and in the **Peace and Justice** dimension we added political stability. Finally, we added suicide rate under the **Health** dimension in recognition of its importance in the context of LMICs. New indicators included in this report are noted in Figure 3.2 in orange.

The HWI comprises 31 indicators of HWB. It makes it possible to identify disparities in HWB across countries and examine how climate solutions can improve HWB. It is important to note that, although the HWI provides the relative status of national HWB globally, levels of inequality differ widely within and among regions, countries, communities, families, and gender.

We obtained national-level data predominantly from the Word Bank, UN, NGOs such as Transparency International, and global indices such as the Multidimensional Poverty Index (MPI) and Social Progress Index (SPI). Most of the indicators used are also included in the SDGs. Several indicators have a significant intranational rural–urban divide and gender gap. This analysis does not reflect those differences, primarily due to the lack of availability of intranational and gender-resolved data. Gender goes beyond the indicators in the **Gender Equality**^x dimension, as it is important to know how women fare compared with men for dimensions like **Food, Education**, and **Income and Work**.

A brief description of the indicators and a comparison of other indices is provided in Appendix B (Tables B.1 and B.2). Methods for calculating the HWI are in Appendix C.

ix Insufficient data prevented us from disaggregating by gender. We recognize the value and importance of genderdisaggregated data (particularly regarding education and land tenure) and hope to utilize such data in the future.

x Gender is a crosscutting issue, and disparities between women and men exist across all the aspects of HWB, including access to energy, income, health, primary and secondary education, and housing. However, including these disparities across all the dimensions of HWB is beyond the scope of this report. While **Gender Equality** is an outcome in the HWI, gender is also an input and a crosscutting requirement.

| EDUCATION (SDG 4) | GENDER EQUALITY (SDG 5) |
|---|---|
| Literacy rate (DE) Children enrolled in secondary school (DE/SPI) Education quality (SPI) | Women's representation in government (DE) Child, early, and forced marriage and unions (SPI) Intimate partner violence National abortion policies |
| INCOME AND WORK (SDGS 1 & 8) | SOCIAL EQUITY (SDG 10) |
| Poverty rate (DE) Unemployment rate (DE) Gender wage gap (DE) | Income distribution (DE) Civil liberties |
| NETWORKS (SDG 9) | PEACE & JUSTICE (SDG 16) |
| Access to internet (DE/SPI) | Corruption Perception (DE) |
| Access to mobile telephone (SPI) Freedom of movement | Homicide rate (DE/SPI) Political stability |
| · · · · · · · · · · · · · · · · · · · | |
| | Literacy rate (DE) Children enrolled in secondary school (DE/SPI) Education quality (SPI) INCOME AND WORK (SDGS 1 & 8) INCOME AND WORK (SDGS 1 & 8) Gender wage gap (DE) NETWORKS (SDG 9) |

Figure 3.2 — Drawdown Lift Human Well-Being Index Dimensions and related SDGs

DE: Doughnut Economics | SPI: Social Progress Index | MPI: Multidimensional Poverty Index Indicators highlighted in orange are unique to the HWI

The Drawdown Lift Human Well-Being Index uses national-level data for 31 indicators across the 12 dimensions in the Doughnut Economics framework. Indicators from several multidimensional poverty indices are used. Sustainable Development Goals (SDGs) corresponding to each dimension are noted above. SDG 13, Climate Action, contributes to nearly all other 16 SDGs. See also Footnote X on gender. Methods described in Appendix C.

3.3 Global patterns of human well-being

High-income countries have higher overall HWB than LMICs across many dimensions of the HWI (Figure 3.3 and Figures C.1–C.3 in Appendix C). The results are similar to those of the SPI, HDI, and MPI. High-income countries in Western Europe, Australia, Canada, Japan, and New Zealand, followed by the United States, and a subset of countries in South America, Eastern Europe, and Central Asia, have the highest HWI dimensions. It is important to note that greater overall HWI dimensions in high-income countries mask data around specific indicators, such as homicide rate (which is very high in the United States), as well as inequities within high-income countries on specific indicators such as women's representation in government and maternal mortality.

Generally, SSA and South Asian countries have lower HWI values. However, reflective of countries' diverse cultures, history, geopolitics, and governance, there is significant variation in HWB among LMICs. Politically unstable countries such as South Sudan and Somalia have the lowest HWI, reflective of very low values in dimensions focused on human rights, such as **Gender Equality**, **Social Equity**, **Political Voice**, and **Peace and Justice** in addition to high needs in dimensions such as **Food**, **Energy**, and **Education**. Countries in the eastern Sahel, Central Africa, Ethiopia, Eritrea,

and Madagascar have low overall HWI values because they are low in fundamental dimensions such as **Energy**, **Water and Sanitation**, **Food**, and **Health**. Niger and Madagascar are among the world's poorest countries and also suffer from climate-exacerbated droughts and floods.



Figure 3.3 — Drawdown Lift Human Well-Being Index

The Drawdown Lift Human Well-Being Index has lower values in low- and middle-income countries, particularly in sub-Saharan Africa and South Asia. This pattern is largely driven by limited access to energy, clean water, and sanitation; inadequate education; and poor health. Appendix C describes the methods and includes maps of each of the 12 dimensions. *Source: Project Drawdown*

The status of individual HWB dimensions varies among countries and regions (Appendix C, Figures C.1–C.2). Maps in Appendix C can be used to prioritize countries with the most need to boost specific human well-being dimensions and highlight regions where Project Drawdown solutions might be most impactful. For example, **Energy, Water and Sanitation, Education**, and **Health** are lowest in SSA, except South Africa. Although access to electricity, water, and sanitation is moderately high in South Asia, the status of education and health similar to many countries in SSA. Despite a higher degree of electrification in South Asia (> 80 percent of households, except in Myanmar) compared with SSA, electricity deprivation remains an issue due to frequent long power outages in both rural and urban areas. Furthermore, lack of access to clean cooking fuel—an indicator within the **Energy** dimension—is pervasive in South Asia.

Unfortunately, the countries with medium to low overall HWI values are also among those most vulnerable to climate change and low readiness to adapt.²¹ Furthermore, for several countries (e.g., Democratic Republic of Congo, Niger, and Tanzania) the population is expected to more than double within the next 30 years. In these countries and elsewhere, climate solutions can contribute to sustained boosts in HWB.

LINKING CLIMATE SOLUTIONS AND HUMAN WELL-BEING



Distributed Solar Photovoltaics is a climate solution that has numerous human well-being co-benefits and ripple effects that can benefit rural and under-resourced communities in sub-Saharan Africa and South Asia, particularly those with little to no access to electricity. *Credit: Abbie Trayler-Smith | Panos Pictures | Department for International Development*

Climate solutions generate tangible HWB co-benefits and ripple effects. Documenting the HWB co-benefits of climate solutions can help decision-makers understand how the various solutions not only reduce emissions but also address sustainable development needs.

This section links the HWI dimensions with the subset of 28 Project Drawdown solutions (see Appendix A for definitions of each solution) and presents evidence of the solutions' co-benefits and ripple effects on the different dimensions of HWB laid out in Subsection 3.2. Several solutions share HWB co-benefits and ripple effects and are thus grouped together. For example, *Distributed Solar Photovoltaics, Geothermal Power, Microgrids, Small Hydropower,* and *Micro Wind Turbines* are combined because they have similar co-benefits. The solutions are divided into five groups (Table 4.1).

| CLIMATE SOLUTION GROUP | HUMAN WELL-BEING DIMENSIONS | | | | | | | | | | | |
|--|--------------------------------|----------------------------|--------|--------|-----------------------------|--------------------|--------------------|-----------------------------|---------|------------------|----------------------|--------------------|
| | Food | Water and Sanitation | Energy | Health | Education | Income and Work | Gender Equality | Networks | Housing | Social Equity | Peace and Justice | Political Voice |
| IMPROVING AGRICULTURE & AGROFORESTRY | | - | | | - | | | - | - | | | |
| PROTECTING & RESTORING ECOSYSTEMS | | | | | - | | | | | | | |
| ADOPTING CLEAN COOKING | | | • | | | - | | | | | | |
| PROVIDING CLEAN ELECTRICITY | | | | | | - | | - | | | | |
| FOSTERING EQUALITY | | | | | | | | | | | | |
| | | rect co-ber rong evider | | | direct co-be rong evider | | | lirect co-be oderate evi | | Rip | ople effects | |

Table 4.1 — Climate solution groups and human well-being dimensions

The five Project Drawdown solution groups^{xi} offer direct co-benefits, indirect co-benefits, and ripple effects for each of the 12 human well-being dimensions.

Evidence of co-benefits is classified as **strong** when systematic reviews, syntheses, or multiple studies across diverse geographies of peer-reviewed and gray literature report a positive correlation between the climate solution group and the dimension of HWB (Table 4.1). Evidence of co-benefits is classified as **moderate** when there is either limited geographic distribution of the studies or a similar number of studies report a mix of positive, neutral, and negative correlations. Climate solutions can have ripple effects across all aspects of HWB when incorporated into a synergistic set of solutions.

The quality and duration of connections within and across climate solutions and HWI dimensions depend on appropriate policies and investments, community-driven engagement, and local capacity strengthening. Actual development pathways and climate solutions are complex and interwoven, and multidisciplinary cooperation across emissions reduction, poverty alleviation, and HWB dimensions is needed to achieve maximum benefit.

Adopting Clean Cooking: Biogas for Cooking, Improved Clean Cookstoves.

Providing Clean Electricity: Distributed Solar Photovoltaics, Geothermal Power, Micro Wind Turbines, Microgrids, and Small Hydropower.

Fostering Equality: Health and Education.

xi Project Drawdown solutions comprising the solution groups are as follows:

Improving Agriculture and Agroforestry: Abandoned Farmland Restoration, Conservation Agriculture, Farm Irrigation Efficiency, Improved Rice Production, Multistrata Agroforestry, Nutrient Management, Reduced Food Waste, Regenerative Annual Cropping, Silvopasture, Sustainable Intensification for Smallholders, System of Rice Intensification, Tree Intercropping. Protecting and Restoring Ecosystems: Coastal Wetland Protection, Coastal Wetland Restoration, Forest Protection, Grassland Protection, Indigenous Peoples' Forest Tenure, Peatland Protection and Rewetting, Temperate Forest Restoration, Tropical Forest Restoration.

Co-benefits are described as direct if there is an explicit—often material—link between action and outcomes. For instance, the link between the *Improving Agriculture and Agroforestry* solutions and **Income and Work** is direct because higher crop yield and improved crop resilience to climate extremes results in higher income. A co-benefit is indirect when a direct co-benefit enables other co-benefits, such as an increase in income due to *Improving Agriculture and Agroforestry* solutions enabling better **Education**. Direct and indirect cobenefits of climate solutions can also lead to cascading effects on other HWB dimensions. These are classified as ripple effects.

Subsections 4.1–4.5 provide detailed evidence of the HWB co-benefits of each climate solution group. The solution groups appear in order of their total global climate mitigation potential from greatest to least.^{xii} Hence, we first present the evidence for *Improving Agriculture and Agroforestry* (highest net CO₂-eq reduction potential) followed by *Protecting and Restoring Ecosystems, Adopting Clean Cooking, Providing Clean Electricity,* and *Fostering Equality.* Within each solution group, we present the HWI dimensions in descending order of the strength of the evidence.

We discuss the evidence around direct co-benefits of *Improving Agriculture and Agroforestry, Protecting and Restoring Ecosystems,* and *Fostering Equality* in subsections 4.1, 4.2, and 4.5. **Energy** is the only direct HWB co-benefit for *Providing Clean Electricity and Adopting Clean Cooking;* without energy, none of the indirect co-benefits are possible. Thus, in subsections 4.3 and 4.4, we also provide evidence for indirect co-benefits.

In subsection 4.6, we discuss the indirect co-benefits of *Improving Agriculture and Agroforestry, Protecting and Restoring Ecosystems*, and *Fostering Equality* (the direct co-benefits are discussed in each group's subsection) as well as the ripple effects of all five solution groups. We combined discussion of indirect co-benefits and ripple effects to avoid repetition of evidence as the co-benefit and ripple effect pathways for these HWB dimensions were similar for all of the solution groups.

4.1 Climate solutions for improving agriculture and agroforestry

In 2020, more than 2.3 billion people—one-third of the global population—experienced moderate or severe food insecurity.⁷⁷ The burden is not evenly distributed. Hunger, measured as undernourishment, is most prevalent for people living in Africa (21 percent of the population), Asia (9 percent), and Latin America and the Caribbean (9 percent).⁷⁷ Food insecurity among women is 10 percent higher than for men.⁷⁷

The regions in which hunger is prevalent also have high rates of poverty. More than threequarters of the world's most under-resourced people live in rural areas.⁷⁸ Over half of people living in poverty are in SSA.⁷⁸ The percentage of people living in extreme poverty in South Asia has been declining rapidly (8.6 percent of people in 2018), but the trend is slowing.⁷⁸

xii The CO₂-eq calculations for each solution group are at the global level. We used Project Drawdown's 1.5°C Ambitious scenario, which calculates the total atmospheric CO₂-eq reduction between 2020 to 2050 for each solution (updated in early 2022) compared with a baseline scenario that utilizes a business-as-usual approach.



Climate solutions that improve agriculture and agroforestry are key to addressing poverty and eliminating hunger. Land tenure for women is an important factor to address; research shows that women farmers make up around 40 percent of the agricultural labor force in six sub-Saharan Africa countries. *Credit: Jonathan Torgovnik* | *Getty Images* | *Images of Empowerment*

Improving agricultural production is a key pathway for eliminating poverty and hunger. The Food and Agriculture Organization (FAO) estimates that improvements in agriculture are 11 times more effective at reducing extreme poverty in SSA than investments in other sectors.⁷⁹ Another study estimated that the country-level internal rate of return for investments in the agricultural sector across SSA ranged from 17 percent to 43 percent.⁸⁰ Despite its promise, substantial investments in agriculture in LMICs are still needed. In Africa, millions of people are already suffering the effects of climate change on rural livelihoods, where, by 2030, almost 200 million people will reside in areas where temperatures will have increased more than 2°C, affecting US\$23 billion in crop and livestock production value.⁸¹

Investments in improving agriculture must also focus on women. In six SSA countries, women farmers make up around 40 percent of the agricultural labor force.⁸² Although most women in LMICs are employed in agriculture, 85 percent lack the right to own land.⁸³ Achieving the direct and indirect co-benefits of land ownership is closely linked to land tenure. Secure land tenure, such as the *Indigenous Peoples' Forest Tenure* solution, increases the probability that farmers will adopt sustainability practices⁸⁴ like those described above, which lead to improved food security.⁸⁵ Increasing land tenure is particularly important for women⁸⁶ as it is linked to greater adoption of sustainable practices and better ecological outcomes.⁸⁷

Regions where hunger, poverty, and farming are all prevalent also are the most vulnerable to climate change. Crop yields are projected to decrease most strongly across the subtropics and tropics⁸⁸ in both moderate and high warming scenarios. More alarming, climate change has already decreased staple crop yields in many countries,⁸⁹ including a 30–35 percent decrease across SSA.⁹⁰

Globally, producing crops and livestock on croplands, pastures, and agroforestry systems accounts for ~15 percent of GHG emissions.⁹¹ The primary sources of these emissions are methane from livestock and rice production as well as nitrous oxide (N_2O) emissions from

fertilizer and manure.⁹¹ Although total GHG emissions are much lower in most LMICs than high-income countries, agriculture is the source of a high percentage of emissions in countries where there is extensive production of rice or cattle (from methane), excessive fertilizer application (from N₂O), or deforestation for agricultural expansion (from CO₂). The critical role of food production in climate mitigation is underscored by the fact that more than 90 percent of Nationally Determined Contributions reference agriculture.⁹²

The primary climate benefit of *Improving Agriculture and Agroforestry* is increased carbon storage in soil and vegetation, particularly in agroforestry systems where trees are prevalent,^{93, 94} although the amount of storage is context specific. Similarly, environmental, social, and economic factors influence the HWB co-benefits. Solutions must incorporate a complementary approach to minimize trade-offs.⁹⁵ For example, fertilizer inputs to boost crop yields can lead to reduced water quality and increased N₂O emissions.

Twelve of Project Drawdown's agriculture- and agroforestry-related climate solutions (*Abandoned Farmland Restoration, Conservation Agriculture, Farm Irrigation Efficiency, Improved Rice Production, Multistrata Agroforestry, Nutrient Management, Reduced Food Waste, ^{xiii} Regenerative Annual Cropping, Silvopasture, Sustainable Intensification for Smallholders, System of Rice Intensification, and Tree Intercropping),^{53, 96} have direct co-benefits for HWB, particularly around Food and Income and Work, as well as several indirect co-benefits (Figure 4.1).*



Figure 4.1 — Human well-being benefits of climate solutions for Improving Agriculture and Agroforestry

This systems diagram highlights the direct (thick gray line) and indirect (thinner gray line) co-benefits of *Improving Agriculture and Agroforestry* for the 12 dimensions of human well-being. Ripple effects are shown by dashed gray lines. Local stakeholders, governments, and change makers are implicit along the connecting arrows. Potential CO₂-eq reduction for 2020 to 2050 is 277.6 gigatons.

xiii The *Reduced Food Waste* solution focuses on both food loss and waste. Given that this report discusses LMICs, where food loss is generally more of an issue, that is the focus for this section.

Human well-being benefits of improving agriculture and agroforestry

A group of agriculture and agroforestry climate solutions have direct co-benefits for **Food**, **Income and Work**, and **Water and Sanitation**. The high number of solutions reflects the diverse

environmental and social conditions that span croplands, rangelands, and agroforestry systems. Several climate solutions—such as Sustainable Intensification for Smallholders, Conservation Agriculture, Regenerative Agriculture, Nutrient Management, Improved Rice Production, and Farm Irrigation Efficiency—increase food production while reducing climate and other environmental impacts. A second set of solutions, including Multistrata Agroforestry and Silvopasture, diversify food production, which can lead to more diverse diets and sources of income. Other solutions such as Reduced Food Waste and Loss and Abandoned Farmland Restoration increase the amount of food that is available or can be produced on existing agricultural lands.



Climate-smart agriculture can provide increased income, lead to more diverse diets, and provide environmental benefits. *Credit: Neil Palmer | Center for Tropical Agriculture (CIAT) 2011*

Food: Climate solutions improve food security directly by increasing yields and dietary diversity and buffering against weather shocks. Several climate solutions increase production on current fields, reducing the need to farm additional land. This is particularly important in densely populated rural areas, where land is limited.⁹⁷ Although yields of staple grains, which account for the majority of caloric intake, have stagnated across LMICs,⁹⁸ 70 percent of studies in a recent review of intensifying production on smallholder farms found increases in yields and/or income.⁹⁹

Increasing yields of rice while also reducing its climate impact is critical, as rice is the largest source of GHG emissions from croplands^{100, 101} and the staple food of 3.5 billion people. Using the best management practices for planting times, nutrient application, crop varieties, irrigation timing, and pest control has been shown to increase yields,¹⁰² reduce methane emissions,¹⁰⁰ and improve water availability.¹⁰² However, some solutions, like *System of Rice Intensification*, have limited adoption or applicability in certain geographies because of increased costs, time, soil conditions, and labor requirements.

Several studies found that changing management practices to improve soil fertility increases yields through application of organic matter and fertilizer.⁹⁹ Because the relationship between fertilizer application and N₂O emissions is nonlinear,¹⁰³ small additions of nitrogen fertilizer where little (or no) nitrogen is currently applied can increase crop yields with very low increases in N₂O emissions. For example, small additions of fertilizer improved crop yields and increased net income by an average of 33 percent in six countries across SSA.¹⁰⁴ The opposite is true in India and China, where government subsidies have led to the highest rates of nitrogen fertilizer application globally. In these countries, a small decrease in application rates has no impact on yield but dramatically reduces N₂O emissions.¹⁰³ Further, excess nitrogen on farmlands degrades air and water quality,^{91,105} and high amounts can acidify soils,^{106,107} in turn decreasing crop yields.

While the potential outcomes are highly context dependent, *Multistrata Agroforestry*, *Silvopasture*, and *Tree Intercropping* solutions all increase carbon storage and improve food security.¹⁰⁸ Although there is mixed evidence of the effects of these practices on yield,⁹⁹ the higher diversity of food products¹⁰⁹ in agroforestry systems is associated with higher dietary diversity.¹¹⁰ These solutions also improve food production stability and drought resilience by creating diverse microclimates^{111, 112} and increasing soil organic matter that can hold more water and nutrients.¹¹³

Income and Work: Increasing **Income and Work** is primarily associated with increasing yields. Additional production can be sold and/or reduce the income spent on food to supplement household farm production. Although the yields of individual crops can be lower in agroforestry systems,⁹⁹ the diversity of foods produced provides a buffer against economic shocks due to market prices or crop failure.^{114, 115}

Reducing food waste and loss reduces the amount of food that must be produced or purchased. Globally, about one-third of all food produced is wasted.¹¹⁶ In LMICs, food loss primarily occurs during production, post-harvest handling, and storage.¹¹⁶ Effective strategies for reducing food loss in on-farm practices, such as post-harvest storage^{117, 118} and integrated pest management, as well as off-farm efforts to improve roads, access to electricity, centralized storage, processing, and market development have been demonstrated to improve food security and increase income.¹¹⁷

Water and Sanitation: Increasing *Farm Irrigation Efficiency* has direct benefits to food security, income, and water availability and quality. Crop production in China and India accounts for 49 percent of water consumption in irrigation and 43 percent of global N₂O emissions.¹¹⁹ Pakistan consumes another 15 percent of global water, largely for cotton production.¹¹⁹ Although much of the irrigation is from storing and redirecting surface water (so there are few GHGs to mitigate), water consumption in those countries is greater than the amount replenished each year.¹²⁰ Reducing water for irrigation increases water available for people and avoids increased vulnerability should withdrawals become unsustainable.

Increasing *Farm Irrigation Efficiency* has also been shown to increase food production, income, and nutritional intakes.^{121, 122} In Zimbabwe, drip irrigation enabled rural households in a dry village to significantly improve their nutrition and food consumption.¹²³ In Benin, households using solar-powered drip irrigation increased their fruit and vegetable production 26 percent and 55 percent, respectively.¹²⁴ Similar results were observed in the Sahel region of northern Benin, where solar-powered drip irrigation improved both household nutritional intake and income in the dry season.¹²⁵

The need for irrigation can be reduced through *Multistrata Agroforestry* and *Silvopasture*, which increase soil carbon found in organic matter^{93, 94} and boost soils' ability to store water.
FEATURE BOX Climate-Smart Village project, India: Improving food security, income, and education

Nearly 30 percent of India's population experiences multidimensional poverty,¹²⁶ most of whom live in rural¹²⁷ areas and work in agriculture.¹²⁸ Climate change has already decreased the yields of the two main crop staples—wheat and rice.⁸⁹ The four-year (2016–2020) Climate-Smart Village project, managed by the CGIAR Research Program on Climate Change and Food Security, highlights how climate solutions can be combined with a suite of additional interventions that improve livelihoods.¹²⁹ Climate solutions such as increasing irrigation efficiency, composting, and reducing fertilizer all reduced GHG emissions and increased yields, water availability, and income. These practices and outcomes were further reinforced by establishing small businesses that loan and sell machinery, fertilizer, and pesticides. Women played central roles in decision-making and governance. Adding household gardens provided another source of income as well as improved nutrition. Installing clean cookstoves added benefits by reducing cooking time and the need to gather fuelwood and improving household air quality. The improved food security, lowered agricultural input costs, diversified income sources, and reduced time for gathering fuelwood enabled women to focus more attention on childhood education. Additional climate solutions, such as Distributed Solar Photovoltaics or Microgrids, would be needed to enable these benefits in places that currently lack access to electricity or where access to electricity is patchy and intermittent. Further, since increased production can suppress crop prices, policies that create markets and storage and processing facilities may be needed to ensure increased farm income.^{129, 130, 131}



The Climate-Smart Village Project in northern India implemented irrigation and fertilizer efficiency, composting, and clean cookstoves while advancing women's decision-making authority and establishing new businesses. *Credit: Jacquelyn Turner | CGIAR Research Program on Climate Change, Agriculture and Food Security (CCAFS)*

4.2 Climate solutions for protecting and restoring ecosystems



Protecting forests, such as the Southwest Mau in southern Kenya, can help to support the more than 1 billion people experiencing extreme poverty who rely on forests to meet their basic needs for housing, water, fuel, and income. *Credit: Patrick Sheperd | Center for International Forestry Research*

Ecosystems provide multiple benefits for people. These include provisions such as food, water, and timber; protection from or reducing the impact of flooding, storm surges, and drought; and cultural and spiritual significance.^{132, 133} Globally, more than 1 billion people depend on nature to meet their basic needs for housing, water, and fuel, as well as their primary source of income.¹³⁴ And over 1 billion people living in extreme poverty rely on forests for their livelihoods.¹³⁵ About 75 percent of people in Central and Eastern Africa rely on nature¹³⁴ —mostly for subsistence living. Thus, they are particularly vulnerable to extreme weather, climate change, and environmental degradation.

Indigenous peoples rely heavily on natural ecosystems and play a central role in conserving them, as they manage or have tenure on 25 percent of the world's land—and 40 percent of that land overlaps with protected areas or intact ecosystems.¹³⁶ Most of these lands are in boreal and tropical forest regions, although other regions are also critical. For example, around 100–200 million people are nomadic pastoralists¹³⁷ who move animals across large natural landscapes that cross countries and span diverse regions of the Sahel, Serengeti plains, Mongolian grasslands, and Arctic tundra. Aquatic and coastal regions are also important for conservation; for example, the UN Environment Programme (UNEP) estimates that the livelihoods of 1.2 million people depend on mangroves in the Western Indian Ocean countries of Kenya, Madagascar, Mozambique, and Tanzania.¹³⁸

Habitat destruction and degradation reduce nature's many benefits to people and are a major source of GHG emissions. Consumption in high-income countries drives appropriation of natural resources from LMICs with negative consequences for protecting and sustaining ecosystems, while also fueling inequality.¹³⁹ Deforestation throughout SSA is largely due to clearing land for subsistence agriculture; a few commodities driven by large-scale agriculture account for most of the deforestation in Southeast Asia.¹⁴⁰ In addition to national policies that either encourage agricultural expansion or do not regulate deforestation, large-scale land acquisitions (or land grabs), commonly by foreign investors, accelerate deforestation and other habitat destruction,^{141, 142} and reduce food security.^{142, 143}

To stop biodiversity loss, researchers have proposed formally designating the protection of 30 percent of all terrestrial and marine areas.¹⁴⁴ In the past decade, at the global level, the area designated for land and marine biodiversity conservation expanded by 42 percent.¹⁴⁵ However, the increase in protected areas on land (28.1 million square kilometers)¹⁴⁵ was much smaller than the area deforested,¹⁴⁶ which does not include the loss of grasslands, savannas, coastal areas, and other ecosystem types. By storing carbon in vegetation and soil, nature-based climate solutions that conserve, restore, and improve land management across forests, wetlands, grasslands (and agricultural lands) can contribute to climate mitigation needed to hold global average temperature increases to well below 2°C, while also enhancing soil health, cleaning air and water, and sustaining biodiversity.¹⁴⁷



Healthy forests often provide a variety of nutrient-rich foods that women commonly forage for and cultivate to provide for their families. *Credit: Joe Nkadaani | CIFOR*

Clearing natural ecosystems releases carbon stored in vegetation and soils into the atmosphere. These emissions from land use change account for 11 percent of global GHG emissions,⁹¹ primarily driven by deforestation for agricultural expansion in the tropics.¹⁴⁸ Protecting ecosystems maintains the carbon they store (avoided emissions) and enables them to continue storing more carbon each year. Restoring ecosystems benefits climate through vegetation pulling CO₂ from the air and storing it in plants and soils. Project Drawdown estimates that *Protecting and Restoring Ecosystems* could reduce emissions between 2020 and 2050 by as much as 63.1 gigatons (Gt) and store an additional 120.1 Gt of CO₂.⁵³

Eight Project Drawdown solutions focused on *Protecting and Restoring Ecosystems* (Coastal Wetland Protection, Coastal Wetland Restoration, Forest Protection, Grassland Protection, Indigenous Peoples' Forest Tenure, Peatland Protection and Rewetting, Temperate Forest Restoration, and Tropical Forest Restoration) have direct and indirect co-benefits for HWB.^{XIV}

Human well-being benefits of protecting and restoring ecosystems

Climate solutions related to *Protecting and Restoring Ecosystems* provide many benefits for HWB. Nature provides many goods and services that benefit people^{133,149} and a critical safety net for those who rely on nature to meet their basic daily needs through subsistence use of lands and waters.

xiv Project Drawdown will add more marine-based solutions in 2022.

Securing *Indigenous Peoples' Forest Tenure*^{xv} and involving local communities in resource management decisions are critical to achieving the dual objectives of improving people's livelihoods and conserving nature.⁸⁴ Approximately 22 percent of all forest carbon in tropical and subtropical countries is on lands managed by Indigenous peoples.¹⁵⁰ The forest carbon on Indigenous forested lands is primarily (65 percent) in soils¹⁵⁰ in the three largest tropical peatland regions (*Peatland Protection and Rewetting*): the islands of Southeast Asia,¹⁵¹ the central Congo River Basin,¹⁵² and southwestern Amazonia.¹⁵³

Positive ecological outcomes are more associated with land tenure security than with particular types of land tenure.¹⁵⁴ Most studies that examine the effect of interventions that improve land tenure security show positive outcomes for both HWB and nature.⁸⁴ Although securing land tenure creates a strong foundation for improving HWB, protecting nature and land tenure strategies must be part of a systemic approach that designs interventions to realize multiple benefits and ensures they are equitably distributed. Co-management by Indigenous peoples¹⁵⁵ and local community-based forest management¹⁵⁶ are associated with better conservation and social outcomes, although maintaining such efforts can be challenging without sustainable sources of funding. However, in some instances, efforts to formalize tenure for community-based forest management areas reduced people's access to and use of the land.¹⁵⁶



Figure 4.2 — Securing land tenure for women leads to:

Strengthening land rights for women provides many human well-being benefits. Credit: Mariola Acosta, Research Fellow | IITA/CCAFS

xv For consistency with other Project Drawdown materials, this report uses the solution name *"Indigenous Peoples' Forest Tenure."* However, the evidence presented here also includes land tenure security for local communities and women.

Studies across South Asia, SSA, and Latin America have demonstrated that women's involvement in forest governance and management resulted in higher adoption of sustainable resource management practices that led to improved ecological conditions.

Women play important roles in land tenure security (Figure 4.2). Studies across South Asia, SSA, and Latin America have demonstrated that women's involvement in forest governance and management resulted in higher adoption of sustainable resource management practices that led to improved ecological conditions.^{157, 158} Women's influence on improving forest ecological conditions is likely a result of women's greater dependence on and knowledge about forest resources, as women are the primary providers and collectors of food and fuelwood for the family and increase income by selling non-timber forest products (NTFPs).¹⁵⁸ As a result, the direct and indirect co-benefits (Figure 4.3) are highly dependent on land tenure security for women and more fully engaging women in all aspects of *Protecting and Restoring Ecosystems*.



Figure 4.3 — Human well-being benefits of climate solutions for Protecting and Restoring Ecosystems

This systems diagram highlights the direct (thick gray line) and indirect (thinner gray line) co-benefits of *Protecting and Restoring Ecosystems* for the 12 dimensions of human well-being. Ripple effects are shown by dashed gray lines. Local stakeholders, governments, and change makers are implicit along the connecting arrows. Potential CO₂-eq reduction for 2020 to 2050 is 181.5 gigatons.

Most studies that assess the impact of *Protecting and Restoring Ecosystems* on HWB focus on income and food security in forested environments.^{84,154,155,159}

Income and Work: The strongest evidence linking *Forest Protection* with poverty alleviation is for ecotourism, community forestry, and agroforestry as drivers of income generation.¹⁶⁰ Positive economic outcomes for local communities are associated with land tenure security through community-based management.^{87,155,156,161}

The most widespread source of income from nature is the sale of NTFPs such as fuel, food, and medicine. NTFPs make up 22–28 percent of the income of rural people experiencing poverty who live in or adjacent to forests.^{162,163} Dependence on NTFPs for income is highest for those living in extreme poverty.^{163,164} However, income from NTFPs may not reduce poverty, as NTFPs are largely consumed by individuals in the community rather than sold outside the community.¹⁶⁵ In such cases, forests still benefit HWB by reducing the income needed to purchase the goods and services that can be accessed from natural areas.

Ecotourism can be an important source of income. Tourism related to coral reefs alone generates US\$36 billion each year.¹⁶⁶ However, several studies show that individuals who benefit from ecotourism were already doing relatively well,¹⁶⁰ so ecotourism could increase inequality. Community involvement in decision-making about ecotourism increases the benefits to local communities.¹⁶⁷

Food: *Protecting and Restoring Ecosystems* can improve food security. Several studies across LMICs found that living in or near forested areas improved food security by either increasing food availability and dietary diversity^{168,169} and/or increasing capacity to purchase food with tourism-generated income.¹⁶⁹ In grass-dominated ecosystems (*Grassland Protection*), nomadic pastoralists move cattle across large natural landscapes to produce food; these ecosystems generate the majority of annual GDP from agriculture in arid countries like Niger and Senegal.¹⁷⁰ In coastal regions, particularly in small island developing states, mangroves and other coastal ecosystems (*Coastal Wetland Protection, Coastal Wetland Restoration*) support subsistence fisheries that are critical for food security.^{133, 171} For example, a 10 percent increase in restored mangrove area (*Coastal Wetland Restoration*) near a national park in Tanzania doubled income from shrimping near the restoration site.¹⁷² Marine protected areas can lead to increased dietary diversity.¹⁷³



Indigenous peoples, such as the Maasai in Tanzania and Kenya, are among those most impacted by climate change despite contributing the least to its causes. They are keepers of important traditional knowledge. *Credit: William Warby*

Water and Sanitation: The condition of ecosystems affects water quantity and quality.^{132, 133, 174} Human demand for water is changing the quantity and distribution of freshwater resources¹⁷⁵ and has resulted in scarcity in many regions.¹²⁰ Fertilizers, soil erosion, sewage, and industrial chemicals are major sources of water quality degradation. *Protecting and Restoring Ecosystems* improves water quantity and quality by facilitating filtration of precipitation by vegetation and soils. For example, a study of children across 35 countries found that increasing upstream forest cover (*Forest Protection, Temperate Forest Restoration,* and *Tropical Forest Restoration*) in a watershed by 30 percent decreased diarrheal disease to a rate similar to that from improvements in sanitation.¹⁷⁶

Health: Natural ecosystems reduce the incidence of vector-borne diseases and exposure to extreme weather, improving the living and working environment and boosting mental health.

Ecosystem degradation can increase the likelihood that pathogens such as SARS-CoV-2 and Ebola viruses will jump from wildlife to people.¹⁷⁷ Deforestation and other forms of ecosystem degradation also increase the incidence of several vector-borne diseases. For example, the incidence of malaria increased following deforestation in the rainforests of Brazil,^{178, 179} Indonesia,¹⁸⁰ and Uganda.¹⁸¹ Deforestation is also linked to increased incidence of arboviruses, Chagas, and other vector-borne diseases.¹⁸²

Meta-analyses highlight several studies that report a positive correlation between biodiversity and better health.¹⁸³ Children who live near protected areas like national parks were found to be healthier than children living far from protected areas.¹⁶⁹ Similarly, children living near forested areas (*Forest Protection, Temperate Forest Restoration*, and *Tropical Forest Restoration*) have higher dietary diversity than those living farther away from forests across 27 LMICs.¹⁶⁸ As forests are cleared or degraded, people must spend more time hunting and/or gathering food and fuelwood in the forest, which can have negative impacts on their health.

Children who live near protected areas like national parks were found to be healthier than children living far from protected areas.

Extreme weather poses a major global public health challenge for both psychological health and well-being, especially for the most vulnerable people in LMICs (and those who lack accessible, quality mental health care infrastructure)^{184, 185} and children.¹⁸⁶ Natural ecosystems can buffer exposure to such events. Mangroves (*Coastal Wetland Protection, Coastal Wetland Restoration*) and coral reefs reduce the force of harmful waves during storm surges.^{187, 188} Deforestation elevates heat stress for workers, reducing cognition and productivity,^{189, 190} particularly in LMICs where heat waves hit under-resourced regions hardest.¹⁹¹ Experiences of extreme weather events have been linked to anxiety and depression in children.¹⁹²

FEATURE BOX

Gorongosa National Park, Mozambique: Lifting people out of poverty through conservation

The Gorongosa region experienced civil war and conflict for decades, devastating people. wildlife, and the ecosystem. The Gorongosa project began in 2008 as a 20-year public-private partnership between the government of Mozambique and the Carr Foundation to revive and conserve Gorongosa National Park, reduce conflict in the region, and sustainably unlock the park's economic potential as a driver for human development in nearby communities. It is funded primarily by USAID, Canada, Norway, and Irish Aid along with the Carr Foundation and includes collaboration among multiple local and international organizations, government agencies, and universities. The project recognizes that environmental conservation is possible by lifting people out of poverty. Integrating development and conservation are key components of the initiative, which addresses all 17 SDGs. Wildlife poaching incidence has decreased by 72 percent, and populations (including several endangered species) have increased. Girls' retention in school has significantly improved. Youth skills in environmental education, eco-tourism, and conservation have been enhanced. More than 11,000 farmers have increased crop yields due to better farming practices. The opening and improvement of health centers and training of health professionals has led to a significant decrease in malaria cases and improved child vaccination rates, and contributed to nutritional interventions for more than 30,000 children. Investments in sustainable agriculture activities boost livelihoods; products include Gorongosa coffee, cashew, fish, honey, and crafts. Peace clubs reintegrate former combatants into society. Health programs, in collaboration with the Ministry of Health, focus on water, sanitation, and hygiene (WASH), including in schools. These programs also support community health workers who provide primary health care services to

rural communities, such as treatment for common diseases, family planning, and prenatal and postnatal care. The new forestry and climate action program protects and restores forests and plans to seek carbon offset certification. The project's success has led to an extension of the public–private partnership for another 25 years, a recognition of Gorongosa as a global model for alleviating poverty and improving human well-being while conserving biodiversity.¹⁹³

Gorongosa National Park in Mozambique focuses on both conservation and poverty alleviation while addressing nearly all of the SDGs. *Credit: Brett Kuxhausen | Gorongosa National Park*

4.3 Climate solutions for adopting clean cooking

Approximately 2.6 billion people across 71 countries—one-third of the global population—use solid fuels, including wood, charcoal, dung, and crop residues, for cooking.¹⁹⁴ Household air pollution (HAP) resulting from the use of traditional cookstoves contributes to stroke, ischemic heart disease, chronic obstructive pulmonary disease, lung cancer, and adverse birth outcomes including low birth weight,¹⁹⁵ particularly in the most under-resourced settings.¹⁹⁵ HAP is estimated to cause more than 4 million premature deaths¹⁹⁴ every year and accounts for almost

4 percent of the global burden of disease,¹⁹⁵ and air pollution (including ambient ozone, ambient PM_{2.5}^{xvi}, and household pollution) contributes to nearly 12 percent of global deaths worldwide.¹⁹⁵ According to the most recent State of Global Air report,¹⁹⁵ in 2019, 0.5 million infant deaths in the first month of life can be attributed to poor air quality. The global loss of life expectancy attributable to HAP is 0.7 years, but it is approximately two years or higher in several SSA countries.¹⁹⁶ In Myanmar, where the use of traditional cookstoves is common, the relative risk of neonatal, infant, and under-5 mortality was two times higher in households using solid fuel than in those using clean fuel.¹⁹⁷ An analysis of children's exposure to HAP across 14 countries in SSA found a 30 percent higher risk of under-5 mortality for children who lived in homes with indoor cooking than for children in homes where cooking is done in a separate building or outside.198



Scalable, affordable, acceptable clean cooking technologies and user-informed, culturally grounded approaches will help to increase adoption of clean cooking that can lead to many Energy, Health, Income and Work, Education, and Gender Equality human well-being co-benefits. *Credit: Horst Vogel | Flickr*

Households must be able to secure fuel to provide food for their families, despite high socioeconomic costs. It is estimated that the combined adverse impacts of solid fuel cooking on health, the environment, and gender equality costs US\$2.4 trillion annually.¹⁹⁹ Traditional wood and coal cookstoves can also have adverse environmental and climatic impacts, although it is difficult to delineate household fuel use from extraction for other purposes such as timber and challenging to accurately account for demand and regeneration.²⁰⁰ And yet, creating scalable, affordable, and acceptable clean cooking^{xvii} options remains a major challenge.²⁰¹

xvi PM_{25} is defined as fine inhalable particles that are ≤ 2.5 micrometers in diameter.

xvii We define clean cooking as primary use of modern stoves and fuels (e.g., solar, electric, biogas, natural gas, liquefied petroleum gas [LPG], and alcohol, including ethanol) instead of coal and biomass fuels (e.g., wood, charcoal, dung, crop residues). However, Project Drawdown's calculation of the climate impact of these solutions excludes natural gas and LPG as fuel sources. We recognize that LPG has a much lower climate impact than coal and biomass and has an important role to play in cleaner cooking options in LMICs.

FEATURE BOX

Time spent by women and girls every day collecting fuel for cooking in several sub-Saharan African countries

Collecting fuelwood can be time-consuming, taking up to six to thirty hours per week. In many communities—largely dependent on cultural values and norms—the use of traditional cookstoves disproportionately affects women and girls as they spend hours daily collecting fuel, with associated physical, mental, and economic impacts.^{202,203} In most countries gathering fuelwood falls predominantly on women and girls, although in some countries, such as India, Indonesia, and Nepal, men also collect fuel.^{204,205} Adoption of clean cooking can substantially reduce the "time poverty" associated with collecting cooking fuel, providing family members, especially women and girls, the opportunity to pursue education and economic activities. Research in The Gambia, Guinea-Bissau, and Senegal showed that redeploying 50 percent of the time saved on gathering firewood to income-generating activities could result in an approximate annual income of US\$83, US\$100, and US\$170, respectively.²⁰⁶



Household members in some African countries spend anywhere from less than one to five hours per day collecting fuel; in several countries, nearly the entire population relies on biomass to cook food. *Source: Organization for Economic Co-operation and Development.*²⁰⁷

At the global level, 1.9–2.3 percent of GHG emissions is derived from the use of wood fuels.²⁰⁸ A World Bank analysis²⁰⁹ showed that replacement of a traditional solid-fuel cookstove with a low-polluting cookstove can save up to 0.94 metric tons of CO_2 -eq per year per cookstove; full adoption of clean cooking technologies could reduce global emissions by up to 1 billion metric tons of CO_2 -eq every year between 2020 and 2050.²¹⁰ Further, traditional wood and coal stoves also produce black carbon, a pollutant that can accelerate near-term regional warming.

Phasing out traditional cookstoves could cool global temperature by 0.1°C by 2050²¹¹ and improve human health. The highest potential impacts for both global temperature change and health (due to ambient air pollution) are in LMICs in Asia and SSA—including Bangladesh, Ethiopia, India, and Nigeria.²¹¹

Clean cooking is a powerful climate solution, both in the short and long term, and it also has impressive HWB co-benefits across multiple dimensions. Many people have benefited from impactful clean cooking projects in LMICs; since 2010, the Clean Cooking Alliance has worked to support rigorous research. And yet, the evidence regarding the potential co-benefits for HWB remains mixed.²¹² Several studies have shown positive effects, while others have found limited benefits due to lack of oversight and monitoring of projects, mixed use of clean cooking and traditional methods at the household level, and unaffordability.^{213, 214} The majority of studies analyzing the benefits of clean cooking focus on environmental and health benefits rather than time savings.²¹⁵ A systematic review of 45 publications on clean cooking from biomass to clean fuel, both of which showed significant time savings.²¹⁵ To this end, there is a call to include additional questions in national surveys on the gendered dimensions of household energy use²¹⁶ and to explore the co-benefits of clean cooking on time poverty and gender equality.

The unsatisfactory adoption of clean cooking despite decades of efforts (and therefore poor research outcomes) could be due to a lack of understanding of the lived experience of extreme poverty and attendant fears and stresses.²¹⁷ Notwithstanding the research-focused limitations and shortcomings, we present the HWB co-benefits evidence below.



Adopting clean cooking stoves reduces the time spent collecting fuelwood (typically 10–30 hours each week per household) and improves indoor air quality for women and children. *Credit: Ollivier Girard | CIFOR*

Two Project Drawdown solutions, *Biogas for Cooking* and *Improved Clean Cookstoves*^{xviii} generate significant HWB co-benefits and ripple effects for people in under-resourced, rural communities in LMICs.

xviii We recognize that the terminology around "improved" or "clean" cookstoves can be contentious. We have retained its use in the report as *Improved Clean Cookstoves* is the name of the Project Drawdown solution related to clean cooking. It is important to note that the "clean" aspect of such cookstoves is reliant on the type of fuel used. Cookstoves should be both user- and women-informed to meet the diverse energy needs of households.

Human well-being benefits of clean cooking

Adopting clean cooking provides direct co-benefits for the **Energy** dimension and reduces energy poverty. The indirect co-benefits of clean energy are wide-ranging and particularly strong for **Health** and **Income and Work** (Figure 4.4).



Figure 4.4 — Human well-being benefits of climate solutions for Adopting Clean Cooking

This systems diagram highlights the direct (thick gray line) and indirect (thinner gray line) co-benefits of *Adopting Clean Cooking* for the 12 dimensions of human well-being. Ripple effects are shown by dashed gray lines. Local stakeholders, governments, and change makers are implicit along the connecting arrows. Potential CO_2 -eq reduction for 2020 to 2050 is 86 gigatons.

Health: As exposure to HAP remains pervasive, reducing exposure through the promotion of less-polluting stoves and fuels may be a population-wide strategy to improve global health. Review studies and global analyses have shown that clean cooking can have positive impacts on adults, including a decline in respiratory and non-respiratory symptoms, chronic obstructive pulmonary disease, cough, conjunctivitis, and blood pressure.^{218,219,220,221,222} Switching to clean cooking could prevent more than 22.5 million premature deaths between 2000 and 2100.²¹¹

Though the evidence for co-benefits of clean cooking in children is mixed, trials from Nigeria have shown that mothers cooking with clean fuel had babies with higher birth weight and lower prenatal mortality than households cooking with fuelwood and kerosene.²²³ A study in Guatemala found that reduced exposure to wood smoke from cooking— achieved with a chimney stove—significantly reduced cases of severe childhood pneumonia.²²⁴ However, large studies in Ghana, Kenya, and Malawi suggest that the provision of a clean stove or fuel did not result in significant reductions in pneumonia or improved birth weight.^{225,226,227} This result was likely attributable to lack of uptake or adoption of the new stove technology, continued use of solid fuel stoves alongside clean fuel stoves, and the contributions of other local and regional air pollution sources.²²⁸

Shifting to clean stove stacking—a suite of electric, gas, and biomass options—instead of relying on just one type of stove for all cooking needs can help meet the diverse energy needs of a household while also reducing HAP.²⁰⁰

Income and Work: Access to clean cooking is hypothesized to provide households—and women in particular—additional time for income-generating activities.^{205, 229, 230, 231} Studies from several SSA countries have shown that households using or switching to clean cooking fuel spent the most of their time savings in economic activities.^{203, 204, 232, 233} A recent analysis from six LMICs showed that improved cookstoves save households an average of 34 minutes per day. The time savings were consistent across both rural and urban areas, were greatest for fuel collection and preparation, accrued to both women and men, and were highest for more advanced technologies and fuels (e.g., electricity, LPG, and biogas).²³⁴

Education: The time burden of fuel collection on children negatively impacts their school attendance.^{235, 236, 237} Field studies in Ethiopia, Ghana, and Kenya have demonstrated that children's participation in collecting resources (such as wood and water) impedes school attendance.^{238, 239, 240} The gendered nature of fuelwood gathering^{241, 242, 243} disproportionately impacts girls' education.^{203, 244} With rapid deforestation in many countries, the time burden has increased further, leading to an increase in girls' exposure to gender-based attacks or injury while collecting fuel resources, potentially further diminishing their educational opportunities.^{203, 204, 245}

Relatively few studies directly examine how access to clean cooking affects school attendance, although limited evidence suggests that the impacts are largely positive. Households in South Asia that used improved cooking fuel enrolled their children in school at higher rates.²³⁰ In Bangladesh, 18 percent of the time saved from using biogas for cooking was spent on education.²³²

Gender Equality: Besides negative health impacts, the time burden—which reflects cultural definitions of "gender-appropriate" work that disadvantages girls and women—associated with fuel collection leads to women's time poverty (a measure of women's subordinate status within the household and wider social arenas) that reduces time available for education, economic activities, childcare, and leisure. Improved cooking contributes to gender equality primarily through improvements in women's health and boosting opportunities to pursue economic activities and higher school enrollment by reducing time poverty.

Clean cooking also reduces risks of gender-based violence (GBV). Fuelwood collection increases the risk that women and girls will be exposed to GBV. The risks are particularly high in countries experiencing political instability and for those living in refugee camps. Reports of sexual attacks on women in refugee camps in Ethiopia, Kenya, Tanzania, and Uganda are widespread.²⁴⁶ In Kenya's Dadaab refugee camp, more than 90 percent of GBV cases occurred when women ventured outside the camps in search of firewood or other sustenance.²⁰⁴ GBV and fuelwood collection has been studied to a lesser extent²⁰⁴ in South Asia. However, GBV related to resource collection and hygiene is pervasive; women in low-income resettlement areas in India reported near daily harassment while collecting water and using sanitation facilities.²⁴⁷

Culturally grounded approaches are needed in which roles are redefined through social change in ways that could help leverage the potential of clean cooking to truly benefit women (M. Greene, personal communication, January 21, 2022). Research that doesn't critique or promote gender myths about women and their relationship with energy and clean cooking is also necessary.²⁴⁸ For best results for both positive climate impacts and HWB, we must complement advances in technology with a more equitable and just configuration of male and female roles (M. Greene, personal communication, January 21, 2022).

4.4 Climate solutions for providing clean electricity

Meeting basic energy needs (e.g., ensuring all households have access to electricity) and meeting energy requirements for broad, large-scale economic development are separate but related challenges. Modeling estimates show that to provide universal access to electricity by 2030, more than half of households that currently do not have electricity access would need to access mini-grid and off-grid technologies powered largely by solar photovoltaics.²⁴⁹ Most energy-related GHG emissions are and will continue to be driven by consumption in places that largely do not have energy access problems, even as those economies shift to clean electricity. However, that doesn't preclude clean electricity from being a major part of the equation to meet needs at both the household and country level in LMICs. This report centers on the household level, as we are focused on getting electricity access for basic well-being to the approximately 750 million²⁵⁰ who currently do not have it, particularly those living in rural areas.

Electricity and HWB are inextricably linked. A recent study²⁵¹ of linkages between multidimensional poverty and electricity found that people in countries with limited access to electricity also suffer from high food insecurity, lack access to improved water and sanitation, experience low income and work, and endure poor health (Figure 4.5).





Sub-Saharan African countries that have low levels of access to electricity also tend to have low values in other dimensions of the Drawdown Lift Human Well-Being Index.

Off-grid distributed clean electricity could contribute to transforming the well-being of remote, under-resourced, vulnerable communities.^{252, 253} In Bangladesh, which has championed off-grid and microgrid solar photovoltaics (SPV), 20 million people have gained access to electricity in the last few years, changing their lives in multiple positive ways.²⁵⁴

Grid-based clean geothermal energy in East Africa's Rift Valley could address rural and urban energy needs. However, most countries in the region, other than Kenya, have not yet pursued geothermal extensively due to lack of finance, knowledge, and workforce.²⁵⁵ Kenya has increased its geothermal electricity production dramatically over the last five years; in 2019, geothermal contributed 47 percent of all electricity consumed in the country, and this number has been rising since then, although challenges related to unequal access in highly vulnerable rural areas remain.²⁵⁶

Minimizing electricity poverty is critical to improve HWB in SSA, where around 570 million people lack access to electricity.²⁵⁰ A recent study²⁵⁷ that analyzed current trends in electricity generation in Africa projected that less than ten percent of the total electricity generated by 2030 will come from solar, wind, or geothermal energy. However, according to the International Renewable Energy Agency (IRENA), appropriate government policies, regulation, and access to finance could allow SSA to meet more than 67 percent of its electricity needs from renewable sources by 2050 while also generating 2 million green jobs.²⁵⁸ And, solar mini-grids are more cost-effective than fossil fuel–based mini-grids (Figure 4.6).



Figure 4.6 — Competitiveness of solar- and diesel-based mini-grids

Orange, yellow, and light blue represent regions where solar photovoltaics-based mini-grids are more cost-effective (cheaper) than diesel-based mini-grids. kWh: kilowatt-hour *Source: modified from Szabo et al. 2021.*

As presented below, several studies and reports have highlighted the positive impacts of renewable energy, especially SPV, on **Income, Networks, Education,** and **Gender Equality.** However, the number of studies rigorously analyzing the human and environmental co-benefits of these interventions are still limited,²¹² primarily because widespread implementation of off-grid and microgrid networks is fairly recent. While more research is needed to evaluate whether solar technologies can deliver services beyond basic lighting, charging, and entertainment (such as powering a refrigerator),²¹² there is evidence that basic access to clean energy provides multiple HWB co-benefits, which we detail below.

Five Project Drawdown electricity solutions—*Distributed Solar Photovoltaics, Geothermal Power, Micro Wind Turbines, Microgrids,* and *Small Hydropower*—can rapidly increase clean energy access to large populations that are underserved by grids or cannot afford or access grid connections. Except for *Geothermal Power*, which can address both rural and urban power needs, these solutions address the basic household electricity needs of largely rural people who lack access to electricity.

The majority of the evidence presented here is from studies and analyses conducted on off-grid and microgrid SPV, as they are more widely applicable and available than micro wind turbines and small hydropower, they have greater emissions reduction potential, and there is more evidence of co-benefits for HWB. However, the HWB co-benefits of the individual solutions in the *Providing Clean Energy* solution group should be similar as they all provide clean energy and reduce energy poverty.

Human well-being benefits of clean electricity

Access to clean electricity fundamentally addresses energy poverty and provides direct cobenefits for the **Energy** dimension and through it, all other dimensions of HWB (Figure 4.7). While electricity—generated either by renewable or nonrenewable sources—is foundational for many aspects of HWB, it is crucial that households that are currently not connected to the grid be able to leapfrog to electricity generated by clean renewable sources. Off-grid and microgrid electricity would not only allow households to reap the benefits of electricity sooner, but they

are also cheaper and, unlike fossil-fuel based energy, they are less polluting and don't have negative health effects.

Electricity is foundational for many aspects of HWB; it contributes to job creation and supports education, increases income, reduces food insecurity, and improves access to clean water, sanitation, and health.^{259.} ²⁶⁰ It also contributes to opening doors that advance women's economic empowerment²⁶¹ and builds resilience that promotes peace and justice by minimizing risks of conflict and largescale migration.^{262, 263}



Access to clean water is a basic human right and can be advanced through access to electricity. *Credit: Arne Hoel | World Bank*





This systems diagram highlights the direct (thick gray line) and indirect (thinner gray line) co-benefits of *Providing Clean Electricity* for the 12 dimensions of human well-being. Ripple effects are shown by dashed gray lines. Local stakeholders, governments, and change makers are implicit along the connecting arrows. Potential CO_2 -eq reduction for 2020 to 2050 is 77 gigatons.

Rural electrification positively impacts women's health and well-being; research in Africa shows that electrifying clinics to provide lighting and refrigeration for medicines improves maternal health, and public lighting enhances safety for all—particularly women and girls.²⁶⁴ In India and Indonesia, evidence links more TV watching through new television access with smaller family size, a decrease in domestic violence, and increased female autonomy.²⁶⁴ Although the mechanism through which electricity access in rural areas may impact gender relations is unclear, data support a positive correlation with women's welfare.²⁶⁴

Income and Work: Access to electricity leads to higher income, particularly for women, primarily through increased working hours.^{265, 266, 267, 268} An analysis of the economic benefits of SPV from more than 2,300 off-grid SPV users from Kenya, Mozambique, Rwanda, Tanzania, and Uganda reported that 69 percent of SPV customers increased their incomes through longer working hours or by obtaining new jobs, and 33 percent of SPV owners' incomes increased by US\$420 per year.²⁵² Additionally, most households decreased their energy expenditures for lighting by switching from kerosene to solar lamps.²⁵² Some studies, however, have suggested that the benefit of SPV for income, if any, is limited.^{269, 270, 271}

Investments in decentralized and clean energy technologies can also create jobs. Decentralized renewable energy already employs more people than traditional energy in India, Kenya, and Nigeria.²⁷² A meta-analysis of more than 80 studies found that more granular technologies (such as *Distributed Solar Photovoltaics* and *Micro Wind Turbines*) generate more jobs over their lifetimes²⁷³ than centralized technologies. Off-grid renewable energy could generate 4.5 million jobs globally by 2030.²⁷⁴ For example, in Nigeria, the government-led Solar Power Naija program is projected to incentivize the creation of nearly 250,000 jobs in solar manufacturing

and assembly; in Burkina Faso, renewable energy equipment is being manufactured for domestic SPV use.²⁷⁵ Large-scale implementation of Project Drawdown solutions that address energy poverty could significantly increase the incomes of households close to or below the poverty line, while creating millions of jobs in LMICs.

Networks: Off-grid SPV plays a vital role in increasing connectivity,²⁶⁶ which contributes to increased job opportunities and income. In India, better connectivity resulting from solar-powered mobile phone chargers increased income for the users.²⁶⁷ In Kenya, Mozambique, Rwanda, Tanzania, and Uganda, 89 percent of SPV users reported increased mobile phone usage, implying improved connectivity.²⁵²



About 750 million people, living mainly in rural areas of low- and middle-income countries, don't have access to electricity. Electrifying schools has led to increases in time studying, years of education, school enrollment, and digital literacy. *Credit: Chetan Soni | United Nations*

SPV-powered devices have played an important role in receiving early weather forecasts that contributed to saving thousands of lives; this is particularly important for countries such as Bangladesh that are prone to natural hazards.^{268, 269, 276, 277} SPV has also contributed to increased access to TV and radio, resulting in increased awareness and positive behavior change around health, including family planning, sanitation, diarrhea, hygiene, and nutrition.^{276, 278, 279, 280, 281} Penetration of mobile phones has led to an increase in financial inclusion for many low-income households. In Kenya, mobile finance boosted inclusion in the formal economic system from 26 percent in 2006 to 87 percent in 2019,²⁸² enhancing access to finance for education, business, and housing.

Education: Electrification of schools that are off-grid has led to increases in children's study time, years of schooling, and school enrollment.^{277,283,284,285,286} Teachers are more willing to relocate to schools that are electrified.²⁸³ SPV has played a crucial role in the computer education and training that is indispensable for many jobs; in several countries, digital literacy in rural schools is gained exclusively through use of solar-powered computers. School electrification can boost educational outcomes in LMICs and build skilled worker capacity to address future labor needs.

In India, studies suggest that access to high-quality light facilitates studying and reading for children as well as adults.^{283, 287} However, the effect on educational quality outcomes is unclear. Whereas some studies have argued that SPV leads to better education outcomes, such as higher grades,²⁸⁸ others have reported no improvement in performance due to the use of SPV-powered solar lanterns on national exams or test scores.^{289, 290}

A major boost to education comes from the spillover effects of increased income due to SPV. Households in Kenya, Mozambique, Rwanda, Tanzania, and Uganda that experienced increases in income from SPV used around 25 percent of their additional income on education.²⁵² Students from low-income households with unstable income are more likely to drop out of school and less likely to graduate from secondary school.²⁹¹



Solar-powered lamps enhance opportunities to study and work after sunset. They also reduce reliance on kerosene lamps, which pollute the indoor air and require purchase of fuel. *Credit: Solar Sister*

The COVID-19 pandemic has further highlighted opportunities to provide clean electricity to schools and households as part of a low-carbon pathway to development that also addresses multiple aspects of poverty alleviation, particularly energy poverty. Schoolchildren in LMICs suffer disproportionately due to lack of digital connectivity; in LMICs, less than 6 percent of children have Internet access at home, and only 10 percent of schools in South Asia have Internet.²⁹² Further, the effects of remote learning during the pandemic have not been equal across genders; for example, gendered demands on girls' time in East Africa has been detrimental to their remote learning.²⁹³

Health: SPV contributes to indirect human health co-benefits, such as improving pulmonary function and life expectancy due to reduced HAP. While kerosene lamps contribute less to HAP than cooking, replacing them with solar lamps results in fewer lung infections and diseases.^{284, 294, 295, 296, 297} Adoption of solar lamps has also greatly reduced incidences of burns from kerosene lamps in LMICs that disproportionately affect adolescents' health.^{298, 299, 300, 301, 302}



Access to electricity in health-care facilities improves the ability to provide services at night and store medicines properly and is linked to reduced maternal mortality. *Credit: Jake Lyell | Alamy Stock Photo*

Off-grid SPV can greatly enhance access to quality, modern health services in remote areas by providing services at night, enhancing access to equipment that runs on electricity, increasing retention of qualified staff, and ensuring proper storage of vaccines and medications.³⁰³ Such benefits reduce maternal and child mortality rates in LMICs³⁰⁴ and improve overall health. Several studies report significant decreases in maternal mortality rates, more deliveries in health centers per month, and better outpatient services due to SPV.^{250, 303, 305, 306}

Evidence regarding the effect of SPV for mental health benefits is limited. However, increases in entertainment options,^{265, 276, 277} social interactions, and standards of living^{265, 267} can contribute to improved mental health.

Water and Sanitation: Clean electricity has played a vital role in providing clean and reliable water to communities in South Asia and SSA.^{307, 308} It has also contributed to improved sanitation in rural communities by providing continuous access to clean water and light.^{309, 310} Further, water pumps powered by clean energy have lower operational costs than those powered by diesel.^{311, 312}

Clean electricity can play a role in providing clean water and improving sanitation facilities in rural communities, which could contribute to reducing the incidence of diarrhea and waterborne diseases that together are the third leading cause of death among children under age 5 globally.³¹³ Two-thirds of people in SSA rely on contaminated water sources.³¹⁴ Inadequate access to WASH accounts for 10 percent of the global disease burden.³¹⁵

Food: A study across 54 LMICs found that access to electricity provides immediate benefits for both the production and utilization (preparation and storage) of food and is recommended to improve food security through household-level off-grid electricity investments.³¹⁶ Clean electricity boosts opportunities to implement water-efficient technologies such as solar-powered drip irrigation,¹²⁵ improves water governance, allows rural farmers to charge mobile phones that they use to boost agriculture productivity,³¹⁷ and reduces food waste.³¹⁸ Irrigation through solar water pumps can improve household nutritional intake and dietary diversity by increasing agriculture yield and household income for smallholders and facilitating livestock watering.^{319, 320, 321, 322, 323, 324}

In Ethiopia, households that irrigate their lands are more likely than non-irrigated ones to meet minimum dietary requirements for women.^{122, 325} For an extremely under-resourced community in Benin (living on less than US\$1.25 per day), solar-powered drip irrigation in an unelectrified area increased income, food security, and micronutrient uptake.¹²⁵ However, other research suggests that the effect of irrigation on nutritional outcomes is context-specific and that positive outcomes are not always observed.^{122, 326} More rigorous analysis is needed to evaluate the linkages between solar-powered irrigation and malnutrition and undernourishment within the context of extension agent support (e.g., so farmers can diversify into horticulture with higher water requirements).



Solar-powered pumps provide access to clean water and enable efficient use of water for irrigation to increase food production. *Credit: Joerg Boethling | Alamy Stock Photo*

Only about 5–6 percent of SSA cropland is irrigated.³²⁷ The region frequently suffers from food insecurity³²⁸ and accounts for the majority of the undernourished and malnourished population globally.³²⁹ When accompanied by supply chain enhancement, extension and market support, and effective water governance and infrastructure maintenance, the application of renewable energy (powered by properly maintained solar water pumps) to irrigate millions of hectares of land in SSA³²⁸ could be crucial to boost nutrition, food security, income, and employment²²⁹ and would contribute to developing sustainable and resilient food systems, increasing income, and generating employment.^{283, 319, 328, 330}

Gender Equality: By reducing women's household chores and extending useful hours in the day, clean electricity contributes to improving women's income and total hours of paid work, reducing time poverty and increasing participation in non-household work.^{264,268,277,331,332} For example, in Tanzania, solar lamp ownership led to an increase in female labor supply and income;³³³ women-inclusive solar lightning programs have been shown to positively impact household health, education, savings, and women's economic productivity and empowerment.²⁶¹ In Bangladesh, women have been shown to use the extra time for resting and entertainment.^{268,285} Studies have reported that SPV leads to an increase in women's household decision-making power.³³⁴

SPV has provided clean water³⁰⁷ and improved sanitation in schools. Some studies have shown that water and sanitation can reduce dropout rates and absenteeism among girls^{335,336,337} while others have reported limited links between girls' absenteeism and school-based water and sanitation facilities.³³⁸

FEATURE BOX Solar Sister: Improving gender equality, income, health, and education by reducing energy poverty

Solar Sister is a women-centered network of entrepreneurs that brings off-grid solar electricity and cleaner cooking to underserved communities in SSA. The initiative was launched in Uganda in 2010 and has since expanded to Tanzania and Nigeria. It recruits and trains new entrepreneurs to sell durable, affordable solar-powered products (e.g., lights and phone chargers) and cleaner cookstoves to households without access to electricity or clean cooking. It has resulted in co-benefits for multiple HWB dimensions, including improved children's educational outcomes, health, and financial well-being. Household members can work more hours due to the addition of evening light indoors, or start new businesses using the money saved by not buying kerosene. Networking has also improved, as the solar batteries are used to charge mobile phones. Finally, Solar Sister has supported women to become entrepreneurs and leaders and has helped many feel more respected in their households and community.³³⁹



The Solar Sister program generates multiple human well-benefits for women through the provision of solar power and clean cooking.



4.5 Climate solutions for fostering equality

High-quality education is linked to increases in income, participation in public service, and improved health. Credit: United Nations Children's Fund (UNICEF) Ethiopia

In SSA, about 30 percent (97.5 million) and in South Asia about 20 percent (93 million) of children and adolescents do not attend primary or secondary school.³⁴⁰ Three-quarters of children who never enter primary school are girls from rural, under-resourced communities;³⁴¹ according to UNICEF, "44% of girls and 34% of boys (10–19 years old) from the poorest families have never attended school or dropped out before completing primary education."³⁴² In LMICs, 218 million women of reproductive age want to avoid becoming pregnant but do not use a modern method of contraception.^{343, 344, 345}

Rights-based, voluntary family planning and quality universal education for all are essential human rights that can also generate cascading benefits for climate change by impacting population growth over time.^{53,346,347,348,349,350} In addition, the benefits of contraception and high-quality universal education are well-established for many aspects of HWB, including maternal and child health, nutrition, economic development, gender equality, and resilience,^{39,351,352,353,354,355,356} and contribute to overall planetary health.³⁴⁹ Men, too, benefit from the opportunity to regulate their own fertility and play an important role in deciding with their partners whether and when to have children.³⁵⁷

Family planning services, educational attainment, and fertility are closely linked,³⁴⁸ although additional research is needed on the causal relationship between education and sexual and reproductive health outcomes.³⁵⁸ Countries in SSA with low female literacy and low demand for family planning satisfied with modern methods are also highly vulnerable to climate change.^{21, 359} Climate shocks and stressors often exacerbate gendered responsibilities such as performing household chores (e.g., fetching water and fuel sources, cooking, and sibling care), which can further impact girls' ability to remain in school.²³⁷

Family planning—ensuring everyone's individual contraceptive needs are met in a rights-based manner that centers bodily autonomy—is not in itself a climate mitigation strategy. Rather, it is the outcome of family planning, slower population growth, that is the climate mitigation strategy. Demography and climate change are linked;^{347, 360, 361} slowing future population growth could translate into a 15 percent reduction in global CO₂ emissions by 2050 due to reduced demand for resources.³⁴⁶ Project Drawdown estimates that meeting all needs for voluntary family

planning and providing universal education for all boys and girls between 2020 and 2050 could lead to a reduction of almost 70 Gt of GHG emissions at a global level due to slower population growth rates.³⁵⁰

Project Drawdown's *Health and Education^{xix}* solution centers around actions that foster equality in society—specifically, rights-based, voluntary family planning and high-quality, inclusive education— and offer clear health and economic as well as long-term ancillary climate benefits (Figure 4.8).





Equal access to high-quality education for girls and boys and rights-based voluntary family planning contribute to enhanced well-being in low- and middle-income countries.

Human well-being benefits of fostering equality

Rights-based family planning and improving access to and quality of universal education can boost HWB at the individual and household level, particularly for rural, under-resourced populations in LMICs where access to a wide range of contraception and high-quality education has typically been low (Figure 4.9). When people everywhere can make decisions about their fertility through the realization of universal access to rights-based sexual and reproductive health services, including voluntary family planning, the end result of slower global population growth can contribute to a long-term reduction in emissions.^{346, 349}

xix It is important to note that Project Drawdown's *Health and Education* solution focuses only on 1) high-quality, voluntary family planning (no additional aspects of health) and 2) 12 years of universal, quality education for all children.



Figure 4.9 — Human well-being benefits of climate solutions for Fostering Equality

This systems diagram highlights the direct (thick gray line) and indirect (thinner gray line) co-benefits of *Fostering Equality* for the 12 dimensions of human well-being. Ripple effects are shown by dashed gray lines. Local stakeholders, governments, and change makers are implicit along the connecting arrows. Potential CO₂-eq reduction for 2020 to 2050 is 69 gigatons.

Health:^{xx, xxi} Voluntary family planning programs—when accompanied by cultural shifts that give women full bodily autonomy—help women and couples realize their desired family size. When population growth rates slow, governments are better able to keep pace with population-driven needs—such as building and staffing health clinics and schools and staffing extension agents to meet rural populations' food security and natural resources needs.

The ability to plan and space births yields numerous benefits for mothers and children, notably decreased morbidity and mortality. Some 99 percent of all maternal deaths occur in LMICs, 66 percent in SSA;^{363, 364} a recent study estimated that meeting the contraceptive and maternal care needs of women in LMICs could prevent nearly three-quarters of maternal deaths and dramatically decrease newborn mortality.³⁵⁵ Children born four years after a previous birth are less likely to be underweight and stunted than children born after a two-year interval.³⁴³ Family planning decreases vulnerability to environmental shocks and stressors such as flooding, drought, and food and water scarcity, all of which have adverse effects on women's health and are becoming more frequent due to climate change.^{347, 356}

Educating girls, particularly through secondary school, is one of the most powerful ways to address gender inequality and long-term poverty. Increases in educational attainment globally among women of reproductive age led to 4.2 million fewer deaths among children younger than 5 years from 1970 to 2009.³⁵¹ In fact, the relationship between the length of a mother's schooling

xx There are myriad non-family-planning-related linkages between health and climate change; see, for example,⁷³ and follow the Pathfinder Initiative³⁶² on health and climate.

xxi Health is both an HWB dimension and part of a Project Drawdown climate solution, *Health and Education* (see also <u>Footnote X</u> on gender). As such, in this report, the **Health** dimension is an indirect co-benefit of other Project Drawdown climate solutions in the solution groups *Providing Clean Electricity* and *Adopting Clean Cooking*. *Health and Education* is a Project Drawdown solution with direct co-benefits to the Food, Income and Work, Gender Equality, Social Equity, Education, and Health dimensions.

and child survival is linear, with no threshold.^{351, 365} Researchers found that each additional year of secondary education was associated with decreases in HIV prevalence among all adolescents, but especially young women, particularly in South Asia, Latin America, and SSA.^{354, 366, 367, 368}

Contraception enables sexually active adolescents to remain in school and pursue higher education, with sustained lifelong benefits. In LMICs, the unmet need for modern contraception among almost half of the 32 million adolescents ages 15–19 contributes to about 10 million unintended pregnancies each year.³⁴⁵ The longer girls stay in school, on average, the longer they delay transitions such as childbearing.³⁶⁹ Ensuring pregnant adolescents have full access to high-quality adolescent-friendly reproductive health services also helps them continue their education through and beyond secondary school. Should girls choose to become mothers, research has shown that mothers' schooling has a deeper positive impact on children than fathers' schooling.^{348, 365, 370, 371}

Education:^{xxii} Formal education is strongly associated with improving livelihoods, economic development, social progress, and individual and household well-being. Education has been shown to lead to decent jobs and freedom from poverty,^{372, 373} adoption of healthier water and sanitation practices and clean cooking, participation in public life, and better health decisions.^{374, 375, 376} Increasing climate literacy is a key tool for engaging in climate action and contributing to an enabling environment for mitigation, adaptation, and low-carbon pathways to development.³⁷⁷



Universal education and rights-based voluntary family planning provide benefits across the lifespan, including reduced maternal and child mortality; the outcomes of family planning contribute to slowing population growth at a global level. *Credit: Yagazie Emezi | Getty Images | Images of Empowerment*

xxii Education is both one of the HWB dimensions and part of a Project Drawdown climate solution, Health and Education. As such, in this report, the Education dimension is an indirect co-benefit of other Project Drawdown climate solutions, such as the set of solutions Providing Clean Electricity and Adopting Clean Cooking. Health and Education is a Project Drawdown solution with direct benefits to the Health, Food, Income and Work, Gender Equality, Social Equity, and Education dimensions.

Education is a fundamental human right for both girls and boys. Moreover, girls' education plays an important role in building adaptive capacity and reducing vulnerability to climate-related extreme events.^{378, 379, 380} Educated women can better protect themselves and their families from environmental shocks and are able to participate more fully in decision-making.²³⁷ Today, 129 million girls are out of school worldwide.³⁸¹ The COVID-19 pandemic has caused a substantive disruption in education, and projections suggest that as many as 11 million girls might not return to school.³⁸² This deprivation can hinder the ability to build long-term community resilience to climate change, enhance health, improve gender equality and social equity, eradicate poverty, and foster peace and justice.

Income and Work: Education and access to modern contraception provide women the opportunity to pursue and continue secondary and higher education and engage in formal and informal work. In recent years, this has become increasingly important in semi-arid regions of Africa and Asia, where the aspirations of many youth are shifting from the pastoral or agricultural lifestyles of their parents to education and work in the formal sector.³⁸³ Opportunities are rising around climate-specific jobs in SSA and South Asia; globally, one-third of renewable energy sector employees are women, and plans are underway to increase gender parity.²⁷⁵ The transition to a green economy in Africa could provide millions of jobs for women if barriers are addressed—namely, access to land and finance, gender discrimination in the education system and labor market, laws that restrict women's access to certain tasks and jobs, and inequalities reflecting social norms around unpaid care work, which largely fall on women.³⁸⁴

With better health and the ability to avoid unplanned pregnancies and lower childcare needs, women have more time and energy for other opportunities such as taking part in the labor force or building businesses.^{347, 385} Improved reproductive health due to contraceptive uptake and girls' educational attainment, particularly at the secondary level, are closely linked.^{348, 371} An educated woman is better able to translate her fertility preferences into practice with the use of contraception, which opens up income-based opportunities outside the home. In Bangladesh, women with access to contraception have fewer children and higher household incomes than those without.³⁸⁶

Food: Family planning and education help meet fundamental human needs for high-quality, culturally appropriate, healthy food. Family planning can influence the nutritional outcomes of infants and young children; when births are spaced and planned, mothers are healthier and often have more resources and time to breastfeed.³⁶⁴ In addition, women who use or have better access to modern family planning are more likely to participate in household decision-making around nutrition and food security.^{71, 364}

Higher educational attainment for household heads, particularly women, has been associated with beneficial nutrition practices that prevent childhood malnutrition.³⁸⁷ Schools play a vital role in providing nutritious meals to children and in building capacity and knowledge for better food production and consumption.⁷⁷ Children who attended school and had higher literacy levels experienced lower levels of stunting and low weights.³⁸⁸

In Bangladesh and India, research has shown that child malnutrition and undernourishment is significantly related to a mother's education; children with a mother who is illiterate or undereducated are more likely to suffer from malnourishment, stunting, and wasting.^{389, 390} Similar findings have been reported from Nigeria and Kenya.^{391, 392}

Gender Equality: High-quality education—particularly for girls—and family planning help address underlying drivers of gender inequality. However, tackling patriarchy, the root cause of gender inequality, is essential to shift gender values and open opportunities for women and girls.³⁹³ Low education levels limit economic opportunity and prolong social inequalities. Education and knowledge building are important for girls to become active community members and leaders³⁹⁴ and involved in political, social, and economic spheres of life.

Breaking patriarchal structures and enabling girls to go to school multiplies their economic choices, and increases their voice and influence in society.

GRAÇA MACHEL, FIRST LADY OF MOZAMBIQUE AND SOUTH AFRICA, AND PRIME MINISTER ERNA SOLBERG, NORWAY³³⁸

Access to high-quality rights-based family planning fundamentally advances human rights.^{71, 395} Access to safe abortion services and post-abortion care are included in WHO's list of essential health-care services; the lack of safe abortion services, which are still illegal in many countries, leads to an increase in unsafe abortions and associated morbidities.^{355, 396} Almost 50 percent of all abortions globally are unsafe; 97 percent of those are in LMICs.³⁹⁷

Child, early, and forced marriage and unions are driven by many intersecting factors, including social norms, economic insecurity, lack of education, and the absence of legal frameworks.^{398, 399, 400,401,402} They are a major societal issue with impacts on gender equality: As of 2018, globally more than 650 million girls and women—primarily in SSA and South Asia—were married before they turned 18.⁴⁰³ Early marriage has several adverse impacts on girls' education, social life, health, economic opportunities, and overall well-being.⁴⁰¹ Studies have shown that overcoming household poverty and providing secondary and higher education to girls significantly reduces their risk of child marriage,^{404, 405, 406} while primary education alone is insufficient to reduce the practice.⁴⁰⁷ Early marriage followed by early pregnancy and subsequent termination of a girl's education can contribute to a vicious cycle of intergenerational poverty and vulnerability.³⁴¹

About one in three women worldwide experience GBV, often from an intimate partner.³⁵⁵ According to Castañeda Camey et al., "Those who have experienced GBV may be discouraged from engaging in decision making, leadership, education and income generating opportunities all of which directly affects economic empowerment and jeopardizes progress toward poverty reduction."²⁴⁷ Research from Niger and India has found links between reproductive coercion and higher rates of GBV against adolescents and women, respectively.^{408, 409} Addressing GBV is complex and multifaceted, requiring gender-transformative programming accompanied by an enabling policy environment to address power, social norms, and other drivers of inequality and violence.

FEATURE BOX

Tuungane Project, Tanzania: Creating healthier and more resilient families, fisheries, and forests

In western Tanzania, Lake Tanganyika and the adjacent forested mountains are home to 250 fish species found nowhere else and more than 90 percent of Tanzania's endangered chimpanzees. More than 500,000 people depend on farming and fishing within the 2 million-hectare landscape. Yet they face challenges around health and government services due to their remote location and additional challenges as the climate changes. The Tuungane Project, launched in 2012 by The Nature Conservancy and Pathfinder International, collaborates with local people, leaders, and government to develop community-led projects that conserve biodiversity-including habitats for forest reserves and grazing lands protected by eight communities—by generating revenue from the sale of carbon credits and protecting vital endangered chimpanzee habitats. Performance-based payments generate income for remote villages, transforming how local people and the government value forests. Conservation-smart agricultural practices adopted by more than 12,000 farmers have doubled maize yields, reduced pressure on chimpanzee habitats, and improved water quality in the lake. Newly community-managed fish reserves and fisher-enforced sustainable fisheries laws help ensure continued food and income. More than 50 community conservation banks have provided small loans to 1,700 people to start businesses and diversify income. The project used additional funding to build a dormitory for girls attending secondary school, reducing the dropout rate to zero and increasing the percentage of girls applying to colleges and vocational schools. Health clinics have been upgraded with solar suitcases that

can provide electricity during nighttime births, dramatically reducing maternal and neonatal deaths. Reproductive health outreach and services reached more than 55,000 community members in 2020. Nearly 6,000 families volunteer to educate by example by modeling healthy and sustainable behaviors, such as using hand-washing stations, installing upgraded latrines, prioritizing voluntary family planning, using energy-saving stoves, and setting agricultural plots away from the lake (Sources: The Nature Conservancy Tuungane Project,410 Craig Leisher, personal communication, January 13, 2022).



Projects like Tuungane integrate solutions for improving people's livelihoods and health and protecting and managing the natural resources they rely on. *Credit: Ami Vitale*

4.6 Additional indirect co-benefits and ripple effects on human well-being dimensions of Project Drawdown climate solutions

The five groups of climate solutions generate direct and indirect co-benefits and ripple effects for all HWI dimensions. As depicted in the systems diagrams (Figures 4.1, 4.3, 4.4, 4.7, 4.9) for each solution group, direct co-benefits flow to specific indirect co-benefits, and indirect co-benefits flow to specific ripple effects.

In this subsection, we detail the evidence around indirect co-benefits for the *Improving Agriculture and Agroforestry, Protecting and Restoring Ecosystems,* and *Fostering Equality* solution groups. The indirect co-benefits for the *Adopting Clean Cooking* and *Providing Clean Electricity* solution groups were covered in their respective subsections, above.

In addition, we present the evidence for the ripple effects that the solution groups generate for **Networks, Housing, Gender Equality, Social Equity, Peace and Justice,** and **Political Voice.**

As can be seen in the systems diagrams for the five solution groups, certain HWB dimensions generate many indirect co-benefits and ripple effects. **Income and Work** and **Education** generate the most indirect and ripple effects across all solution groups. These two HWB dimensions not only reinforce each other—e.g., increase in income leads to better education access and outcomes and vice versa^{411, 412, 413}—but can also lead to indirect co-benefits and ripple effects on multiple other dimensions including **Energy, Health, Networks, Housing, Social Equity, Peace and Justice,** and **Political Voice** in various solution groups.

For instance, the co-benefits of **Education** on **Health** are well-established: Increased educational attainment has been linked with reduction in child mortality and improvement in public health, child health, and health equity.^{351, 414, 415} The high rate of poor mental health among food insecure people in Africa⁴¹⁶ could be alleviated in part through *Improving Agriculture and Agroforestry* and other climate solutions that improve **Food** security. Suicide rates among farmers are high in several LMICs. While the causes driving rural people's, (including farmer's) suicide are complex, a study in India has linked it with socioeconomic factors as well as reduced yield from extreme weather events.⁴¹⁷ Increase in **Income and Work** and better **Food** security can potentially contribute to a reduction in suicide rates.

Here we present evidence of how these seven dimensions can generate co-benefits through ripple effects created by climate solutions when they are folded into climate-smart development projects and policies. While development pathways are complex, interrelated, and regionally specific, the ripple effects of climate solutions can contribute to resilience, promote gender equality, diminish inequality, and lift up the most vulnerable who are disproportionately impacted by climate-conflict shocks.⁴¹⁸

Energy: Although access to clean energy depends on multiple factors, including government policies, supply and demand, and social factors, affordability is a major roadblock; household income and consumption of clean energy are closely intertwined. In SSA, high electric cost is a major hurdle for households that are within the reach of an electric grid to get connected to the grid.⁴¹⁹ Even for off-grid decentralized solar home systems, the upfront cost continues to be a major barrier for the lowest income households.⁴²⁰

Government policies that focus on educating households regarding the benefits of clean energy and supporting affordable finance of off-grid electricity and clean cooking could allow households to afford clean energy for basic needs. Increases in **Income and Work** from *Improving Agriculture and Agroforestry, Protecting and Restoring Ecosystems* and *Fostering Equality* solutions can contribute to a reduction in energy poverty for rural farming households in LMICs. Income and education can also be important determinants of cooking fuel use in urban and rural areas; households with lower education level and income more often rely on wood, biomass, and charcoal fuel for cooking, according to studies conducted in northern Sudan, Ghana, and Afghanistan.^{421, 422, 423} In addition, boosting access to clean electricity in homes can also help address heat-stress inequality in LMICs by increasing electricity access while not adding pressure to fossil fuel–driven power grids.¹⁹¹

Housing: Income and education play critical roles in access to improved housing. A large-scale study from SSA showed that the prevalence of improved housing doubled (from 11 percent to 23 percent) between 2000 and 2015.⁴²⁴ The analysis suggested strong links among improved housing, education, and wealth. The odds ratio of living in improved housing when the head of household was educated was 1.8 and for wealthy households was 2.0.⁴²⁴ The proportion of improved housing increased the least in politically unstable countries, highlighting the role of national stability in improving this aspect of HWB.⁴²⁴



The TOSTAN Community Empowerment Program offers Senegalese women the opportunity to learn about their right to health and to be free from violence. *Credit: Jonathan Torgovnik* | *Getty Images* | *Images of Empowerment*

A major challenge to improving housing for rural populations and people living in poverty is lack of access to capital.⁴²⁵ Most people experiencing poverty work in the informal sector, and 90 percent of the rural lands in SSA are not formally registered, thereby restricting access to formal financing.⁴²⁵ The pay-as-you-go model has been successful in microfinancing the specific energy needs of rural and low-income households.⁴²⁶ The model provides a unique opportunity to explore similar approaches to microfinance for improved housing in LMICs. A detailed survey in Kenya and Uganda found strong evidence that offering microfinance to

low-income households can lead to improved housing.⁴²⁷ The co-benefits and ripple effects of Project Drawdown solutions in providing access to clean electricity and improving education can increase household incomes and improve networks, bringing more households into the formal economy and enabling them to obtain microfinance loans to improve their housing.

Networks: Studies and surveys have shown that **Income and Work** and **Education** are strongly linked with **Networks**. Income is a key barrier to accessing the Internet;^{428, 429} in LMICs, smartphones are more common among highly educated, wealthier populations.⁴³⁰ Improvements in **Education** and **Income and Work** can contribute to improving **Networks**.

Peace and Justice: Lack of economic development, joblessness, and poverty are strongly linked with conflict.⁴³¹ Development increases safety and promotes even more development; conversely, when development fails, the risk of conflict increases. This can have a great impact on the economy and further entrench a country in conflict.⁴³¹ Climate change has been described as a threat multiplier for global peace and security; it can collapse food production, decrease access to fresh water, and affect public health—things that are needed to promote security.⁴³²



Access to sanitation is a basic human right and leads to better health, which can contribute to improved capacity to work and ability to adapt to a changing climate. *Credit: United Nations Children's Fund Ethiopia 2014 | Nesbitt*

Long-term droughts and climate change have been linked to decreases in peace and security. One study across all SSA countries⁴³³ estimated that a rise in average air temperature of 1.0°C could potentially increase the likelihood of internal armed conflicts by 4.5 percent in the same year and by 0.9 percent in the following year. A global meta-analysis of 54 studies that analyzed conflicts after the 1950s suggested climate's influence on modern conflict could be substantial.⁴³⁴ It has been estimated that climate change could lead up to 216 million people to migrate within their own countries by 2050 due to a lack of basic needs, including food and water.²⁶³ More than two-thirds of the migration is expected to occur within SSA and South Asia.²⁶³ However, if the global temperature stabilizes at 1.5°C above preindustrial levels, 80 percent of that migration could be prevented.²⁶³ Whereas intensifying climate change can increase conflicts, it is worth noting that other drivers, such as socioeconomic development and low capabilities of state, have substantially greater influence on regional stability.⁴³⁵

Ripple effects are complex and have been understudied. However, climate solutions identified in this report can contribute to peace and justice through two pathways. The direct one is through reducing GHG emissions or enhancing carbon sinks, which mitigates climate change and thereby contributes to reducing drivers of conflict.²⁶³ The other pathway is through co-

impacts with fundamental HWB dimensions that interact synergistically and can be additive. For instance, better food security, increased economic stability and income, and greater gender equality can play a part in minimizing risks of conflicts, reducing inequality, and improving peace and justice.⁴³⁶

Health and Education can also contribute to peace and justice by building the educational capacity of youth and helping countries make changes in fertility, thereby contributing to maturing the population age structure over time. This leads to a higher share of the working-age population able to contribute to economic growth⁴³⁷ and stability.⁴³⁸

Robust studies analyzing the co-benefits of *Providing Clean Energy* for peace and justice are lacking.²⁶⁸ However, anecdotal evidence suggests that SPV-powered electrification of neighborhoods can contribute to **Peace and Justice** through decreased crime and increased perceived safety by women and homeowners.^{254, 265, 277, 332, 439, 440} (Such benefits can, however, also be realized by conventional [grid] energy sources.)

Gender Equality: Among the multiple solutions in the *Improving Agriculture and Agroforestry* and *Protecting and Restoring Ecosystems* groups, securing land tenure is a key solution that contributes tremendously to **Gender Equality**. Studies suggest that land tenure rights have several positive impacts on **Gender Equality**, such as increasing women's power socially, boosting ability to participate in community-level decision making, reaping economic benefits, benefiting from family assets, and reducing vulnerability to GBV.^{247, 441} Other climate solutions in *Improving Agriculture and Agroforestry* and *Protecting and Restoring Ecosystems* groups further contribute to **Gender Equality** through increase in **Income**, strengthening of **Food** security, and improvements in **Water and Sanitation**.

Social Equity: More than three-quarters of global wealth is concentrated in the top 10 percent of wealthiest people, while the bottom 50 percent owns only 2 percent.⁴⁴² Income inequality has increased globally over the last few decades,⁴⁴³ but this difference is highest in LMICs.⁴⁴⁴ In India—home to the world's second largest population—the richest 1 percent own 60 percent of total wealth.²²² Income inequality affects HWB in myriad ways; its effect on public health is well-documented. It leads to increased mortality rates in infants and adolescents and lowers life expectancy.^{222, 445, 446} Studies have also shown a higher prevalence of depression in countries with high income inequality.²²²

Climate change continues to exacerbate economic inequality,⁴⁴⁷ as people experiencing poverty are generally more exposed to the adverse effects of climate change. Although low-income households are very resourceful and have high social capital, many have limited ability to cope with and recover from climate shocks, which can lead to higher levels of poverty.³⁴ Income inequality and climate change are mutually reinforcing.⁴⁴⁸ Climate solutions identified in this report can address poverty when intentionally designed to do so. Low-carbon development pathways can not only contribute to lowering existing income inequality but can also protect vulnerable communities and individuals from falling back into poverty by increasing their resilience and decreasing future vulnerability to climate change.

Political Voice: Education and increased income can improve political voice. Several studies and surveys across the world have shown that education advancement leads to greater political participation and civic engagement.^{449, 450, 451, 452, 453} However, the opposite has also been observed in authoritarian and non-democratic countries; education can decrease individual political participation as people may believe that participation legitimizes autocrats or is futile.⁴⁵⁴

Good governance and the collection and redistribution of tax revenue, particularly in natural resource–rich nations, plays a role in political voice and accountability. A low tax-to-GDP ratio enables governments to remain unaccountable to citizens for how they spend revenue. At the same time, citizens are not incentivized to pay taxes as they are not confident that the funds will be spent in ways that benefit them.⁴⁵⁵ As LMICs, particularly in SSA, transition from economies focused on natural resource extraction to more diversified economies, tax revenue from manufacturing facilities linked to the green economy could be substantial.

Where women can overcome structural barriers to their participation, they are uniquely positioned to contribute to sustainable natural resource management, climate-resilient communities, and enhanced peace and stability.

JESSICA M. SMITH, LAUREN OLOSKY, AND JENNIFER GROSMAN FERNÁNDEZ,⁴¹⁸ THE CLIMATE-GENDER-CONFLICT NEXUS, GEORGETOWN INSTITUTE FOR WOMEN, PEACE AND SECURITY, 2021



4.7 Summary and conclusion

In addition to reducing carbon emissions, climate solutions like *Conservation Agriculture, Nutrient Management, and Distributed Solar Photovoltaics* can be combined to improve people's food security, access to water and electricity, and income, which can lead to positive ripple effects across their lives. *Credit: Ayush Manik (Independent Consultant)* | *CGIAR*

This literature review provides solid evidence for numerous co-benefits of climate change solutions for alleviating poverty and improving HWB. Many development initiatives in LMICs have a single focus, such as climate-smart agriculture, maternal health, or water and sanitation. Yet the challenges that under-resourced rural communities in LMICs face—such as poverty, hunger, lack of electricity, and poor water quality—are often linked, as shown in the HWI (Figure 3.3 and Appendix C).

The COVID-19 pandemic has amplified inequities in health, socioeconomic status, livelihoods, and access to crucial resources. At the same time, people around the world are demanding more urgent climate mitigation and adaptation actions as climate-driven drought, famine, wildfires, and floods become more devastating and frequent.

For people experiencing poverty in rural areas in LMICs—who have no insurance policies and limited government assistance to face and recover from pandemic and climate-induced shocks—climate solutions that complement and reinforce each other while also boosting HWB can be a part of a low-carbon pathway to prosperity, particularly when accompanied by stronger accountability from governments to citizens, cross-sectoral collaboration, and strengthened development planning.

Many communities on the front lines of the climate crisis are already experiencing loss and damage. Communities cannot adapt to extinction, communities cannot adapt to starvation. The climate crisis is pushing so many people in places where they cannot adapt anymore.

VANESSA NAKATE, CLIMATE ACTIVIST, UN VOICES OF CHANGE,456 JANUARY 2022

MOBILIZING AROUND THE CLIMATE AND POVERTY NEXUS: CRITICAL ACTIONS

Improving the well-being of people experiencing extreme poverty and addressing climate change can and must be complementary: There should be no trade-offs between human well-being and climate action. Climate must become a central concern for development, and sustainable development must become a central path to fulfilling climate solutions.

The HWI, introduced in this report, serves as a framework for assessing 12 health, socioeconomic, and societal dimensions of human well-being and highlighting evidence of the nexus between climate mitigation solutions and human well-being in a scientifically consistent manner (Figure 3.3 and Appendix C). This report presents detailed scientific evidence of the many co-benefits and ripple effects for HWB that can be achieved by pursuing 28 Project Drawdown climate solutions at scale. Deeper understanding of the relative HWI value of a country and awareness of barriers to improving it are critical to better contextualize and prioritize climate solutions that both reduce emissions and alleviate poverty. Climate solutions not only provide mitigation benefits but can also improve HWB and boost economies in LMICs. This report intends to raise awareness of the socioeconomic co-benefits of climate solutions, promote cross-sectional alliances to mobilize ambitious solutions, and channel funding for synergistic solutions from public, philanthropic, and private sectors.

The win-win solutions presented here can provide tremendous benefits to HWB while simultaneously addressing climate change. Specifically, the total global CO₂-eq reduction potential of the 28 solutions presented here is approximately 690 Gt between 2020 and 2050 (Project Drawdown 2022 estimates^{xxiii}); for comparison, global GHG emissions were between 48.1 and 50.6 Gt-CO₂-eq annually in 2018–2020. Several of the solutions with large GHG reduction potential at a global level are widely applicable in LMICs—*Tropical Forest Restoration, Indigenous Peoples Forest Tenure,* and *Improved Clean Cookstoves*—with tremendous

xxiii Updates to current solutions and new solutions will be posted to the Project Drawdown website in 2022.
potential benefits for people and the climate. These solutions alone contribute to a net reduction of 173 Gt CO_2 -eq—about 25 percent of the CO_2 -eq reduction potential of the solutions presented here.

All five of the climate solutions groups—*Improving Agriculture and Agroforestry, Protecting and Restoring Ecosystems, Adopting Clean Cooking, Providing Clean Electricity,* and *Fostering Equality*—generate enormous direct and indirect co-benefits to HWB. In particular, they generate benefits for people around Income and Work, Health, Food Security, Education, Gender Equality, and Energy that would be transformational in alleviating poverty and increasing resilience to current and future climate change. In addition, the *Providing Clean Electricity* solution group has strong co-benefits for Networks, while *Fostering Equality* has cobenefits for Social Equity and Peace and Justice (Figure 5.1).





Project Drawdown's climate solutions offer substantial co-benefits for human well-being dimensions included in the Drawdown Lift Human Well-Being Index, from Income and Work (greatest) to Political Voice (least). (Gt CO₂-eq is gigatons of carbon dioxide equivalent.)

*Note: Family planning and education are not in themselves climate mitigation strategies. Rather, it is the outcome of Fostering Equality, slower population growth, that is a climate solution.

The opportunity to address climate change and improve HWB, particularly in LMICs, is enormous. Implementing the solutions presented here could make a dramatic difference for climate mitigation, climate adaptation, and improving HWB—especially when the solutions are implemented synergistically with cross-sectional collaboration.

In addition to implementing the solutions analyzed in this report, we find that several additional actions can also contribute to mobilizing win-win solutions to climate and poverty (Figure 5.2):

Listen to, engage with, and work alongside diverse stakeholders—from local practitioners to senior policymakers—to build capacity and plan for synergistic climate-development

actions. Climate considerations need to become part of mainstream development efforts around green jobs, clean energy, food, health, infrastructure, and education—and vice versa. Global and national leaders must listen to and support women, youth, Indigenous peoples, and local communities in climate negotiations, include them in national and regional policy dialogues, and respect their demands for climate action. Simultaneously addressing climate change and fulfilling the fundamental needs of people around the world—in ways that incorporate their guidance and knowledge—is critical to advance equitable and climatesmart development.



The Health Development Initiative, founded by Rwandan physicians, facilitates conversations about sexual and reproductive health and rights. *Credit: Yagazie Emezi | Getty Images | Images of Empowerment*

Include HWB co-benefits in multisector climate finance. Development organizations and government ministries must collaborate with diverse organizations and institutions, policymakers, the private sector, and donors to scale the implementation of climate solutions that boost HWB. Extensive coordination is required to shed the tendency of development organizations and government ministries to focus on sector-specific goals and overlook climate policy co-benefits. In addition, stakeholders need to engage climate finance project teams that prepare, design, implement, and monitor climate finance projects and the implementing agencies and country focal points that oversee and collaborate with them to purposefully include HWB co-benefits in climate finance projects. More synergistic collaboration and coordination and intentional inclusion of co-benefits as part of climate finance also addresses inequality and bends the curve toward gender equality. For example, leaders can align and complement national climate strategies with other sector-specific approaches, such as national gender policies and national education plans.



Students in Lushoto, Tanzania, care for seedlings at school. Credit: Georgina Smith | CIAT 2014

Accelerate the development of innovative finance for climate solutions that include HWB co-benefits. As LMICs track the US\$100-billion-per-year pledge for adaptation and mitigation measures, as well as new climate pledges from the philanthropic and private sectors, it is crucial to link climate mitigation solutions with HWB. Most climate finance goes to projects in middle-income countries and is focused on reducing emissions, which can often generate return on investment. Yet, needs are extremely high in low-income countries for adaptation. Linking climate mitigation solutions that provide clear co-benefits for HWB would enable the least developed and the most vulnerable countries to drive a shift in the approaches that private and public funding partners take to meet their self-identified environmental, social, and governance needs. Country focal points and implementing agencies could incorporate these win-win solutions more prominently into their Nationally Determined Contributions and National Adaptation Plans and tap into multilateral climate funds like the Adaptation Fund, Green Climate Fund, Climate Investment Funds, and Global Environment Facility to de-risk private investments. Decisionmakers must collectively leverage the incentives and strengths of public and private finance and use existing finance tools and institutions to manage risk and pursue innovation. Blended finance approaches whereby the private sector contributes to clean energy and conserving ecosystems, such as those developed by the Global Environment Facility, are becoming more common. The cost of inaction today outweighs the risks of financing climate action for the future.

Assemble additional evidence of HWB co-benefits to further drive a more holistic approach to climate solutions and poverty alleviation. As the report shows, many climate solutions can serve dual roles of mitigating GHG emissions and improving HWB. However, their efficacy in addressing HWB has not been widely tested in real-world settings. Research that evaluates the multiple co-benefits of a solution or a group of solutions is rare. Evaluations of potential co-benefits (and trade-offs) of climate solutions for HWB would be beneficial, along with analysis that looks for unintended consequences and differential impacts by gender, cultural integrity, and income. Researchers should collaborate with organizations advancing synergistic initiatives like the ones highlighted in case studies in this report that jointly address climate change and HWB.

Figure 5.2 — Key actions to foster synergistic solutions



Addressing multidimensional poverty is exceedingly complex; collaboratively aligning poverty alleviation efforts with climate-smart development is daunting. Countries and stakeholders at all levels of society should not have to choose among human development, climate mitigation, and climate adaptation; win-win solutions are at hand. The world has a window of opportunity to align strategies, complement funding, and harmonize policies at local, national, and international scales to simultaneously address climate change, alleviate poverty, and boost human well-being. By charting a more synergistic path, decisionmakers, policymakers, and funders—informed by and in collaboration with women, youth, Indigenous peoples, and local communities—can ensure a brighter future for people and the planet.

The global community must ensure that those most impacted by climate change (and often the least responsible for it) have access to resources necessary to adapt to climate change and mitigate future emissions to meet their self-identified needs and ensure a sustainable future for generations to come. Credit: Ollivier Girard | CIFOR

APPENDIX A

Descriptions of the 28 Project Drawdown solutions

Note: The hyperlink in each solution leads to the solution summary on Project Drawdown's website.

Abandoned Farmland Restoration: Degraded farmland is often abandoned, but it need not be. Restoration can bring these lands back into productivity and sequester carbon in the process.

Biogas for Cooking: Anaerobic digesters process backyard or farmyard organic waste into biogas and digestate fertilizer. Biogas stoves can reduce emissions when replacing biomass or kerosene for cooking.

<u>Coastal Wetland Protection</u>: Mangroves, salt marshes, and seagrasses sequester huge amounts of carbon in plants and soil. Protecting them inhibits degradation and safeguards their carbon sinks.

<u>Coastal Wetland Restoration</u>: Agriculture, development, and natural disasters have degraded many coastal wetlands. Restoring mangrove forests, salt marshes, and seagrass beds to health revives carbon sequestration.

<u>Conservation Agriculture</u>: Conservation agriculture uses cover crops, crop rotation, and minimal tilling in the production of annual crops. It protects soil, avoids emissions, and sequesters carbon.

Distributed Solar Photovoltaics: Whether grid-connected or part of stand-alone systems, rooftop solar panels and other distributed solar photovoltaic systems offer hyper-local, clean electricity generation.

<u>Farm Irrigation Efficiency</u>: Pumping and distributing water is energy-intensive. Drip and sprinkler irrigation, among other practices and technologies, make the use of farm water more precise and efficient.

Forest Protection: In their biomass and soil, forests are powerful carbon storehouses. Protection prevents emissions from deforestation, shields that carbon, and enables ongoing carbon sequestration.

<u>Geothermal Power</u>: Underground reservoirs of steamy hot water are the fuel for geothermal power. The water can be piped to the surface to drive turbines that produce electricity without pollution.

<u>Grassland Protection</u>: Grasslands hold large stocks of carbon, largely underground. Protecting them shields their carbon stores and avoids emissions from conversion to agricultural land or development.

<u>Health and Education</u>: Some initiatives, designed primarily to ensure rights and foster equality, also have cascading benefits to climate change. They include access to high-quality, voluntary reproductive health care and to high-quality, inclusive education, which are fundamental human rights and cornerstones of gender equality.

Improved Clean Cookstoves: Improved clean cookstoves can address the pollution from burning wood or biomass in traditional stoves. Using various technologies, they reduce emissions and protect human health.

Improved Rice Production: Flooded rice paddies produce large quantities of methane. Improved production techniques, including alternate wetting and drying, can reduce methane emissions and sequester carbon. Indigenous People's Forest Tenure: Secure land tenure protects Indigenous peoples' rights. With sovereignty, traditional practices can continue—in turn protecting ecosystems and carbon sinks and preventing emissions from deforestation.

<u>Micro Wind Turbines</u>: Micro wind turbines can generate clean electricity in diverse locations, from urban centers to rural areas, without access to centralized grids.

<u>Microgrids</u>: A microgrid is a localized grouping of distributed electricity generation technologies, paired with energy storage or backup generation and tools to manage demand or "load."

<u>Multistrata Agroforestry</u>: Multistrata agroforestry systems mimic natural forests in structure. Multiple layers of trees and crops achieve high rates of both carbon sequestration and food production.

<u>Nutrient Management</u>: Overuse of nitrogen fertilizers—a frequent phenomenon in agriculture creates nitrous oxide. More efficient use can curb these emissions and reduce energy-intensive fertilizer production.

Peatland Protection and Rewetting: Forestry, farming, and fuel extraction are among the threats to carbon-rich peatlands. Protection and rewetting can reduce emissions from degradation while supporting peatlands' role as carbon sinks.

<u>Reduced Food Waste</u>: Roughly one-third of the world's food is never eaten, which means the land and resources used and GHGs emitted in producing it were unnecessary. Interventions can reduce loss and waste as food moves from farm to fork, thereby reducing overall demand.

Regenerative Annual Cropping: Building on conservation agriculture with additional practices, regenerative annual cropping can include compost application, green manure, and organic production. It reduces emissions, increases soil organic matter, and sequesters carbon.

<u>Silvopasture</u>: An agroforestry practice, silvopasture integrates trees, pasture, and forage into a single system. Incorporating trees improves land health and significantly increases carbon sequestration.

Small Hydropower: Small hydropower systems capture the energy of free-flowing water without using dams. They can replace dirty diesel generators with clean electricity generation.

<u>Sustainable Intensification for Smallholders</u>: Sustainable intensification practices can increase smallholder yields which, in theory, reduce demand to clear additional land. Practices include intercropping, ecosystem-based pest management, and equal resources for women.

<u>System of Rice Intensification</u>: This is a holistic approach to sustainable rice cultivation. By minimizing water use and alternating wet and dry conditions, it minimizes methane production and emissions.

<u>Temperate Forest Restoration</u>: Almost all temperate forests have been altered in some way timbered, converted to agriculture, or disrupted by development. Restoring them sequesters carbon in biomass and soil.

Tree Intercropping: Growing trees and annual crops together is a form of agroforestry. Tree intercropping practices vary, but all increase biomass, soil organic matter, and carbon sequestration.

<u>Tropical Forest Restoration</u>: Tropical forests have suffered extensive clearing, fragmentation, degradation, and depletion of biodiversity. Restoring these forests also restores their function as carbon sinks.

APPENDIX B

Summary of common human well-being indices and description of indicators used in this report

Table B.1 — Summary of commonly used socioeconomic and human well-being indices

| | Developer/ | No. of | | HUMAN WELL-BEING DIMENSIONS | | | | | |
|---|---|--------|-------------------|-----------------------------|-----------|-------------|--------|----------|--------|
| | Institution | | Coverage | Social | Education | Environment | Health | Economic | Gender |
| <u>GLOBAL</u> MULTIDIMENSIONAL POVERTY_ INDEX (MPI)* | UNDP/Oxford Poverty Human Development Initiative (OPHI) | 10 | LMICs | | 1 | | - | | |
| <u>HUMAN</u> <u>DEVELOPMENT</u> INDEX | UNDP | 4 | Global | | 1 | | | 1 | |
| ENVIRONMENTAL PERFORMANCE INDEX | Yale/Columbia University | 32 | Global | | | | • | | |
| BETTER LIFE INDICATOR | OECD | 24 | OECD countries | | - | | - | - | |
| Social progress INDEX | Social Progress Imperative | 53 | Global | | • | • | • | | • |
| DOUGHNUT ECONOMICS | Kate Raworth/Oxfam International | 30** | Global | | | - | | | |

* Disaggregation of data by gender was introduced in the 2021 MPI.

** Two of these, air pollution and chemical pollution, did not have globally defined measures in the original framework.

Table B.2 — Brief description and data sources for the indicators included in the Drawdown Lift Human Well-Being Index

| HWB DIMENSION | INDICATOR | DESCRIPTION | SOURCE | |
|-------------------------|---|--|--|--|
| ENERGY | Access to electricity | Percentage of population with access to electricity | World Bank | |
| | Access to clean cooking fuel and technology | Percentage of the population with primary reliance on clean fuels and technology for cooking | <u>World Bank</u> | |
| WATER AND SANITATION | Access to clean water | Percentage of the population with access to water that is sourced from an improved source with a collection time of < 30 minutes | World Health Organization Joint Monitoring Programme (WHO JMP) | |
| | Access to basic sanitation | Percentage of population with access to improved facilities not shared with other households | WHO JMP | |
| FOOD | Population undernourished | Percentage of population that is undernourished (three-year average) | FAO | |
| | Prevalence of child malnutrition | Percentage of children under age 5 who are stunted | <u>FAO</u> | |
| HEALTH | Child mortality rate | Mortality rate under age 5 per 1,000 live births | World Bank | |
| | Maternal mortality ratio ^{xxiv} | Number of maternal deaths per 100,000 live births | <u>WHO</u> | |
| | Life expectancy | Life expectancy at birth (years) | <u>WHO</u> | |
| | Demand satisfied for family planning | Percentage of women of reproductive age (15–49 years) whose need for family planning is satisfied with modern methods | <u>United Nations</u> | |
| | Suicide rate | Number of suicide deaths per 100,000 people per year | <u>WHO</u> | |
| EDUCATION | Literacy rate | Percentage of people aged 15 and above who can read and write, with understanding of a short, simple statement about their everyday life | United Nations Children's Fund | |
| | Children enrolled in secondary school | Percentage of all eligible children enrolled in secondary education | <u>World Bank</u> | |
| | Education quality | Expert evaluation of the extent to which high- quality basic education is guaranteed to all people, measured on a scale of 0 (extremely unequal access) to 4 (equal access) | <u>SPI</u> | |
| INCOME AND WORK | Poverty rate | Percentage of people living on less than US\$1.90 per day at 2011 international prices ^{xxv} | World Bank | |
| | Unemployment rate | Percentage of the total workforce that is unemployed | World Bank | |
| | Gender wage gap | Pay gap between men and women workers (%) | International Labor Organization | |

xxiv The Maternal Mortality Ratio is defined as the number of female deaths from pregnancy-related causes during pregnancy and childbirth or within 42 days of the termination of pregnancy during a given time period per 100,000 live births during the same time period. It represents the risk of maternal death relative to the number of live births and essentially captures the risk of death in a single pregnancy or live birth.

xxv Please refer to the World Bank's PovCal website for additional information on the methodology of calculating the poverty rate.

| HWB DIMENSION | INDICATOR | DESCRIPTION | SOURCE |
|----------------------|--|--|--|
| NETWORKS | Access to internet | Percentage of people that used the internet in the last three months | <u>World Bank</u> |
| | Access to mobile phone | Percentage of people with cell phone subscriptions | <u>World Bank</u> |
| | Freedom of movement | Composite of four indicators capturing domestic and international freedom of movement | <u>World Bank</u> |
| HOUSING | Inadequate housing | Households with inadequate housing materials in any of three components: floor, roof, or walls | Multidimensional Poverty Index |
| GENDER EQUALITY | Women's representation in government | Percentage of parliamentary seats held by women | World Bank |
| | Child, early, and forced marriage and unions | Percentage of women aged 20–24 who were married or in a union before age 18 | <u>UN Stats</u> |
| | Intimate partner violence | Percentage of girls and women aged 15 and older that has ever experienced physical and/or sexual violence from an intimate partner | <u>Our World in Data</u> |
| | National abortion policies | Composite of four indicators (available at woman's request, legal and gestational limits around economic or social reasons, health, and life of the mother) to provide the extent of abortion restrictions | <u>WHO Global</u> <u>Abortion Policy</u> <u>Database</u> |
| SOCIAL EQUITY | Income distribution | The degree of inequality in the distribution of income or wealth among households or individuals | World Bank |
| | Civil liberties | Composite of four indicators: rule of law, personal autonomy and individual rights, freedom of expression and belief, and associational and organizational rights, ranging from 1 (strong liberties) to 7 (no liberties) | The Freedom House |
| PEACE AND JUSTICE | Corruption perception | Perceptions of administrative and political corruption by country experts and business people (calculated using data from surveys collected by reputable institutions) | Transparency_ International |
| | Political stability | Likelihood that a government will be removed through unconstitutional or violent means (index) | World Bank |
| | Homicide rate | Number of homicides per 100,000 people per year | The UN Office on Drugs and Crime |
| POLITICAL VOICE | Voice and accountability | Citizens' perception of the extent to which they can participate in selecting their government, as well as freedom of expression, freedom of association, and media freedom (index) | World Bank |

APPENDIX C Calculating the Drawdown Lift Human Well-Being Index

For each indicator, we used the most recently available country-level data collected in the pre-COVID-19 era (e.g., child and maternal mortality rates from 2019). For a few countries and a select set of indicators, data from the same year were unavailable. For instance, the most recent Gini coefficient value (which measures the wealth inequality in a country) available for each country was included because this figure is not calculated every year. When calculating the HWI for a country, we did not include indicators with missing data. A brief description of each indicator and its data source(s) is provided in **Appendix B**.

The approach to calculate the HWI was straightforward. First, for each indicator, outliers defined as values falling outside the minimum (Q1 – 1.5*Q1) and the maximum (Q3 + 1.5*Q3) values, where Q1 and Q3 are the first and the third quartile—were removed to minimize the disproportionate effect of very low or high values. Data were then divided into five equal intervals, allowing for categorization into five percentile groups: 0–20 percent, 20–40 percent, 40–60 percent, 60–80 percent and 80–100 percent. Countries with values lower or higher than the defined minimum and maximum values were assigned to groups 0–20 percent and 80–100 percent, respectively. For indicators where increasing values reflect lower HWB (e.g., poverty rate), the ranks were inverted, assigning rank 1 to the 80–100 percentile. Finally, the ranks of the indicators were averaged to obtain the relative well-being of a country for a given dimension. A constraint of the Drawdown Lift Human Well-Being Index is that calculating the relative well-being of a country for a given dimension does not capture regional or community-level variations. All dimensions are weighted equally, and they were then averaged to obtain the overall HWI of a country. This approach was used to highlight the relative (not the absolute) difference between the countries.

Figure C.1 — Drawdown Lift Human Well-Being Index provides an indication of relative human well-being for A: Income and Work, B: Water and Sanitation, C: Food, D: Health, E: Gender Equality, and F: Energy



A: Income and Work

B: Water and Sanitation



C: Food

D: Health















B: Networks



C: Housing

D: Peace and Justice



To Data



Low High





REFERENCES

- 1 Clark, D. A. (2014). Defining and measuring human well-being. In B. Freedman (Ed.), *Global environmental change* (pp. 833–855). Springer Netherlands. <u>https://doi.org/10.1007/978-94-007-5784-4_66</u>
- 2 Lutz, W., Striessnig, E., Dimitrova, A., Ghislandi, S., Lijadi, A., Reiter, C., ... & Yildiz, D. (2021). Years of good life is a well-being indicator designed to serve research on sustainability. *Proceedings of the National Academy of Sciences*, 118(12), e1907351118. <u>https://doi.org/10.1073/pnas.1907351118</u>
- 3 World Bank (2018). *Poverty and shared prosperity 2018*: Piecing together the poverty puzzle. <u>https://www.</u> worldbank.org/en/publication/poverty-and-shared-prosperity-2018
- 4 World Bank (2020). Poverty and shared prosperity 2020: Reversals of fortune. The World Bank. https://openknowledge.worldbank.org/handle/10986/34496
- 5 Maurer, P. (2021). Protecting the natural environment in armed conflict. *Environmental Policy and Law*, (Preprint), 1–4. https://www.icrc.org/en/publication/4487-when-rain-turns-dust
- 6 Barrett, C., Carter, M., & Chavas, J.-P. (2018). Shifting Assets, Capacities and Risk to Aid Resilience and End Chronic Poverty. Feed the Future. <u>https://basis.ucdavis.edu/sites/g/files/dgvnsk466/files/2018-11/Feed%20the%20Future%20</u> AMA%20Innovation%20Lab%20Brief%202018-08%20-%20Barrett%20Carter%20Chavas%20Poverty%20Traps%20 -%20WEB.pdf
- 7 International Monetary Fund (2021). World Economic Outlook: Recovery During a Pandemic Health Concerns, Supply Disruptions, and Price Pressures, 172. International Monetary Fund. <u>https://www.imf.org/en/Publications/</u> WEO/Issues/2021/10/12/world-economic-outlook-october-2021
- 8 United Nations Women (2022). From Insights to Action: Gender Equality in the Wake of COVID-19, 14. UN Women. https://doi.org/10.18356/f837e09b-en
- 9 United Nations Children's Fund (2021). COVID-19: A threat to progress against child marriage. UNICEF, New York. https://data.unicef.org/resources/covid-19-a-threat-to-progress-against-child-marriage/
- 10 Silwal, A. R., Engilbertsdottir, S., Cuesta, J., Newhouse, D., & Stewart, D. (2020). Global estimate of children in monetary poverty. <u>https://doi.org/10.1596/34704</u>
- Munoz Boudet, A. M., Bhatt, A., Azcona, G., Yoo, J., & Beegle, K. (2021). A global view of poverty, gender, and household composition. (Policy Research Working Paper No. 9553, 29). The World Bank. <u>https://doi.org/10.1596/1813-9450-9553</u>
- 12 Romanello, M., McGushin, A., Di Napoli, C., Drummond, P., Hughes, N., Jamart, L., ... & Hamilton, I. (2021). The 2021 report of the Lancet Countdown on health and climate change: code red for a healthy future. *The Lancet,* 398(10311), 1619–1662. <u>https://doi.org/10.1016/S0140-6736(21)01787-6</u>
- 13 Pörtner, H. O., Roberts, D. C., Adams, H., Adler, C., Aldunce, P., Ali, E., ... & Ibrahim, Z. Z. (2022). Climate change 2022: impacts, adaptation and vulnerability. <u>https://report.ipcc.ch/ar6wg2/pdf/IPCC_AR6_WGII_FinalDraft_FullReport.pdf</u>
- 14 Boyer, A. E., Meijer, S. S., & Gilligan, M. (2020). Advancing gender in the environment: Exploring the triple nexus of gender inequality, state fragility, and climate change, 141. IUCN. <u>https://www.climatelinks.org/resources/advancing-gender-environment-exploring-triple-nexus-gender-inequality-state-fragility-and</u>
- 15 Watts, N., Amann, M., Arnell, N., Ayeb-Karlsson, S., Belesova, K., Boykoff, M., ... & Montgomery, H. (2019). The 2019 report of The Lancet Countdown on health and climate change: ensuring that the health of a child born today is not defined by a changing climate. *The Lancet*, *394*(10211), 1836–1878. <u>https://doi.org/10.1016/S0140-6736(19)32596-6</u>
- 16 Langer, A., Meleis, A., Knaul, F. M., Atun, R., Aran, M., Arreola-Ornelas, H., ... & Frenk, J. (2015). Women and health: the key for sustainable development. *The Lancet, 386*(9999), 1165–1210. <u>https://doi.org/10.1016/S0140-6736(15)60497-4</u>
- 17 van Daalen, K., Jung, L., Dhatt, R., & Phelan, A. L. (2020). Climate change and gender-based health disparities. *The Lancet Planetary Health*, 4(2), e44–e45. https://doi.org/10.1016/S2542-5196(20)30001-2
- 18 Muttarak, R., & Lutz, W. (2014). Is education a key to reducing vulnerability to natural disasters and hence unavoidable climate change? *Ecology and Society*, 19(1), 8. <u>https://doi.org/10.5751/ES-06476-190142</u>
- 19 Johnson, J. A., Ruta, G., Cervigni, R., Chonabayashi, S., Corong, E., Gavryliuk, O., Gerber, J., Hertel, T., Nootenboom, C., & Polasky, S. (2021). The economic case for nature: A global Earth-economy model to assess development policy pathways, 183. World Bank. <u>https://www.worldbank.org/en/topic/environment/publication/the-economic-casefor-nature</u>
- 20 World Bank (2020). World Bank Group Strategy for Fragility, Conflict, and Violence 2020–2025, 138. (2020). The World Bank. <u>https://doi.org/10.1596/978-1-4648-0484-7</u>

- 21 Chen, C., Noble, I., Hellmann, J., Coffee, J., Murillo, M., & Chawla, N. (2015). University of Notre Dame Global Adaptation Index. *University of Notre Dame: Notre Dame, IN, USA*. Retrieved from <u>https://gain.nd.edu/our-work/</u> <u>country-index/rankings/</u>
- 22 Vincent, K., Tschakert, P., Barnett, J., Rivera-Ferre, M. G., & Woodward, A. (2014). Cross-chapter box on gender and climate change. In *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment of the Intergovernmental Panel on Climate Change* (pp. 105–107). Cambridge University Press. <u>https://kulima.com/wp-content/uploads/2011/03/Cross-chapterbox-on-gender-and-climate-change.pdf</u>
- 23 FAO (2018). *Tackling climate change through rural women's empowerment,* 20. Food and Agriculture Organization of the United Nations. <u>http://www.fao.org/3/ca0178en/CA0178EN.pdf</u>
- 24 UNDP and OPHI (2021). Global Multidimensional Poverty Index 2021: Unmasking disparities by ethnicity, caste and gender, 41. United Nations Development Programme and Oxford Poverty and Human Development Initiative. <u>http://</u> <u>hdr.undp.org/en/2021-MPI</u>
- 25 Neumayer, E., & Plümper, T. (2007). The Gendered nature of natural disasters: The impact of catastrophic events on the gender gap in life expectancy, 1981–2002. *Annals of the Association of American Geographers*, *97*(3), 551–566. https://doi.org/10.1111/j.1467-8306.2007.00563.x
- 26 UNFCCC (2021). Gender and climate change: Draft conclusions proposed by the Chair, 3. United Nations Framework Convention on Climate Change. Glasgow. <u>https://unfccc.int/sites/default/files/resource/sbi2021_L13E.pdf</u>
- 27 UNDP (2021). Gender equality in public administration, 184. United Nations Development Programme and University of Pittsburgh. https://www.undp.org/publications/global-report-gender-equality-public-administration
- 28 Leveraging Co-benefits between gender equality and climate action for sustainable development: Mainstreaming gender considerations in climate change projects. (2016). UN Women. <u>https://unfccc.int/files/gender_and_climate_ change/application/pdf/leveraging_cobenefits.pdf</u>
- 29 Luthen, S., Ryan, E., & Wakefield, J. (2021). Born into the climate crisis: Why we must act now to secure children's rights, 50. <u>https://resourcecentre.savethechildren.net/pdf/born-into-the-climate-crisis.pdf/</u>
- 30 Thiery, B. W., Lange, S., Rogelj, J., Schleussner, C.-F., Gudmundsson, L., Seneviratne, S. I., Andrijevic, M., Frieler, K., Emanuel, K., Geiger, T., Bresch, D. N., Zhao, F., Willner, S. N., Büchner, M., Volkholz, J., Bauer, N., Chang, J., Ciais, P., Dury, M., ... Wada, Y. (2021). Intergenerational inequities in exposure to climate extremes. *Science*, *374*(6564), 158–160. <u>https://doi.org/10.1126/science.abi7339</u>
- 31 Hallegatte, S., & Rozenberg, J. (2017). Climate change through a poverty lens. Nature Climate Change, 7(4), 250–256. https://doi.org/10.1038/nclimate3253
- 32 Rees, N. (2021). The Climate crisis is a child rights crisis: Introducing the Children's Climate Risk Index, 128. UNICEF. https://www.unicef.org/reports/climate-crisis-child-rights-crisis_
- 33 Niles, M. T., Emery, B. F., Wiltshire, S., Brown, M. E., Fisher, B., & Ricketts, T. H. (2021). Climate impacts associated with reduced diet diversity in children across nineteen countries. *Environmental Research Letters*, 16(1), 015010. <u>https://doi.org/10.1088/1748-9326/abd0ab</u>
- 34 Hallegatte, S. (2016). Shock Waves: Managing the Impacts of Climate Change on Poverty (p. 227). World Bank Publications. <u>https://doi.org/10.1596/978-1-4648-0673-5</u>
- 35 UNFCCC (2016). Report of the Conference of the Parties on its twenty-first session, held in Paris from 30 November to 11 December 2015. Addendum. Part two: Action taken by the Conference of the Parties at its twenty-first session. (Report of the Conference of the Parties FCCC/CP/2015/10/Add.1, 36. United Nations Framework Convention on Climate Change. <u>https://unfccc.int/sites/default/files/resource/docs/2015/cop21/eng/10a01.pdf</u>
- 36 Charles, A., Kalikoski, D., & Macnaughton, A. (2019). Addressing the climate change and poverty nexus: A coordinated approach in the context of the 2030 Agenda and the Paris Agreement, 104. Rome, FAO. <u>https://www.fao.org/publications/card/en/c/CA6968EN/</u>
- 37 Markkanen, S., & Anger-Kraavi, A. (2019). Social impacts of climate change mitigation policies and their implications for inequality. *Climate Policy*, *19*(7), 827–844. <u>https://doi.org/10.1080/14693062.2019.1596873</u>
- 38 Cohen, B., Cowie, A., Babiker, M., Leip, A., & Smith, P. (2021). Co-benefits and trade-offs of climate change mitigation actions and the Sustainable Development Goals. Sustainable Production and Consumption, 26, 805–813. <u>https:// doi.org/10.1016/j.spc.2020.12.034</u>
- 39 Frischmann, C. J., Mehra, M., Allard, R., Bayuk, K., Gouveia, J. P., & Gorman, M. R. (2020). Drawdown's "System of Solutions" helps to achieve the SDGs. In W. Leal Filho, A. M. Azul, L. Brandli, A. Lange Salvia, & T. Wall (Eds.), *Partnerships for the Goals* (pp. 1–25). Springer International Publishing. <u>https://doi.org/10.1007/978-3-319-71067-9_100-1</u>
- 40 Helgenberger, S., Jänicke, M., & Gürtler, K. (2019). Co-benefits of climate change mitigation. In Leal Filho, W., Azul, A. M., Brandli, L., Özuyar, P., & Wall, T. (Eds.), *Climate Action. Encyclopedia of the UN Sustainable Development Goals.* (pp. 1–13). Springer, Cham. <u>https://doi.org/10.1007/978-3-319-71063-1_93-1</u>

- 41 McKinsey Sustainability (2022). The net-zero transition: What it would cost, what it could bring. <u>https://www.mckinsey.com/business-functions/sustainability/our-insights/the-net-zero-transition-what-it-would-cost-</u> what-it-could-bring
- 42 Puliti, R. (2022, January 31). Putting Africa on the path to universal electricity access [Sustainable Energy for All]. World Bank Blogs. <u>https://blogs.worldbank.org/energy/putting-africa-path-universal-electricity-access</u>
- 43 Hickel, J. (2020). Quantifying national responsibility for climate breakdown: An equality-based attribution approach for carbon dioxide emissions in excess of the planetary boundary. *The Lancet Planetary Health, 4*(9), e399–e404. https://doi.org/10.1016/S2542-5196(20)30196-0
- 44 Bruckner, B., Hubacek, K., Shan, Y., Zhong, H., & Feng, K. (2022). Impacts of poverty alleviation on national and global carbon emissions. *Nature Sustainability*, 1–10. <u>https://doi.org/10.1038/s41893-021-00842-z</u>
- 45 Williams, J. (2022, January 26). Why climate change is inherently racist. *BBC Future*. <u>https://www.bbc.com/future/article/20220125-why-climate-change-is-inherently-racist</u>
- 46 Karlsson, M., Alfredsson, E., & Westling, N. (2020). Climate policy co-benefits: A review. *Climate Policy*, 20(3), 292–316. <u>https://doi.org/10.1080/14693062.2020.1724070</u>
- 47 Deng, H. M., Liang, Q. M., Liu, L. J., & Anadon, L. D. (2018). Co-benefits of greenhouse gas mitigation: A review and classification by type, mitigation sector, and geography. *Environmental Research Letters*, *12*(12), 123001. <u>https://doi.org/10.1088/1748-9326/aa98d2</u>
- 48 Kongsager, R., Locatelli, B., & Chazarin, F. (2016). Addressing climate change mitigation and adaptation together: A global assessment of agriculture and forestry projects. *Environmental Management*, 57(2), 271–282. <u>https://doi.org/10.1007/s00267-015-0605-y</u>
- 49 Zhenmin, L., & Espinosa, P. (2019). Tackling climate change to accelerate sustainable development. *Nature Climate Change*, 9(7), 494–496. <u>https://doi.org/10.1038/s41558-019-0519-4</u>
- 50 USAID (2017). *Hariyo Ban Program–Phase II*. U.S. Agency for International Development, 4. <u>https://www.usaid.gov/sites/default/files/documents/1861/Hariyo_Ban_II_-Fact_Sheet_-Draft_09142017.pdf</u>
- 51 World Wildlife Fund (n.d.) *Hariyo Ban: Mitigating and adapting to climate change in Nepal.* World Wildlife Fund. Retrieved from <u>https://www.worldwildlife.org/projects/hariyo-ban-mitigating-and-adapting-to-climate-change-in-nepal</u>
- 52 Wilson Center (2021, March 18). What the Hariyo Ban Program taught us about integrating community resilience, climate adaptation, and biodiversity conservation. <u>https://www.wilsoncenter.org/event/10-years-nepal-what-hariyo-</u> ban-program-taught-us-about-integrating-community-resilience
- 53 Wilkinson, K. (Ed.). (2020). The Drawdown Review: Climate Solutions for a New Decade (p. 104). Project Drawdown. https://www.drawdown.org/publications/the-drawdown-review
- 54 Climate Policy Initiative (2021). *Global landscape of climate finance 2021*, 48. Climate Policy Initiative. <u>https://www.climatepolicyinitiative.org/publication/global-landscape-of-climate-finance-2021/</u>
- 55 Global Center on Adaptation (2021). State and Trends in Adaptation Report 2021: How Adaptation can make Africa safer, greener and more prosperous in a warming world, 290. Global Center on Adaptation. <u>https://gca.org/reports/</u> state-and-trends-in-adaptation-report-2021/
- 56 Roberts, J. T., Weikmans, R., Robinson, S., Ciplet, D., Khan, M., & Falzon, D. (2021). Rebooting a failed promise of climate finance. *Nature Climate Change*, 11(3), 180–182. <u>https://doi.org/10.1038/s41558-021-00990-2</u>
- 57 Timperley, J. (2021). The broken \$100-billion promise of climate finance—And how to fix it. *Nature*, 598(7881), 400–402. https://doi.org/10.1038/d41586-021-02846-3
- 58 Neufeldt, H., Christiansen, L., & Dale, T. W. (2021). Adaptation Gap Report 2021–The Gathering Storm: Adapting to climate change in a post-pandemic world, 104. <u>https://resiliencelinks.org/building-resilience/reports/unep-2021</u>
- 59 UNFCCC (2009). Conference of the Parties Fifteenth session: Draft decision-/CP.15 (Copenhagen Accord) (Proposal by the President FCCC/CP/2009/L.7, 5. United Nations Framework Convention on Climate Change.
- 60 de Coninck, H., Revi, A., Babiker, M., Bertoldi, P., Buckeridge, M., Cartwright, A., Dong, W., Ford, J., Fuss, S., & Hourcade, J.-C. (2018). Strengthening and implementing the global response. In *Global warming of 1.5 C: Summary for policy makers*, 313–443. IPCC-The Intergovernmental Panel on Climate Change.
- 61 USAID (2021). Catalyzing private finance for climate action: Learning Brief, 22. USAID. <u>https://www.usaid.gov/</u> documents/catalyzing-private-finance-climate-action-learning-brief
- 62 Carney, M. (2021, October 14). *Making Climate Action Count: Turning Ambition Into Reality*. Making Climate Action Count: Turning Ambition Into Reality. https://live.worldbank.org/making-climate-action-count
- 63 United Nations (n.d.). *Financing climate action.* United Nations; United Nations. <u>https://www.un.org/en/</u> <u>climatechange/raising-ambition/climate-finance</u>
- 64 United Nations Environment Programme (2022). UN-convened Net-Zero Asset Owner Alliance: Target Setting Protocol—Second edition. <u>https://www.unepfi.org/net-zero-alliance/resources/target-setting-protocol-second-edition/</u>

- 65 *Blended finance* (n.d.). Global Environment Facility. Retrieved from <u>https://www.thegef.org/what-we-do/topics/</u> <u>blended-finance</u>
- 66 IPCC (2021). Climate Change 2021: The physical science basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S. L., Péan, C., Berger, S., Caud, N., Chen, Y., Goldfarb, L., Gomis, M. I., Huang, M., Leitzell, K., Lonnoy, E., Matthews, J. B. R., Maycock, T. K., Waterfield, T., Yelekçi, O., Yu, R., & Zhou, B. (Eds.)]. Cambridge University Press. <u>https://www.ipcc.ch/report/ar6/wg1/downloads/report/IPCC_AR6_WGI_Full_Report.pdf</u>
- 67 Project Drawdown (n.d.). Project Drawdown. Retrieved from https://www.drawdown.org/
- 68 Hawken, P. (Ed.). (2017). Drawdown: The Most Comprehensive Plan Ever Proposed to Reverse Global Warming. Penguin. <u>https://www.drawdown.org/the-book</u>
- 69 Chastin, S., Jenings, N., Toney, J., Anadon, L. D., & Smith, P. (2021). Co-benefits of climate change mitigation and adaptation actions. *COP 26 Universities Network Briefing*. <u>https://www.gla.ac.uk/media/Media_814662_smxx.pdf</u>
- 70 United Nations (2015). Transforming our world: The 2030 agenda for sustainable development. (A/RES/70/1,41). United Nations. <u>https://sdgs.un.org/2030agenda</u>
- 71 Starbird, E., Norton, M., & Marcus, R. (2016). Investing in family planning: Key to achieving the Sustainable Development Goals. *Global Health: Science and Practice*, 4(2), 191–210. <u>https://doi.org/10.9745/GHSP-D-15-00374</u>
- 72 McCollum, D. L., Echeverri, L. G., Busch, S., Pachauri, S., Parkinson, S., Rogelj, J., ... & Riahi, K. (2018). Connecting the sustainable development goals by their energy inter-linkages. *Environmental Research Letters*, *13*(3), 033006. <u>https://doi.org/10.1088/1748-9326/aaafe3</u>
- 73 Mailloux, N. A., Henegan, C. P., Lsoto, D., Patterson, K. P., West, P. C., Foley, J. A., & Patz, J. A. (2021). Climate solutions double as health interventions. *International Journal of Environmental Research and Public Health*, 18(24), 13339. <u>https://doi.org/10.3390/ijerph182413339</u>
- 74 UNDP (2020). *Human Development Report 2020: The next frontier—Human development and the Anthropocene,* 412. United Nations Development Programme. Retrieved February 25, 2021, from http://hdr.undp.org/en/2020-report
- 75 World Bank (2021). Multidimensional Poverty Measure (circa 2017). World Bank. <u>https://www.worldbank.org/en/topic/</u> poverty/brief/multidimensional-poverty-measure
- 76 Raworth, K. (2017). Why it's time for Doughnut Economics. *IPPR Progressive Review, 24*(3), 216–222. <u>https://doi.org/10.1111/newe.12058</u>
- 77 FAO, IFAD, UNICEF, WFP and WHO (2021). The state of food security and nutrition in the world 2021: Transforming food systems for food security, improved nutrition and affordable healthy diets for all. Rome, FAO. <u>https://doi.org/10.4060/cb4474en</u>
- 78 United Nations (2022). SDG Indicators: 1–No Poverty. (2022). UN Stats. Retrieved from https://unstats.un.org/sdgs/ report/2019/goal-01/
- 79 FAO, IFAD, and WFP (2015). The state of food insecurity in the world 2015: Meeting the 2015 international hunger targets: Taking stock of uneven progress, 62. Rome, FAO. <u>https://www.fao.org/publications/sofi/2015/en/</u>
- 80 Fuglie, K. O., & Rada, N. (2013). *Resources, policies, and agricultural productivity in sub-Saharan Africa.* USDA-ERS Economic Research Report, 145.
- 81 Bill and Melinda Gates Foundation (2021). Investing in climate change adaptation and agricultural innovation is essential for our future, 13. <u>https://docs.gatesfoundation.org/documents/bill_and_melinda_gates_foundation_</u> investing_in_climate_change_adaptation_and_agricultural_innovation_is_essential_for_our_future.pdf
- 82 Palacios-Lopez, A., Christiaensen, L., & Kilic, T. (2017). How much of the labor in African agriculture is provided by women? *Food Policy*, 67, 52–63. <u>https://doi.org/10.1016/j.foodpol.2016.09.017</u>
- 83 Stand4Her (n.d.) Why Women's Land Rights Matter for the Future of People and Planet. Stand For Her Land Campaign (Landesa). Retrieved August 31, 2021, from https://stand4herland.org/resource/wlr-infographic/
- 84 Tseng, T. W. J., Robinson, B. E., Bellemare, M. F., BenYishay, A., Blackman, A., Boucher, T., ... & Masuda, Y. J. (2021). Influence of land tenure interventions on human well-being and environmental outcomes. *Nature Sustainability*, 4(3), 242–251. <u>https://doi.org/10.1038/s41893-020-00648-5</u>
- 85 Espinosa, D. (2019). Secure land tenure: food security depends on it. *Gates Open Res, 3*(906), 906. <u>https://gatesopenresearch.org/documents/3-906</u>
- 86 Namubiru-Mwaura, E. (2014). Land tenure and gender: Approaches and challenges for strengthening rural women's land rights (No. 92760; Women's Voice, Agency, & Participation Research Series 2014 No. 6, 36. The World Bank, Washington, D.C. <u>https://openknowledge.worldbank.org/handle/10986/21033</u>
- 87 Elias, M., Ihalainen, M., Monterroso, I., Gallant, B., & Paez Valencia, A. M. (2021). Enhancing synergies between gender equality and biodiversity, climate, and land degradation neutrality goals: Lessons from gender-responsive nature-based approaches, 36. Alliance of Biodiversity International and CIAT. Rome, Italy. <u>https://hdl.handle. net/10568/114844</u>
- 88 Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A. C., Müller, C., Arneth, A., ... & Jones, J. W. (2014). Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. *Proceedings* of the National Academy of Sciences, 111(9), 3268–3273. <u>https://doi.org/10.1073/pnas.1222463110</u>

- 89 Ray, D. K., West, P. C., Clark, M., Gerber, J. S., Prishchepov, A. V., & Chatterjee, S. (2019). Climate change has likely already affected global food production. *PLOS ONE*, *14*(5), e0217148. <u>https://doi.org/10.1371/journal.pone.0217148</u>
- 90 Ortiz-Bobea, A., Ault, T. R., Carrillo, C. M., Chambers, R. G., & Lobell, D. B. (2021). Anthropogenic climate change has slowed global agricultural productivity growth. *Nature Climate Change*, 11(4), 306–312. <u>https://doi.org/10.1038/</u> s41558-021-01000-1
- 91 Smith, P., Clark, H., Dong, H., Elsiddig, E. A., Haberl, H., Harper, R., ... & Tubiello, F. (2014). Agriculture, forestry and other land use (AFOLU). In: Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Edenhofer, O., Pichs-Madruga, R., Sokona, Y., Farahani, E., Kadner, S., Seyboth, K., Adler, A., Baum, I., Brunner, S., Eickemeier, P., Kriemann, B., Savolainen, J., Schlömer, S., von Stechow, C., Zwickel, T., and Minx, J. C. (Eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, N.Y. https://doi.org/10.1017/CB09781107415416
- 92 Ross, K., Hite, K., Waite, R., Carter, R., Pegorsch, L., Damassa, T., & Gasper, R. (2019). Enhancing NDCs: Opportunities in agriculture. WRI & OXFAM Working Paper, 36. <u>https://oxfamilibrary.openrepository.com/bitstream/</u> handle/10546/620922/dp-ndc-enhancement-opportunities-agriculture-011219-en.pdf?sequence=1&isAllowed=y
- 93 Abbas, F., Hammad, H. M., Fahad, S., Cerdà, A., Rizwan, M., Farhad, W., ... & Bakhat, H. F. (2017). Agroforestry: a sustainable environmental practice for carbon sequestration under the climate change scenarios—a review. *Environmental Science and Pollution Research*, 24(12), 11177–11191. https://doi.org/10.1007/s11356-017-8687-0
- 94 De Stefano, A., & Jacobson, M. G. (2018). Soil carbon sequestration in agroforestry systems: A meta-analysis. *Agroforestry Systems*, 92(2), 285–299. <u>https://doi.org/10.1007/s10457-017-0147-9</u>
- 95 Trimmer, J. T., Bauza, V., Byrne, D. M., Lardizabal, A., & Guest, J. S. (2017). Harmonizing goals for agricultural intensification and human health protection in sub-Saharan Africa. *Tropical Conservation Science*, 10, 1–6, 1940082917720666. https://doi.org/10.1177/1940082917720666
- 96 Toensmeier, E., Mehra, M., Frischmann, C., & Foley, J. (2020). A Drawdown Primer: Farming Our Way Out of the Climate Crisis—Changing Our Land Use, Agricultural Practices, and Food System Offers Numerous Opportunities to Reduce Greenhouse Gas Emissions, Sequester Atmospheric Carbon, and Help Address Climate Change (p. 48). <u>https://www.drawdown.org/publications/farming-our-way-out-of-the-climate-crisis</u>
- 97 Giller, K. E., Delaune, T., Silva, J. V., van Wijk, M., Hammond, J., Descheemaeker, K., ... & Andersson, J. A. (2021). Small farms and development in sub-Saharan Africa: Farming for food, for income or for lack of better options? *Food Security*, 13(6), 1431–1454. <u>https://doi.org/10.1007/s12571-021-01209-0</u>
- 98 Ray, D. K., Ramankutty, N., Mueller, N. D., West, P. C., & Foley, J. A. (2012). Recent patterns of crop yield growth and stagnation. *Nature Communications*, 3(1), 1–7. https://doi.org/10.1038/ncomms2296
- 99 Reich, J., Paul, S. S., & Snapp, S. S. (2021). Highly variable performance of sustainable intensification on smallholder farms: A systematic review. Global Food Security, 30, 100553. <u>https://doi.org/10.1016/j.gfs.2021.100553</u>
- 100 Carlson, K. M., Gerber, J. S., Mueller, N. D., Herrero, M., MacDonald, G. K., Brauman, K. A., ... & West, P. C. (2017). Greenhouse gas emissions intensity of global croplands. *Nature Climate Change*, 7(1), 63–68. <u>https://doi.org/10.1038/nclimate3158</u>
- 101 Tubiello, F. N., Rosenzweig, C., Conchedda, G., Karl, K., Gütschow, J., Xueyao, P., ... & Sandalow, D. (2021). Greenhouse gas emissions from food systems: building the evidence base. *Environmental Research Letters*, 16(6), 065007. <u>https://doi.org/10.1088/1748-9326/ac018e</u>
- 102 Satyanarayana, A., Thiyagarajan, T., & Uphoff, N. (2007). Opportunities for water saving with higher yield from the system of rice intensification. *Irrigation Science*, *25*(2), 99–115.
- 103 Gerber, J. S., Carlson, K. M., Makowski, D., Mueller, N. D., Garcia de Cortazar-Atauri, I., Havlík, P., ... & West, P. C. (2016). Spatially explicit estimates of N2O emissions from croplands suggest climate mitigation opportunities from improved fertilizer management. *Global Change Biology*, *22*(10), 3383–3394. <u>https://doi.org/10.1111/gcb.13341</u>
- 104 One Acre Fund (2021). 2020 Annual Impact: Country Report. One Acre Fund. https://oneacrefund.org/impact/ country-detail/
- 105 Galloway, J. N., Townsend, A. R., Erisman, J. W., Bekunda, M., Cai, Z., Freney, J. R., ... & Sutton, M. A. (2008). Transformation of the nitrogen cycle: Recent trends, questions, and potential solutions. *Science*, *320*(5878), 889–892. <u>https://doi.org/10.1126/science.1136674</u>
- 106 Tian, H., Lu, C., Melillo, J., Ren, W., Huang, Y., Xu, X., ... & Reilly, J. (2012). Food benefit and climate warming potential of nitrogen fertilizer uses in China. *Environmental Research Letters*, 7(4), 044020.
- 107 Yu, C., Huang, X., Chen, H., Godfray, H. C. J., Wright, J. S., Hall, J. W., ... & Taylor, J. (2019). Managing nitrogen to restore water quality in China. *Nature*, 567(7749), 516–520. <u>https://doi.org/10.1038/s41586-019-1001-1</u>
- 108 Kuyah, S., Whitney, C. W., Jonsson, M., Sileshi, G. W., Öborn, I., Muthuri, C. W., & Luedeling, E. (2019). Agroforestry delivers a win-win solution for ecosystem services in sub-Saharan Africa. A meta-analysis. Agronomy for Sustainable Development, 39(5), 1–18. <u>https://doi.org/10.1007/s13593-019-0589-8</u>
- 109 Garrity, D. P. (2004). Agroforestry and the achievement of the Millennium Development Goals. Agroforestry Systems, 61(1), 5–17. <u>https://doi.org/10.1023/B:AGFO.0000028986.37502.7c</u>

- 110 Bostedt, G., Hörnell, A., & Nyberg, G. (2016). Agroforestry extension and dietary diversity–an analysis of the importance of fruit and vegetable consumption in West Pokot, Kenya. *Food Security, 8*(1), 271–284. <u>https://doi.org/10.1007/s12571-015-0542-x</u>
- 111 Schwendenmann, L., Veldkamp, E., Moser, G., Hölscher, D., Köhler, M., Clough, Y., ... & Van Straaten, O. (2010). Effects of an experimental drought on the functioning of a cacao agroforestry system, Sulawesi, Indonesia. *Global Change Biology*, *16*(5), 1515–1530. <u>https://doi.org/10.1111/j.1365-2486.2009.02034.x</u>
- 112 de Sousa, K., van Zonneveld, M., Holmgren, M., Kindt, R., & Ordoñez, J. C. (2019). The future of coffee and cocoa agroforestry in a warmer Mesoamerica. *Scientific Reports*, 9(1), 1–9. <u>https://doi.org/10.1038/s41598-019-45491-7</u>
- 113 Garrity, D. P., Akinnifesi, F. K., Ajayi, O. C., Weldesemayat, S. G., Mowo, J. G., Kalinganire, A., ... & Bayala, J. (2010). Evergreen agriculture: A robust approach to sustainable food security in Africa. *Food Security*, 2(3), 197–214. <u>https://doi.org/10.1007/s12571-010-0070-7</u>
- 114 Thorlakson, T., & Neufeldt, H. (2012). Reducing subsistence farmers' vulnerability to climate change: Evaluating the potential contributions of agroforestry in western Kenya. Agriculture & Food Security, 1(1), 15. <u>https://doi.org/10.1186/2048-7010-1-15</u>
- 115 Mbow, C., Van Noordwijk, M., Luedeling, E., Neufeldt, H., Minang, P. A., & Kowero, G. (2014). Agroforestry solutions to address food security and climate change challenges in Africa. *Current Opinion in Environmental Sustainability*, 6, 61–67. https://doi.org/10.1016/j.cosust.2013.10.014
- 116 Food and Agriculture Organization of the United Nations. (2011). Global food losses and food waste—extent, causes and prevention. Food and Agricultural Organisation of the United Nations, 37. FAO. <u>https://www.fao.org/3/i2697e/ i2697e.pdf</u>
- 117 Sheahan, M., & Barrett, C. B. (2017). Review: Food loss and waste in Sub-Saharan Africa. Food Policy, 70, 1–12. <u>https://doi.org/10.1016/j.foodpol.2017.03.012</u>
- 118 Stathers, T., Holcroft, D., Kitinoja, L., Mvumi, B. M., English, A., Omotilewa, O., ... & Torero, M. (2020). A scoping review of interventions for crop postharvest loss reduction in sub-Saharan Africa and South Asia. *Nature Sustainability*, 3(10), 821–835. <u>https://doi.org/10.1038/s41893-020-00622-1</u>
- 119 West, P. C., Gerber, J. S., Engstrom, P. M., Mueller, N. D., Brauman, K. A., Carlson, K. M., ... & Siebert, S. (2014). Leverage points for improving global food security and the environment. *Science*, 345(6194), 325–328. <u>https://doi.org/10.1126/science.1246067</u>
- 120 Brauman, K. A., Richter, B. D., Postel, S., Malsy, M., Flörke, M., & Blum, J. D. (2016). Water depletion: An improved metric for incorporating seasonal and dry-year water scarcity into water risk assessments. *Elementa: Science of the Anthropocene*, 4. <u>https://doi.org/10.12952/journal.elementa.000083</u>
- 121 Bjornlund, H., van Rooyen, A., & Stirzaker, R. (2017). Profitability and productivity barriers and opportunities in smallscale irrigation schemes. *International Journal of Water Resources Development, 33*(5), 690–704. <u>https://doi.org/10.1</u> 080/07900627.2016.1263552
- 122 Passarelli, S., Mekonnen, D., Bryan, E., & Ringler, C. (2018). Evaluating the pathways from small-scale irrigation to dietary diversity: Evidence from Ethiopia and Tanzania. *Food Security*, 10(4), 981–997. <u>https://doi.org/10.1007/s12571-018-0812-5</u>
- 123 Chidavaenzi, F. R., Mazenda, A., & Ndlovu, N. (2021). Household survival and resilience to food insecurity through the drip irrigation scheme in dry rural areas. *Jàmbá: Journal of Disaster Risk Studies, 13(1), 8.*
- 124 Alaofè, H., Burney, J., Naylor, R., & Taren, D. (2016). Solar-powered drip irrigation impacts on crops production diversity and dietary diversity in northern Benin. *Food and Nutrition Bulletin*, 37(2), 164–175. <u>https://doi.org/10.1177/0379572116639710</u>
- 125 Burney, J., Woltering, L., Burke, M., Naylor, R., & Pasternak, D. (2010). Solar-powered drip irrigation enhances food security in the Sudano–Sahel. *Proceedings of the National Academy of Sciences*, 107(5), 1848–1853. <u>https://doi.org/10.1073/pnas.0909678107</u>
- 126 UNDP and OPHI (2021). *Global Multidimensional Poverty Index 2021: Unmasking disparities by ethnicity, caste and gender, 41.* United Nations Development Programme and Oxford Poverty and Human Development Initiative. Retrieved March 6, 2022, from https://ophi.org.uk/global-mpi-2021/
- 127 Rural population (% of total population)—India. (2018). The World Bank Data. <u>https://data.worldbank.org/indicator/</u> SP.RUR.TOTL.ZS?locations=IN
- 128 FAO (n.d.). FAO in India | India at a Glance. Food and Agriculture Organization of the United Nations. Retrieved March 6, 2022, from https://www.fao.org/india/fao-in-india/india-at-a-glance/en/
- 129 Chanana, N., Khatri-Chhetri, A., Pimpale, A., Joshi, R., Saini, S., Shirsath, P. B., ... & Aggarwal, P. K. (2020). Project completion report: Scaling up resilient agricultural practices, technologies and services in the vulnerable areas of India, 123. CGIAR Program on Climate Change, Agriculture and Food Security (CCAFS). Wageningen, the Netherlands <u>https://cgspace.cgiar.org/bitstream/handle/10568/110099/Final%20project%20report.pdf</u>
- 130 Khatri-Chhetri, A., Pande, K., Pant, A., & Sahin, S. (2017). Scaling up resilient agricultural practices, technologies and services in the vulnerable areas of India. CGIAR Research Program on Climate Change, Agriculture and Food Security (South Asia).

- 131 Vyas, S., & Khatri-Chhetri, A. (2018, April 24). Impacts of climate-smart villages: A non-zero sum game. https://ccafs.cgiar.org/news/impacts-climate-smart-villages-non-zero-sum-game
- 132 Reid, W. V., Mooney, H. A., Cropper, A., Capistrano, D., Carpenter, S. R., Chopra, K., ... & Zurek, M. B. (2005). *Ecosystems and human well-being-Synthesis: A report of the Millennium Ecosystem Assessment*. Island Press (p. 155). (2005). [UNEP]. Island Press. <u>https://www.millenniumassessment.org/documents/document.356.aspx.pdf</u>
- 133 IPBES (2019). Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services, 60. IPBES. <u>https://doi.org/10.5281/zenodo.3553579</u>
- 134 Fedele, G., Donatti, C. I., Bornacelly, I., & Hole, D. G. (2021). Nature-dependent people: Mapping human direct use of nature for basic needs across the tropics. *Global Environmental Change*, 71(102368), 9. <u>https://doi.org/10.1016/j.gloenvcha.2021.102368</u>
- 135 Soliev, I., Theesfeld, I., Abert, E., & Schramm, W. (2021). Benefit sharing and conflict transformation: Insights for and from REDD+ forest governance in sub-Saharan Africa. *Forest Policy and Economics*, 133, 102623. <u>https://doi.org/10.1016/j.forpol.2021.102623</u>
- 136 Garnett, S. T., Burgess, N. D., Fa, J. E., Fernández-Llamazares, Á., Molnár, Z., Robinson, C. J., ... & Leiper, I. (2018). A spatial overview of the global importance of Indigenous lands for conservation. *Nature Sustainability*, 1(7), 369–374. <u>https://doi.org/10.1038/s41893-018-0100-6</u>
- 137 McGahey, D., Davies, J., & Ouedraogo, R. (2014). *Pastoralism and the Green Economy—a natural nexus? Status, challenges and policy implications,* 72. IUCN and UNEP. <u>https://www.unep.org/resources/report/pastoralism-and-green-economy-natural-nexus-status-challenges-and-policy</u>
- 138 WWF, IUCN, and Wetlands International (2021). Roots of Hope: The socioeconomic value of mangroves in the western Indian Ocean region, 13. WWF, IUCN and Wetlands International. <u>https://www.mangrovealliance.org/wp-content/uploads/2021/11/SOMN-White-Paper-4Nov21.pdf</u>
- 139 Hickel, J., Dorninger, C., Wieland, H., & Suwandi, I. (2022). Imperialist appropriation in the world economy: Drain from the global South through unequal exchange, 1990–2015. *Global Environmental Change, 73*, 102467. <u>https:// doi.org/10.1016/j.gloenvcha.2022.102467</u>
- 140 Curtis, P. G., Slay, C. M., Harris, N. L., Tyukavina, A., & Hansen, M. C. (2018). Classifying drivers of global forest loss. Science, 361(6407), 1108–1111. <u>https://doi.org/10.1126/SCIENCE.AAU3445</u>
- 141 Davis, K. F., Koo, H. I., Dell'Angelo, J., D'Odorico, P., Estes, L., Kehoe, L. J., ... & Tatlhego, M. (2020). Tropical forest loss enhanced by large-scale land acquisitions. *Nature Geoscience*, *13*(7), 482–488. <u>https://doi.org/10.1038/s41561-020-0592-3</u>
- 142 Mechiche-Alami, A., Yagoubi, J., & Nicholas, K. A. (2021). Agricultural land acquisitions unlikely to address the food security needs of African countries. *World Development*, 141, 105384. <u>https://doi.org/10.1016/j.worlddev.2020.105384</u>
- 143 Cotula, L. (2012). The international political economy of the global land rush: A critical appraisal of trends, scale, geography and drivers. *The Journal of Peasant Studies, 39*(3–4), 649–680. <u>https://doi.org/10.1080/03066150.2012.6</u> 74940
- 144 Dinerstein, E., Vynne, C., Sala, E., Joshi, A. R., Fernando, S., Lovejoy, T. E., ... & Wikramanayake, E. (2019). A Global Deal For Nature: Guiding principles, milestones, and targets. *Science Advances*, 5(4), eaaw2869. <u>https://doi.org/10.1126/sciadv.aaw2869</u>
- 145 UNEP-WCMC and IUCN (2021). Protected Planet Report 2020. UNEP-WCMC, UNEP, IUCN. Cambridge UK: Gland, Switzerland. <u>http://www.unep.org/resources/protected-planet-report-2020</u>
- 146 FAO and UNEP (2020). The State of the World's Forests 2020. Forests, biodiversity and people. Rome. https://doi. org/10.4060/ca8642en
- 147 Griscom, B. W., Adams, J., Ellis, P. W., Houghton, R. A., Lomax, G., Miteva, D. A., ... & Fargione, J. (2017). *Natural climate solutions*. Proceedings of the National Academy of Sciences, 114(44), 11645–11650. <u>https://doi.org/10.1073/pnas.1710465114</u>
- 148 Carter, S., Herold, M., Avitabile, V., Bruin, S. de, Sy, V. D., Kooistra, L., & Rufino, M. C. (2017). Agriculture-driven deforestation in the tropics from 1990–2015: Emissions, trends and uncertainties. *Environmental Research Letters*, 13(1), 014002. <u>https://doi.org/10.1088/1748-9326/aa9ea4</u>
- 149 Dasgupta, P. (2021). The Economics of Biodiversity: *The Dasgupta Review* (p. 610). HM Treasury. https://www.gov.uk/government/publications/final-report-the-economics-of-biodiversity-the-dasgupta-review
- 150 Frechette, A., Ginsburg, C., Walker, W., Gorelik, S., Keene, S., Meyer, C., ... & Veit, P. (2018). *A global baseline* of carbon storage in collective lands. Rights and Resources Initiative. Washington, D.C., USA. <u>https://</u> rightsandresources.org/wp-content/uploads/2018/09/A-Global-Baseline_RRI_Sept-2018.pdf
- 151 Page, S. E., Rieley, J. O., & Banks, C. J. (2011). Global and regional importance of the tropical peatland carbon pool. Global Change Biology, 17(2), 798–818. <u>https://doi.org/10.1111/j.1365-2486.2010.02279.x</u>
- 152 Dargie, G. C., Lewis, S. L., Lawson, I. T., Mitchard, E. T. A., Page, S. E., Bocko, Y. E., & Ifo, S. A. (2017). Age, extent and carbon storage of the central Congo Basin peatland complex. *Nature*, 542(7639), 86–90. <u>https://doi.org/10.1038/</u> <u>nature21048</u>

- 153 Draper, F. C., Roucoux, K. H., Lawson, I. T., Mitchard, E. T. A., Honorio Coronado, E. N., Lähteenoja, O., ... & Baker, T. R. (2014). The distribution and amount of carbon in the largest peatland complex in Amazonia. *Environmental Research Letters*, 9(12), 124017. <u>https://doi.org/10.1088/1748-9326/9/12/124017</u>
- 154 Robinson, B. E., Holland, M. B., & Naughton-Treves, L. (2014). Does secure land tenure save forests? A metaanalysis of the relationship between land tenure and tropical deforestation. *Global Environmental Change, 29,* 281–293. <u>https://doi.org/10.1016/j.gloenvcha.2013.05.012</u>
- 155 Dawson, N. M., Coolsaet, B., Sterling, E. J., Loveridge, R., Gross-Camp, N. D., Wongbusarakum, S., ... & Rosado-May, F. (2021). The role of Indigenous peoples and local communities in effective and equitable conservation. *Ecology* and Society, 26(3), 39. https://doi.org/10.5751/ES-12625-260319
- 156 Hajjar, R., Oldekop, J. A., Cronkleton, P., Newton, P., Russell, A. J. M., & Zhou, W. (2021). A global analysis of the social and environmental outcomes of community forests. *Nature Sustainability*, 4(3), 216–224. <u>https://doi.org/10.1038/s41893-020-00633-y</u>
- 157 Agarwal, B. (2009). Gender and forest conservation: The impact of women's participation in community forest governance. *Ecological Economics*, *68*(11), 2785–2799. <u>https://doi.org/10.1016/j.ecolecon.2009.04.025</u>
- 158 Mwangi, E., Meinzen-Dick, R., & Sun, Y. (2011). Gender and Sustainable Forest Management in East Africa and Latin America. *Ecology and Society*, 16(1). https://doi.org/10.5751/ES-03873-160117
- 159 McKinnon, M. C., Cheng, S. H., Garside, R., Masuda, Y. J., & Miller, D. C. (2015). Sustainability: Map the evidence. *Nature*, 528(7581), 185–187. <u>https://doi.org/10.1038/528185a</u>
- 160 Hajjar, R., Newton, P., Ihalainen, M., Agrawal, A., Alix-Garcia, J., Castle, S. E., ... & Timko, J. A. (2021). Levers for alleviating poverty in forests. *Forest Policy and Economics*, 132, 102589. <u>https://doi.org/10.1016/j.forpol.2021.102589</u>
- 161 Oldekop, J. A., Sims, K. R., Karna, B. K., Whittingham, M. J., & Agrawal, A. (2019). Reductions in deforestation and poverty from decentralized forest management in Nepal. *Nature Sustainability, 2*(5), 421–428. <u>https://doi.org/10.1038/s41893-019-0277-3</u>
- 162 Vedeld, P., Angelsen, A., Bojö, J., Sjaastad, E., & Berg, G. K. (2007). Forest environmental incomes and the rural poor. *Forest Policy and Economics*, 9(7), 869–879. <u>https://doi.org/10.1016/j.forpol.2006.05.008</u>
- 163 Angelsen, A., Jagger, P., Babigumira, R., Belcher, B., Hogarth, N. J., Bauch, S., ... & Wunder, S. (2014). Environmental Income and Rural Livelihoods: A Global-Comparative Analysis. World Development, 64, S12–S28. <u>https://doi.org/10.1016/j.worlddev.2014.03.006</u>
- 164 Neumann, B. P., & Hirsch, E. (2000). Commercialisation of non-timber forest products: Review and analysis of research (p. 187). Center for International Forestry Research (CIFOR). <u>https://doi.org/10.17528/cifor/000723</u>
- 165 Sunderlin, W. D., Angelsen, A., Belcher, B., Burgers, P., Nasi, R., Santoso, L., & Wunder, S. (2005). Livelihoods, forests, and conservation in developing countries: An overview. *World Development*, 33(9), 1383–1402. <u>https://doi.org/10.1016/j.worlddev.2004.10.004</u>
- 166 Spalding, M., Burke, L., Wood, S. A., Ashpole, J., Hutchison, J., & Zu Ermgassen, P. (2017). Mapping the global value and distribution of coral reef tourism. *Marine Policy*, *82*, 104–113. <u>https://doi.org/10.1016/j.marpol.2017.05.014</u>
- 167 Corbera, E., Kosoy, N., & Tuna, M. M. (2007). Equity implications of marketing ecosystem services in protected areas and rural communities: Case studies from Meso-America. *Global Environmental Change*, 17(3), 365–380. <u>https://doi.org/10.1016/j.gloenvcha.2006.12.005</u>
- 168 Rasolofoson, R., Hanauer, M. M., Pappinen, A., Fisher, B., & Ricketts, T. H. (2018). Impacts of forests on children's diet in rural areas across 27 developing countries. *Science Advances*, 4(8), eaat2853. <u>https://doi.org/10.1126/sciadv.aat2853</u>
- 169 Naidoo, R., Gerkey, D., Hole, D., Pfaff, A., Ellis, A. M., Golden, C. D., ... & Fisher, B. (2019). Evaluating the impacts of protected areas on human well-being across the developing world. *Science Advances*, 5(4), eaav3006. <u>https://doi.org/10.1126/sciadv.aav3006</u>
- 170 Hatfield, R., & Davies, J. (2006). Global Review of the Economics of Pastoralism (p. 47). IUCN.
- 171 Spalding, M. (2010). World Atlas of Mangroves. Routledge. https://doi.org/10.4324/9781849776608
- 172 McNally, C. G., Uchida, E., & Gold, A. J. (2011). The effect of a protected area on the tradeoffs between short-run and long-run benefits from mangrove ecosystems. *Proceedings of the National Academy of Sciences, 108*(34), 13945–13950. <u>https://doi.org/10.1073/pnas.1101825108</u>
- 173 Alva, S., Johnson, K., Jacob, A., D'Agnes, H., Mantovani, R., & Evans, T. (2016). Marine protected areas and children's dietary diversity in the Philippines. *Population and Environment*, 37(3), 341–361. <u>https://doi.org/10.1007/s11111-015-0240-9</u>
- 174 Brauman, K. A., Daily, G. C., Duarte, T. K. E., & Mooney, H. A. (2007). The Nature and Value of Ecosystem Services: An Overview Highlighting Hydrologic Services. Annual Review of Environment and Resources, 32, 67–98. <u>https://doi.org/10.1146/annurev.energy.32.031306.102758</u>
- 175 Rodell, M., Famiglietti, J. S., Wiese, D. N., Reager, J. T., Beaudoing, H. K., Landerer, F. W., & Lo, M. (2018). Emerging trends in global freshwater availability. *Nature*, 557(7707), 651–659. <u>https://doi.org/10.1038/s41586-018-0123-1</u>
- Herrera, D., Ellis, A., Fisher, B., Golden, C. D., Johnson, K., Mulligan, M., ... & Ricketts, T. H. (2017). Upstream watershed condition predicts rural children's health across 35 developing countries. *Nature Communications, 8*(1), 811. <u>https://doi.org/10.1038/s41467-017-00775-2</u>

- 177 Plowright, R. K., Reaser, J. K., Locke, H., Woodley, S. J., Patz, J. A., Becker, D. J., ... & Tabor, G. M. (2021). Land useinduced spillover: A call to action to safeguard environmental, animal, and human health. *The Lancet Planetary Health*, 5(4), E237–245. <u>https://doi.org/10.1016/S2542-5196(21)00031-0</u>
- 178 Olson, S. H., Gangnon, R., Elguero, E., Durieux, L., Guégan, J. F., Foley, J. A., & Patz, J. A. (2009). Links between climate, malaria, and wetlands in the Amazon Basin. *Emerging Infectious Diseases*, 15(4), 659–662. PubMed. <u>https:// doi.org/10.3201/eid1504.080822</u>
- 179 Vittor, A. Y., Pan, W., Gilman, R. H., Tielsch, J., Glass, G., Shields, T., ... & Patz, J. A. (2009). Linking deforestation to malaria in the Amazon: Characterization of the breeding habitat of the principal malaria vector, Anopheles darlingi. *The American Journal of Tropical Medicine and Hygiene*, *81*(1), 5–12. PubMed.
- 180 Garg, T. (2019). Ecosystems and human health: The local benefits of forest cover in Indonesia. Journal of Environmental Economics and Management, 98, 102271. <u>https://doi.org/10.1016/j.jeem.2019.102271</u>
- 181 Lindblade, K. A., Walker, E. D., Onapa, A. W., Katungu, J., & Wilson, M. L. (2000). Land use change alters malaria transmission parameters by modifying temperature in a highland area of Uganda. *Tropical Medicine & International Health*, 5(4), 263–274.
- 182 Walsh, J. F., Molyneux, D. H., & Birley, M. H. (1993). Deforestation: Effects on vector-borne disease. *Parasitology*, 106(S1), S55–S75. Cambridge Core. <u>https://doi.org/10.1017/S0031182000086121</u>
- 183 Rohr, J. R., Civitello, D. J., Halliday, F. W., Hudson, P. J., Lafferty, K. D., Wood, C. L., & Mordecai, E. A. (2020). Towards common ground in the biodiversity–disease debate. Nature *Ecology & Evolution*, 4(1), 24–33. <u>https://doi.org/10.1038/</u> <u>s41559-019-1060-6</u>
- 184 Berry, H. L., Bowen, K., & Kjellstrom, T. (2010). Climate change and mental health: A causal pathways framework. International Journal of Public Health, 55(2), 123–132. <u>https://doi.org/10.1007/s00038-009-0112-0</u>
- 185 Sharpe, I., & Davison, C. M. (2021). Climate change, climate-related disasters and mental disorder in low- and middle-income countries: A scoping review. *BMJ Open*, *11*(10), e051908. <u>https://doi.org/10.1136/ bmjopen-2021-051908</u>
- 186 Vergunst, F., & Berry, H. L. (2021). Climate Change and Children's Mental Health: A Developmental Perspective. Clinical Psychological Science, 21677026211040788. <u>https://doi.org/10.1177/21677026211040787</u>
- 187 Beck, M. W., Losada, I. J., Menéndez, P., Reguero, B. G., Díaz-Simal, P., & Fernández, F. (2018). The global flood protection savings provided by coral reefs. *Nature Communications*, 9(1), 1–9 <u>https://doi.org/10.1038/s41467-018-04568-z</u>
- 188 Menéndez, P., Losada, I. J., Torres-Ortega, S., Narayan, S., & Beck, M. W. (2020). The Global flood protection benefits of mangroves. *Scientific Reports*, 10(1), 4404. <u>https://doi.org/10.1038/s41598-020-61136-6</u>
- 189 Masuda, Y. J., Castro, B., Aggraeni, I., Wolff, N. H., Ebi, K., Garg, T., ... & Spector, J. (2019). How are healthy, working populations affected by increasing temperatures in the tropics? Implications for climate change adaptation policies. *Global Environmental Change*, 56, 29–40. <u>https://doi.org/10.1016/j.gloenvcha.2019.03.005</u>
- 190 Suter, M. K., Miller, K. A., Anggraeni, I., Ebi, K. L., Game, E. T., Krenz, J., ... & Spector, J. T. (2019). Association between work in deforested, compared to forested, areas and human heat strain: An experimental study in a rural tropical environment. *Environmental Research Letters*, 14(8), 084012. <u>https://doi.org/10.1088/1748-9326/ab2b53</u>
- 191 Alizadeh, M. R., Abatzoglou, J. T., Adamowski, J. F., Prestemon, J. P., Chittoori, B., Akbari Asanjan, A., & Sadegh, M. (2022). Increasing heat-stress inequality in a warming climate. *Earth's Future*, *10*(2), e2021EF002488. <u>https://doi.org/10.1029/2021EF002488</u>
- 192 Brokamp, C., Strawn, J. R., Beck, A. F., & Ryan, P. (2019). Pediatric psychiatric emergency department utilization and fine particulate matter: A case-crossover study. *Environmental Health Perspectives*, 127(9), 097006. <u>https://doi.org/10.1289/EHP4815</u>
- 193 Parque Nacional da Gorongosa (2021). Parque Nacional da Gorongosa, 23. <u>https://gorongosa.org/wp-content/uploads/2022/01/2021_Gorongosa-Highlights-2.pdf</u>
- 194 World Health Organization (2021). *Household air pollution and health*. Retrieved from <u>https://www.who.int/news-room/fact-sheets/detail/household-air-pollution-and-health</u>
- 195 Health Effects Institute (2020). State of Global Air 2020: A special report on global exposure to air pollution and its health impacts, 28). Health Effects Institute. <u>https://www.stateofglobalair.org/</u>
- 196 Lelieveld, J., Pozzer, A., Pöschl, U., Fnais, M., Haines, A., & Münzel, T. (2020). Loss of life expectancy from air pollution compared to other risk factors: A worldwide perspective. *Cardiovascular Research*, *116*(11), 1910–1917. <u>https://doi.org/10.1093/cvr/cvaa025</u>
- 197 Rana, J., Islam, R. M., Khan, M. N., Aliani, R., & Oulhote, Y. (2021). Association between household air pollution and child mortality in Myanmar using a multilevel mixed-effects Poisson regression with robust variance. *Scientific Reports*, *11*(1), 1–10. <u>https://doi.org/10.1038/s41598-021-92193-0</u>
- 198 Bickton, F. M., Ndeketa, L., Sibande, G. T., Nkeramahame, J., Payesa, C., & Milanzi, E. B. (2020). Household air pollution and under-five mortality in sub-Saharan Africa: An analysis of 14 demographic and health surveys. *Environmental Health and Preventive Medicine*, 25(1), 1–11. <u>https://doi.org/10.1186/s12199-020-00902-4</u>

- 199 Energy Sector Management Assistance Program (ESMAP) (2020). The state of access to modern energy cooking services, 152. Washington, D.C. World Bank. <u>https://documents.worldbank.org/en/publication/documents-reports/</u> documentdetail
- 200 Masera, O. R., Bailis, R., Drigo, R., Ghilardi, A., & Ruiz-Mercado, I. (2015). Environmental burden of traditional bioenergy use. *Annual Review of Environment and Resources*, 40(1), 121–150. <u>https://doi.org/10.1146/annurevenviron-102014-021318</u>
- 201 Vigolo, V., Sallaku, R., & Testa, F. (2018). Drivers and barriers to clean cooking: A systematic literature review from a consumer behavior perspective. *Sustainability*, *10*(11), 4322. https://doi.org/10.3390/su10114322
- 202 Onyeneke, R. U., Nwajiuba, C. A., Munonye, J., Uwazie, U. I., Uwajumogu, N., Uwadoka, C. O., & Aligbe, J.
 O. (2019). Improved cook-stoves and environmental and health outcomes: Lessons from Cross River State, Nigeria. International Journal of Environmental Research and Public Health, 16(19), 3520. <u>https://doi.org/10.3390/</u> ijerph16193520
- 203 Jagoe, K., Rossanese, M., Charron, D., Rouse, J., Waweru, F., Waruguru, M., ... & Ipe, J. (2020). Sharing the burden: Shifts in family time use, agency and gender dynamics after introduction of new cookstoves in rural Kenya. *Energy Research & Social Science*, 64, 101413. 15. <u>https://www.sciencedirect.com/science/article/pii/ S2214629619306668?via%3Dihub</u>
- 204 Putti, V. R., Tsan, M., Mehta, S., & Kammila, S. (2015). The state of the global clean and improved cooking sector (Working Paper No. 96499). World Bank, Washington, D.C. <u>https://openknowledge.worldbank.org/handle/10986/21878</u>
- 205 Rewald, R. (2017). Energy and women and girls: Analyzing the needs, uses, and impacts of energy on women and girls in the developing world, 45. Oxfam Research Backgrounder series. <u>https://www.oxfamamerica.org/explore/</u>research-publications/energy-women-girls/
- 206 Mazorra, J., Sánchez-Jacob, E., de la Sota, C., Fernández, L., & Lumbreras, J. (2020). A comprehensive analysis of cooking solutions co-benefits at household level: Healthy lives and well-being, gender and climate change. *Science of The Total Environment, 707,* 135968. <u>https://doi.org/10.1016/j.scitotenv.2019.135968</u>
- 207 OECD (2021). Women and SDG 7—Affordable and clean energy: Ensure access to affordable, reliable, sustainable and modern energy for all. (2021). In *Gender and the Environment: Building Evidence and Policies to Achieve the* SDGs (pp. 170–185). OECD Publishing. <u>https://doi.org/10.1787/c7cbe91b-en</u>
- 208 Bailis, R., Drigo, R., Ghilardi, A., & Masera, O. (2015). The carbon footprint of traditional woodfuels. Nature Climate Change, 5(3), 266–272. <u>https://doi.org/10.1038/nclimate2491</u>
- 209 Beyene, A., Bluffstone, R. A., Gebreegzhiaber, Z., Martinsson, P., Mekonnen, A., & Vieider, F. (2015). Do improved biomass cookstoves reduce fuelwood consumption and carbon emissions? Evidence from rural Ethiopia using a randomized treatment trial with electronic monitoring (June 22, 2015). World Bank Policy Research Working Paper No. 7324, 41. https://doi.org/10.1596/1813-9450-7324
- 210 Project Drawdown (2020). *Improved clean cookstoves technical summary*. Project Drawdown. https://drawdown.org/solutions/improved-clean-cookstoves/technical-summary
- 211 Lacey, F. G., Henze, D. K., Lee, C. J., van Donkelaar, A., & Martin, R. V. (2017). Transient climate and ambient health impacts due to national solid fuel cookstove emissions. *Proceedings of the National Academy of Sciences*, 114(6), 1269–1274. <u>https://doi.org/10.1073/pnas.1612430114</u>
- 212 Jeuland, M., Fetter, T. R., Li, Y., Pattanayak, S. K., Usmani, F., Bluffstone, R. A., & ... Toman, M. (2021). Is energy the golden thread? A systematic review of the impacts of modern and traditional energy use in low- and middle-income countries. *Renewable and Sustainable Energy Reviews*, 135, 110406. <u>https://doi.org/10.1016/j.rser.2020.110406</u>
- 213 Lindgren, S. A. (2020). Clean cooking for all? A critical review of behavior, stakeholder engagement, and adoption for the global diffusion of improved cookstoves. *Energy Research & Social Science, 68*, 101539. <u>https://doi.org/10.1016/j.erss.2020.101539</u>
- 214 Gill-Wiehl, A., Ray, I., & Kammen, D. (2021). Is clean cooking affordable? A review. Renewable and Sustainable Energy Reviews, 151, 111537. <u>https://doi.org/10.1016/j.rser.2021.111537</u>
- 215 Simkovich, S. M., Williams, K. N., Pollard, S., Dowdy, D., Sinharoy, S., Clasen, T. F., ... & Checkley, W. (2019). A systematic review to evaluate the association between clean cooking technologies and time use in low- and middle-income countries. *International Journal of Environmental Research and Public Health*, 16(13), 2277. <u>https:// doi.org/10.3390/ijerph16132277</u>
- 216 Haddad, Z., Williams, K. N., Lewis, J. J., Prats, E. V., & Adair-Rohani, H. (2021). Expanding data is critical to assessing gendered impacts of household energy use. *BMJ*, 375(2273). <u>https://doi.org/10.1136/bmj.n2273</u>
- 217 Ray, I., & Smith, K. R. (2021). Towards safe drinking water and clean cooking for all. *The Lancet Global Health, 9*(3), e361–e365. <u>https://doi.org/10.1016/S2214-109X(20)30476-9</u>
- 218 McCracken, J., Smith, K. R., Stone, P., Díaz, A., Arana, B., & Schwartz, J. (2011). Intervention to lower household wood smoke exposure in Guatemala reduces ST-segment depression on electrocardiograms. *Environmental Health Perspectives*, 119(11), 1562–1568. <u>https://doi.org/10.1289/ehp.1002834</u>

- 219 Thakur, M., Nuyts, P. A. W., Boudewijns, E. A., Kim, J. F., Faber, T., Babu, G. R., ... & Been, J. V. (2018). Impact of improved cookstoves on women's and child health in low and middle income countries: A systematic review and meta-analysis. *Thorax*, 73(11), 1026–1040. https://doi.org/10.1136/thoraxjnl-2017-210952
- 220 Onakomaiya, D., Gyamfi, J., Iwelunmor, J., Opeyemi, J., Oluwasanmi, M., Obiezu-Umeh, C., ... & Olopade, C. (2019). Implementation of clean cookstove interventions and its effects on blood pressure in low-income and middleincome countries: Systematic review. *BMJ Open*, *9*(5), e026517. https://doi.org/10.1136/bmjopen-2018-026517
- 221 Schraufnagel, D. E., Balmes, J. R., De Matteis, S., Hoffman, B., Kim, W. J., Perez-Padilla, R., ... & Wuebbles, D. J. (2019). Health benefits of air pollution reduction. *Annals of the American Thoracic Society*, *16*(12), 1478–1487. <u>https://doi.org/10.1513/AnnalsATS.201907-538CME</u>
- 222 Patel, V., Burns, J. K., Dhingra, M., Tarver, L., Kohrt, B. A., & Lund, C. (2018). Income inequality and depression: A systematic review and meta-analysis of the association and a scoping review of mechanisms. *World Psychiatry*, 17(1), 76–89. <u>https://doi.org/10.1002/wps.20492</u>
- 223 Alexander, D. A., Northcross, A., Karrison, T., Morhasson-Bello, O., Wilson, N., Atalabi, O. M., ... & Olopade, C. O. (2018). Pregnancy outcomes and ethanol cook stove intervention: A randomized-controlled trial in Ibadan, Nigeria. *Environment International*, 111, 152–163. <u>https://doi.org/10.1016/j.envint.2017.11.021</u>
- 224 Smith, K. R., McCracken, J. P., Weber, M. W., Hubbard, A., Jenny, A., Thompson, L. M., ... & Bruce, N. (2011). Effect of reduction in household air pollution on childhood pneumonia in Guatemala (RESPIRE): A randomised controlled trial. *The Lancet*, 378(9804), 1717–1726. <u>https://doi.org/10.1016/S0140-6736(11)60921-5</u>
- 225 Foote, E. M., Gieraltowski, L., Ayers, T., Sadumah, I., Faith, S. H., Silk, B. J., ... & Quick, R. E. (2013). Impact of locallyproduced, ceramic cookstoves on respiratory disease in children in rural western Kenya. *The American Journal of Tropical Medicine and Hygiene*, 88(1), 132–137. <u>https://doi.org/10.4269/ajtmh.2012.12-0496</u>
- 226 Mortimer, K., Ndamala, C. B., Naunje, A. W., Malava, J., Katundu, C., Weston, W., ... & Gordon, S. B. (2017). A cleaner burning biomass-fuelled cookstove intervention to prevent pneumonia in children under 5 years old in rural Malawi (the Cooking and Pneumonia Study): A cluster randomised controlled trial. *The Lancet, 389*(10065), 167–175. <u>https://</u> doi.org/10.1016/S0140-6736(16)32507-7
- 227 Asante, K., Wylie, B., Chillrud, S., Ae-Ngibise, K., Burkart, K., Boamah-Kaali, E., ... & Jack, D. (2019). The Ghana Randomized Air Pollution and Health Study (GRAPHS): A cluster randomized trial of Liquified petroleum gas (LPG) and efficient biomass cookstoves delivered during pregnancy. *Environmental Epidemiology*, *3*, 296. <u>https://doi.org/10.1097/01.EE9.0000609200.07410.e1</u>
- 228 Ezzati, M., & Baumgartner, J. C. (2017). Household energy and health: Where next for research and practice? *The Lancet, 389*(10065), 130–132. <u>https://doi.org/10.1016/S0140-6736(16)32506-5</u>
- 229 Karekezi, S., McDade, S., Boardman, B., Kimani, J., & Lustig, N. (2012). Energy, poverty, and development. In *Global Energy Assessment Toward a Sustainable Future* (pp. 151–190). Cambridge University Press. <u>https://www.cambridge.org/core/books/abs/global-energy-assessment/energy-poverty-and-development/</u>DC1771AD93DD0A5031A07B057CA3A8C7
- 230 Bloomfield, E., & Malla, M. B. (2015). *Gender and livelihoods impacts of clean cookstoves in South Asia*, 102. Global Alliance for Clean Cookstoves. <u>https://www.gov.uk/research-for-development-outputs/gender-and-livelihoods-impacts-of-clean-cookstoves-in-south-asia-study</u>
- 231 Daly, H., & Walton, M. A. (2017). Energy access outlook 2017: From poverty to prosperity, 144. OECD/International Energy Agency. <u>https://www.iea.org/topics/energy-access</u>
- 232 iDE (2011). Annual Biogas Users Survey 2010, 110. iDE. <u>https://snv.org/assets/explore/download/annual_biogas_users_survey_bangladesh_2010.pdf</u>
- 233 Prah, R. K. D., Carrion, D., Oppong, F. B., Tawiah, T., Mujtaba, M. N., Gyaase, S., ... & Jack, D. W. (2021). Time use implication of clean cookstoves in rural settings in Ghana: A time use study. *International Journal of Environmental Research and Public Health*, 18(1), 166. <u>https://doi.org/10.3390/ijerph18010166</u>
- 234 Krishnapriya, P. P., Chandrasekaran, M., Jeuland, M., & Pattanayak, S. K. (2021). Do improved cookstoves save time and improve gender outcomes? Evidence from six developing countries. *Energy Economics*, 102, 105456. <u>https:// doi.org/10.1016/j.eneco.2021.105456</u>
- 235 Nankhuni, F. J., & Findeis, J. L. (2004). Natural resource-collection work and children's schooling in Malawi. Agricultural Economics, 31(2–3), 123–134. <u>https://doi.org/10.1111/j.1574-0862.2004.tb00251.x</u>
- 236 Levison, D., DeGraff, D. S., & Dungumaro, E. W. (2018). Implications of environmental chores for schooling: Children's time fetching water and firewood in Tanzania. The European Journal of Development Research, 30(2), 217–234. https://doi.org/10.1057/s41287-017-0079-2
- 237 Sims, K. (2021). Education, girls' education and climate change (K4D Emerging Issues Report No. 29), 33. Institute of Development Studies. <u>https://opendocs.ids.ac.uk/opendocs/handle/20.500.12413/16523?show=full</u>
- 238 Ndiritu, S. W., & Nyangena, W. (2010). Environmental goods collection and children's schooling: Evidence from Kenya. *Environment for Development*, 11(3), 531–542. <u>https://doi.org/10.1007/s10113-010-0168-7</u>

- 239 Gebru, B., & Bezu, S. (2014). Environmental resource collection: Implications for children's schooling in Tigray, northern Ethiopia. *Environment and Development Economics*, 19(2), 182–200. <u>https://doi.org/10.1017/</u> S1355770X13000454
- 240 Beyene, A. D., Mekonnen, A., & Gebreegziabher, Z. (2014). Natural resource collection and children's literacy: Empirical evidence from panel data in rural Ethiopia. Environment for Development Initiative. <u>https://www.</u> efdinitiative.org/publications/natural-resource-collection-and-childrens-literacy-empirical-evidence-panel-data-rural
- 241 Clancy, J. S., Winther, T., Matinga, M. N., & Oparaocha, S. (2012). Gender equity in access to and benefits from modern energy and improved energy technologies: World development report background paper. ETC/ENERGIA in Association Nord/Sør-Konsulentene. <u>https://research.utwente.nl/en/publications/gender-equity-in-access-to-andbenefits-from-modern-energy-and-im</u>
- 242 Waris, V. S., C. Antahal, P., & C. Antahal, P. (2014). Fuelwood scarcity, poverty and women: Some perspectives. *IOSR Journal of Humanities and Social Science*, *19*(8), 21–33. <u>https://doi.org/10.9790/0837-19822133</u>
- 243 Scheurlen, E. (2015). Time allocation to energy resource collection in rural Ethiopia: Gender-disaggregated household responses to changes in firewood availability (SSRN Scholarly Paper ID 2567748). Social Science Research Network. <u>https://papers.ssrn.com/abstract=2567748</u>
- 244 Köhlin, G., Sills, E. O., Pattanayak, S. K., & Wilfong, C. (2011). Energy, gender and development: What are the linkages? Where is the evidence? (Policy Research Working Paper No. 5800; Background Paper to the 2012 World Development Report) 75. The World Bank. <u>https://doi.org/10.1596/1813-9450-5800</u>
- 245 Madubansi, M., & Shackleton, C. M. (2007). Changes in fuelwood use and selection following electrification in the Bushbuckridge lowveld, South Africa. *Journal of Environmental Management, 83*(4), 416–426. <u>https://doi.org/10.1016/j.jenvman.2006.03.014</u>
- 246 Ward, J., & Marsh, M. (2006, June). Sexual violence against women and girls in war and its aftermath: Realities, responses, and required resources. In Symposium on Sexual Violence in Conflict and Beyond, 34). UNFPA. <u>https://</u> www.svri.org/sites/default/files/attachments/2016-01-15/CCEF504C15AB277E852571AB0071F7CE-UNFPA.pdf
- 247 Castañeda Camey, I., Sabater, L., Owren, C. & Boyer, A.E. (2020). *Gender-based violence and environment linkages: The violence of inequality.* Wen, J. (ed.). Gland, Switzerland: IUCN. <u>https://portals.iucn.org/library/sites/library/files/</u> <u>documents/2020-002-En.pdf</u>
- 248 Listo, R. (2018). Gender myths in energy poverty literature: A critical discourse analysis. *Energy Research & Social Science, 38,* 9–18. <u>https://doi.org/10.1016/j.erss.2018.01.010</u>
- 249 Bhattacharyya, S. C., & Palit, D. (2016). Mini-grid based off-grid electrification to enhance electricity access in developing countries: What policies may be required? *Energy Policy, 94*, 166–178. <u>https://doi.org/10.1016/j.enpol.2016.04.010</u>
- 250 IEA, IRENA, UNSD, World Bank (2021). Tracking SDG 7: *The Energy Progress Report.* World Bank, Washington, D.C. https://www.cleancookingalliance.org/sector-resources/resource-database/632.html
- 251 Alkire, S., Kanagaratnam, U., & Vollmer, F. (2021). Interlinkages between multidimensional poverty and electricity: A study using the Global Multidimensional Poverty Index (MPI), 44. The Rockefeller Foundation. <u>https://www.</u> <u>rockefellerfoundation.org/report/interlinkages-between-multidimensional-poverty-and-electricity-a-study-using-the-</u> <u>global-multidimensional-poverty-index/</u>
- 252 GOGLA (2018). Powering opportunity—The economic impact of off-grid solar (p. 43). <u>https://www.gogla.org/</u> powering-opportunity-2018
- 253 United Nations (2021). *The Sustainable Development Goals Report 2021*, 68. United Nations. https://unstats.un.org/sdgs/report/2021/
- 254 Cabraal, A., Ward, W. A., Bogach, V. S., & Jain, A. (2021). *Living in the light: The Bangladesh solar home systems story* (p. 144). World Bank Group. <u>https://doi.org/10.1596/35311</u>
- 255 IRENA (2020). Geothermal development in Eastern Africa: Recommendations for power and direct use, 160. (2020). IRENA. https://www.irena.org/publications/2020/Nov/Geothermal-development-in-Eastern-Africa
- 256 Omenda, P., Mangi, P., Ofwona, C., & Mwangi, M. (2020). Country update report for Kenya 2015–2019. In Proceedings World Geothermal Congress, 16.
- 257 Alova, G., Trotter, P. A., & Money, A. (2021). A machine-learning approach to predicting Africa's electricity mix based on planned power plants and their chances of success. *Nature Energy*, 6(2), 158–166. <u>https://doi.org/10.1038/</u> s41560-020-00755-9
- 258 IRENA. (2020). *Global renewables outlook: Energy transformation 2050.* /Publications/2020/Apr/Global-Renewables-Outlook-2020. Retrieved March 5, 2022, from <u>https://www.irena.org/publications/2020/Apr/Global-</u> <u>Renewables-Outlook-2020</u>
- 259 Frankfurt School–UNEP Entre/BNEF (2018). *Global trends in renewable energy investment 2018,* 86. (2018). Frankfurt School: FS-UNEP Collaborating Centre for Climate & Sustainable Energy Finance. <u>https://www.greengrowthknowledge.org/research/global-trends-renewable-energy-investment-report-2018</u>

- 260 Mountford, H., Corfee-Morlot, J., McGregor, M., Banaji, F., Bhattacharya, A., Brand, J., ... & Stern, N. (2018). Unlocking the Inclusive Growth Story of the 21st Century: Accelerating Climate Action in Urgent Times, 208. (2018). New Climate Economy.
- 261 Gray, L., Boyle, A., Francks, E., & Yu, V. (2019). The power of small-scale solar: Gender, energy poverty, and entrepreneurship in Tanzania. *Development in Practice*, 29(1), 26–39. <u>https://doi.org/10.1080/09614524.2018.1526257</u>
- 262 Rigaud, K. K., de Sherbinin, A., Jones, B., Bergmann, J., Clement, V., Ober, K., ... & Midgley, A. (2018). Groundswell: Preparing for Internal Climate Migration. World Bank. https://doi.org/10.1596/29461
- 263 Clement, V., Rigaud, K., de Sherbinin, A., Jones, B., Adamo, S., Schewe, J., ... & Shabahat, E. (2021). Groundswell Part 2: Acting on Internal Climate Migration (p. 362). World Bank. <u>https://openknowledge.worldbank.org/</u> handle/10986/36248
- 264 Pearl-Martinez, R. (2020). Global trends impacting gender equality in energy access. *IDS Bulletin, 51*(1). <u>https://doi.org/10.19088/1968-2020.103</u>
- 265 Buragohain, T. (2012). Impact of solar energy in rural development in India. International Journal of Environmental Science and Development, 3(4), 5.
- 266 Rao, N. D., Agarwal, A., & Wood, D. (2016). Impacts of small-scale electricity systems: A study of rural communities in India and Nepal, 66. World Resources Institute. <u>https://www.wri.org/research/impacts-small-scale-electricity-systems-study-rural-communities-india-and-nepal</u>
- 267 Barman, M., Mahapatra, S., Palit, D., & Chaudhury, M. K. (2017). Performance and impact evaluation of solar home lighting systems on the rural livelihood in Assam, India. *Energy for Sustainable Development, 38,* 10–20. <u>https://doi.org/10.1016/j.esd.2017.02.004</u>
- 268 Khan, I. (2020). Impacts of energy decentralization viewed through the lens of the energy cultures framework: Solar home systems in the developing economies. *Renewable and Sustainable Energy Reviews, 119,* 109576. <u>https://doi.org/10.1016/j.rser.2019.109576</u>
- 269 Rahman, S. M., & Ahmad, M. M. (2013). Solar Home System (SHS) in rural Bangladesh: Ornamentation or fact of development? *Energy Policy*, 63, 348–354. <u>https://doi.org/10.1016/j.enpol.2013.08.041</u>
- 270 Azimoh, C. L., Klintenberg, P., Wallin, F., & Karlsson, B. (2015). Illuminated but not electrified: An assessment of the impact of Solar Home System on rural households in South Africa. *Applied Energy*, *155*, 354–364. <u>https://doi.org/10.1016/j.apenergy.2015.05.120</u>
- 271 Aklin, M., Bayer, P., Harish, S. P., & Urpelainen, J. (2017). Does basic energy access generate socioeconomic benefits? A field experiment with off-grid solar power in India. *Science Advances*, *3*, 9. <u>https://doi.org/10.1126/sciadv.1602153</u>
- 272 Power for All (2019). Powering Jobs Census 2019: The energy access workforce, 48. Power for All. <u>https://www.</u>powerforall.org/resources/reports/powering-jobs-census-2019-energy-access-workforce
- 273 Wilson, C., Grubler, A., Bento, N., Healey, S., Stercke, S. D., & Zimm, C. (2020). Granular technologies to accelerate decarbonization. *Science*, 368(6486), 36–39. <u>https://doi.org/10.1126/science.aaz8060</u>
- 274 IRENA (2013). *Renewable Energy and Jobs*, 144. International Renewable Energy Agency, Abu Dhabi. https://www. irena.org/-/media/Files/IRENA/Agency/Publication/2013/rejobs.pdf
- 275 IRENA and ILO (2021). Renewable energy and jobs—Annual Review 2021 [Special Edition: Labour and Policy Perspectives]. IRENA and ILO, Abu Dhabi, Geneva. <u>https://www.irena.org/publications/2021/Oct/Renewable-Energy-and-Jobs-Annual-Review-2021</u>
- 276 Kabir, E., Kim, K. H., & Szulejko, J. E. (2017). Social impacts of solar home systems in rural areas: A case study in Bangladesh. *Energies*, *10*(10), 1615. https://doi.org/10.3390/en10101615
- 277 Lemaire, X. (2018). Solar home systems and solar lanterns in rural areas of the Global South: What impact? Wiley Interdisciplinary Reviews: Energy and Environment, 7(5), e301. https://doi.org/10.1002/wene.301
- 278 Valente, T. W., Kim, Y. M., Lettenmaier, C., Glass, W., & Dibba, Y. (1994). Radio promotion of family planning in The Gambia. International Family Planning Perspectives, 20(3), 96–100. <u>https://doi.org/10.2307/2133511</u>
- 279 Sharan, M., & Valente, T. W. (2002). Spousal communication and family planning adoption: Effects of a radio drama serial in Nepal. *International Family Planning Perspectives, 28*(1), 16. <u>https://doi.org/10.2307/3088271</u>
- 280 Habibov, N., & Zainiddinov, H. (2015). Effect of TV and radio family planning messages on the probability of modern contraception utilization in post-Soviet Central Asia. *The International Journal of Health Planning and Management*, 32(1), e17–e38. <u>https://doi.org/10.1002/hpm.2318</u>
- 281 Alhassan, N., & Madise, N. J. (2021). Demand for family planning satisfied with modern methods in urban Malawi: CHAID analysis to identify predictors and women underserved with family planning services. *Frontiers in Global Women's Health, 2,* 27. <u>https://doi.org/10.3389/fgwh.2021.652902</u>
- 282 Ndung'U, N. S. (2021). A digital financial services revolution in Kenya: The M-Pesa case study, 98. African Economic Research Consortium, Nairobi. <u>https://www.researchgate.net/publication/349548752_A_Digital_Financial_</u> <u>Services_Revolution_in_Kenya_The_M-Pesa_Case_Study</u>
- 283 Cabraal, R. A., Barnes, D. F., & Agarwal, S. G. (2005). Productive uses of energy for rural development. Annual Review of Environment and Resources, 30(1), 117–144. <u>https://doi.org/10.1146/annurev.energy.30.050504.144228</u>

- 284 Asaduzaman, M., Yunus, M., Haque, A. K. E., Azad, A. A. M., Neelormi, S., & Hossain, A. (2013). Power from the sun: An evaluation of institutional effectiveness and impact of Solar home systems in bangladesh. World Bank Group: Directions in Development: Energy and Mining, 123.
- 285 Khandker, S. R., Samad, H. A., Sadeque, Z. K. M., Asaduzzaman, M., Yunus, M., & Haque, A. K. E. (2014). Surge in Solar-Powered Homes: Experience in Off-Grid Rural Bangladesh (p. 113). World Bank Publications. <u>https://</u> <u>documents1.worldbank.org/curated/en/871301468201262369/pdf/Surge-in-solar-powered-homes-experience-inoff-grid-rural-Bangladesh.pdf</u>
- 286 Diallo, A., & Moussa, R. K. (2020). The effects of solar home system on welfare in off-grid areas: Evidence from Côte d'Ivoire. Energy, 194, 116835. <u>https://doi.org/10.1016/j.energy.2019.116835</u>
- 287 Sharma, R., Choudhary, D., Kumar, P., Venkateswaran, J., & Solanki, C. S. (2019). Do solar study lamps help children study at night? Evidence from rural India. *Energy for Sustainable Development*, 50, 109–116. <u>https://doi.org/10.1016/j.esd.2019.03.005</u>
- 288 Hassan, F., & Lucchino, P. (2016). Powering education. In CEP Discussion Papers (CEP Discussion Papers). Centre for Economic Performance, LSE. https://ideas.repec.org/p/cep/cep/cep/dp1438.html
- 289 Grimm, M., Munyehirwe, A., Peters, J., & Sievert, M. (2016). A first step up the energy ladder?: Low cost solar kits and household's welfare in Rural Rwanda (Policy Research Working Paper No. WPS7859, 80). World Bank. <u>https://doi.org/10.1596/1813-9450-7859</u>
- 290 Stojanovski, O., Thurber, M. C., Wolak, F., Muwowo, G., & Harrison, K. (2018). Assessing opportunities for solar lanterns to improve educational outcomes in rural off-grid regions: Challenges and lessons from a randomized controlled trial (SSRN Scholarly Paper ID 3289855). Social Science Research Network. <u>https://doi.org/10.2139/</u> ssrn.3289855
- 291 van Fleet, J., Watkins, K., & Greubel, L. (1 C.E., November 30). Interactive Africa learning barometer. Brookings. https://www.brookings.edu/interactives/africa-learning-barometer/
- 292 UNICEF. (2020). How many children and young people have internet access at home? Estimating digital connectivity during the COVID-19 pandemic, 17. (2020). UNICEF and International Telecommunication Union. <u>https://</u> data.unicef.org/resources/children-and-young-people-internet-access-at-home-during-covid19/
- 293 Presidential Policy and Strategy Unit (Kenya) and Population Council. (2021). Promises to keep: Impact of COVID-19 on adolescents in Kenya, 94. https://knowledgecommons.popcouncil.org/departments_sbsr-pgy/1334/
- 294 Bond, M., Fuller, R. J., & Aye, L. (2012). Sizing solar home systems for optimal development impact. *Energy Policy*, 42, 699–709. <u>https://doi.org/10.1016/j.enpol.2011.12.052</u>
- 295 Grimm, M., Peters, J., & Sievert, M. (2013). Impacts of Pico-PV systems usage using a randomized controlled trial and qualitative methods, 80. Netherlands Ministry of Foreign Affairs. <u>https://developmentevidence.3ieimpact.</u> <u>org/search-result-details/impact-evaluation-repository/impacts-of-pico-pv-systems-usage-using-a-randomizedcontrolled-trial-and-qualitative-methods/5906</u>
- 296 Marro, P., & Bertsch, N. (2015). *Making renewable energy a success in Bangladesh: Getting the business model right* (2313-5875 (e-ISSN); Working Paper No. 41; ADB South Asia Working Paper Series, 39). Asian Development Bank. https://think-asia.org/bitstream/handle/11540/9743/ban-making-renewable-energy-success.pdf?sequence=1
- 297 Lam, N. L., Muhwezi, G., Isabirye, F., Harrison, K., Ruiz-Mercado, I., Amukoye, E., ... & Bates, M. N. (2017). Exposure reductions associated with introduction of solar lamps to kerosene lamp-using households in Busia County, Kenya. *Indoor Air, 28*(2), 218–227. <u>https://doi.org/10.1111/ina.12433</u>
- 298 Mashreky, S. R., Rahman, A., Chowdhury, S. M., Giashuddin, S., Svanström, L., Linnan, M., ... & Rahman, F. (2008). Epidemiology of childhood burn: Yield of largest community based injury survey in Bangladesh. *Burns, 34*(6), 856–862. <u>https://doi.org/10.1016/j.burns.2007.09.009</u>
- 299 Oludiran, O. O., & Umebese, P. F. A. (2009). Pattern and outcome of children admitted for burns in Benin City, midwestern Nigeria. *Indian Journal of Plastic Surgery, 42*(2), 189–193. <u>https://doi.org/10.1055/s-0039-1699342</u>
- 300 Kimemia, D., Vermaak, C., Pachauri, S., & Rhodes, B. (2014). Burns, scalds and poisonings from household energy use in South Africa: Are the energy poor at greater risk? *Energy for Sustainable Development*, 18, 1–8. <u>https://doi.org/10.1016/j.esd.2013.11.011</u>
- 301 Chamania, S., Chouhan, R., Awasthi, A., Bendell, R., Marsden, N., Gibson, J., ... & Potokar, T. S. (2015). Pilot project in rural western Madhya Pradesh, India, to assess the feasibility of using LED and solar-powered lanterns to remove kerosene lamps and related hazards from homes. *Burns*, 41(3), 595–603. <u>https://doi.org/10.1016/j.burns.2014.09.001</u>
- 302 Mills, E. (2016). Identifying and reducing the health and safety impacts of fuel-based lighting. *Energy for Sustainable Development, 30,* 39–50. https://doi.org/10.1016/j.esd.2015.11.002
- 303 Ramji, A., Patnaik, S., Mani, S., & Dholakia, H. H. (2017). Powering primary healthcare through solar in India: Lessons from Chhattisgarh, 80. Council on Energy, Environment and Water (CEEW). <u>https://www.ceew.in/publications/</u> powering-primary-healthcare-through-solar-india
- 304 World Health Organization (2019). *Maternal mortality*. (2019, September 19). World Health Organization. <u>https://www.who.int/news-room/fact-sheets/detail/maternal-mortality</u>

- 305 Nyamtema, A. S., Mwakatundu, N., Dominico, S., Mohamed, H., Pemba, S., Rumanyika, R., ... & Roosmalen, J. van. (2016). Enhancing maternal and perinatal health in under-served remote areas in sub-Saharan Africa: A Tanzanian model. *PLOS ONE*, *11*(3), e0151419. <u>https://doi.org/10.1371/journal.pone.0151419</u>
- 306 Dholakia, H. H. (2018). Solar powered healthcare in developing countries. Nature Energy, 3(9), 705–707. <u>https://doi.org/10.1038/s41560-018-0205-1</u>
- 307 Jones, L. E., & Olsson, G. (2017). Solar photovoltaic and wind energy providing water. Global Challenges, 1(5), 1600022. <u>https://doi.org/10.1002/gch2.201600022</u>
- 308 Meyer, J., & von Solms, S. (2018). Solar powered water security: An enabler for rural development in Limpopo South Africa. IEEE Access, 6, 20694–20703. <u>https://doi.org/10.1109/ACCESS.2018.2805367</u>
- 309 Nair, S., Rao, R., Kumar, T., Guru Prasad, G., Kumar, M., Khadeeja Henna, P., ... & Mani, M. (2019). Design of a Do-It-Yourself (DIY) Based Solar-Powered LED Lighting System for Training and Empowering Rural Youth. In A. Chakrabarti (Ed.), Research into Design for a Connected World (pp. 451–460). Springer. <u>https://doi.org/10.1007/978-981-13-5974-3_39</u>
- 310 You, J., Staddon, C., Cook, A., Walker, J., Boulton, J., Powell, W., & Leropoulos, I. (2020). Multidimensional benefits of improved sanitation: Evaluating 'PEE POWER®' in Kisoro, Uganda. *International Journal of Environmental Research* and Public Health, 17(7), 2175. <u>https://doi.org/10.3390/ijerph17072175</u>
- 311 Mahjoubi, A., Mechlouch, R. F., Mahdhaoui, B., & Brahim, A. B. (2013). Techno-economic analysis and modelling of a remote photovoltaic water pumping system in the desert of Tunisia. *International Journal of Sustainable Energy*, 32(5), 434–448. <u>https://doi.org/10.1080/14786451.2013.831417</u>
- 312 Hossain, M. A., Hassan, M. S., Mottalib, M. A., & Hossain, M. (2015). Feasibility of solar pump for sustainable irrigation in Bangladesh. International Journal of Energy and Environmental Engineering, 6(2), 147–155. <u>https://doi.org/10.1007/</u> s40095-015-0162-4
- 313 Dadonaite, B., Ritchie, H., & Roser, M. (2018, November 1). *Diarrheal diseases*. Our World in Data. <u>https://ourworldindata.org/diarrheal-diseases</u>
- 314 World Health Organization. (2019). *Progress on drinking water, sanitation and hygiene 2000–2017*: Special Focus on Inequalities, 140. UNICEF and World Health Organization. <u>https://www.unicef.org/reports/progress-on-drinking-water-sanitation-and-hygiene-2019</u>
- 315 WaterAid (2021). *Mission-critical: Invest in water, sanitation and hygiene for a healthy and green economic recovery,* 60. (2021). WaterAid. <u>https://washmatters.wateraid.org/publications/mission-critical-invest-water-sanitation-hygiene-healthy-green-recovery</u>
- 316 Candelise, C., Saccone, D., & Vallino, E. (2021). An empirical assessment of the effects of electricity access on food security. World Development, 141, 105390. <u>https://doi.org/10.1016/j.worlddev.2021.105390</u>
- 317 Quandt, A., Salerno, J. D., Neff, J. C., Baird, T. D., Herrick, J. E., McCabe, J. T., ... & Hartter, J. (2020). Mobile phone use is associated with higher smallholder agricultural productivity in Tanzania, East Africa. *PLOS ONE, 15*(8), e0237337. <u>https://doi.org/10.1371/journal.pone.0237337</u>
- 318 Manas Puri. (2016). *How access to energy can influence food losses. A brief overview.* FAO. https://www.fao.org/documents/card/en/c/86761a85-0e35-4b89-b2ac-691be59c714a/
- 319 Abric, S., Sonou, M., Augeard, B., Onimus, F., Durlin, D., Soumaila, A., & Gadelle, F. (2011). Lessons learned in the development of smallholder private irrigation for high-value crops in West Africa (Joint Discussion Paper No. 66917, 76). World Bank. <u>https://doi.org/10.1596/27375</u>
- 320 Burney, J. A., & Naylor, R. L. (2012). Smallholder irrigation as a poverty alleviation tool in sub-Saharan Africa. World Development, 40(1), 110–123. <u>https://doi.org/10.1016/j.worlddev.2011.05.007</u>
- 321 Burney, J. A., Naylor, R. L., & Postel, S. L. (2013). The case for distributed irrigation as a development priority in sub-Saharan Africa. Proceedings of the National Academy of Sciences, 110(31), 12513–12517. <u>https://doi.org/10.1073/pnas.1203597110</u>
- 322 Kamwamba-Mtethiwa, J., Weatherhead, K., & Knox, J. (2016). Assessing performance of small-scale pumped irrigation systems in sub-Saharan Africa: Evidence from a systematic review. *Irrigation and Drainage*, 65(3), 308–318. <u>https://doi.org/10.1002/ird.1950</u>
- 323 Zhou, D. & Abdullah. (2017). The acceptance of solar water pump technology among rural farmers of northern Pakistan: A structural equation model. *Cogent Food & Agriculture*, 3(1), 1280882. <u>https://doi.org/10.1080/23311932.20</u> 17.1280882
- 324 Baye, K., Hirvonen, K., Dereje, M., & Remans, R. (2019). Energy and nutrient production in Ethiopia, 2011–2015: Implications to supporting healthy diets and food systems. *PLOS ONE, 14*(3), e0213182. <u>https://doi.org/10.1371/journal.pone.0213182</u>
- 325 Baye, K., Choufani, J., Mekonnen, D., Bryan, E., Ringler, C., Griffiths, J. K., & Davies, E. (2019). Irrigation and women's diet in Ethiopia: A longitudinal study (IFPRI Discussion Paper No. 01864, 22). International Food Policy Research Institute. <u>https://www.ifpri.org/publication/irrigation-and-womens-diet-ethiopia-longitudinal-study</u>

- 326 Domenech, L. (2015). Is reliable water access the solution to undernutrition? A review of the potential of irrigation to solve nutrition and gender gaps in Africa south of the Sahara (IFPRI Discussion Paper No. 01428). Social Science Research Network. <u>https://papers.ssrn.com/abstract=2583815</u>
- 327 Thompson, T., & Gyatso, T. (2020). Technology adoption for improving agricultural productivity in sub-Saharan Africa. *Virginia Tech—College of Agriculture and Life Sciences*, 6. <u>https://globalagriculturalproductivity.org/</u> technology-adoption-for-improving-agricultural-productivity-in-sub-saharan-africa/
- 328 Ringler, C., Mekonnen, D., Xie, H., & Uhunamure, A. M. (2020). Irrigation to transform agriculture and food systems in Africa South of the Sahara. In 2020 Annual trends and outlook report: Sustaining Africa's agrifood system transformation: The role of public policies (pp. 57–70). International Food Policy Research Institute. <u>https://doi. org/10.2499/9780896293946_06</u>
- 329 IFPRI (2017). 2017 Global Food Policy Report. International Food Policy Research Institute. <u>https://doi.org/10.2499/9780896292529</u>
- 330 Lefore, N., Giordano, M., Ringler, C., & Barron, J. (2019). Sustainable and equitable growth in farmer-led irrigation in sub-Saharan Africa: What will it take? *Water Alternatives*, *12*(1), 156–168.
- 331 Pueyo, A. (2013). The evidence of benefits for poor people of increased renewable electricity capacity: Literature review: Pro-poor electricity provision, 10. [Evidence Report]. Institute of Development Studies. <u>https://opendocs.ids.</u> <u>ac.uk/opendocs/handle/20.500.12413/2961</u>
- 332 Ulsrud, K. (2020). Access to electricity for all and the role of decentralized solar power in sub-Saharan Africa. Norsk Geografisk Tidsskrift–Norwegian Journal of Geography, 74(1), 54–63. <u>https://doi.org/10.1080/00291951.2020.1736145</u>
- 333 Aevarsdottir, A. M., Barton, N., & Bold, T. (2017). The impacts of rural electrification on labor supply, income and health: Experimental evidence with solar lamps in Tanzania (Working Paper E-89302-TZA-1, 49). International Growth Centre (IGC). https://www.theigc.org/wp-content/uploads/2018/02/Aevarsdottir-et-al-2017-final-report.pdf
- 334 Samad, H. A., Khandker, S. R., Asaduzzaman, M., & Yunusd, M. (2013). The benefits of solar home systems: An analysis from Bangladesh (Policy Research Working Paper No. WPS6724, 41). World Bank. <u>https://papers.ssrn.com/</u> abstract=2367084
- 335 House, S., Mahon, T., & Cavill, S. (2012). Menstrual hygiene matters: A resource for improving menstrual hygiene around the world, 354. WaterAid and Sanitation and Hygiene Applied Research for Equity (SHARE). <u>https://</u> washmatters.wateraid.org/publications/menstrual-hygiene-matters
- 336 Agol, D., & Harvey, P. (2018). Gender differences related to WASH in schools and educational efficiency. Water Alternative, 11(2), 284–296.
- 337 Pouramin, P., Nagabhatla, N., & Miletto, M. (2020). A systematic review of water and gender interlinkages: Assessing the intersection with health. *Frontiers in Water, 2,* 6. <u>https://doi.org/10.3389/frwa.2020.00006</u>
- 338 Sperling, G. B., & Winthrop, R. (2017). What works in girls' education: Evidence for the world's best investment. Brookings Institution Press.
- 339 Solar Sister. (2018). Solar Sister. https://solarsister.org/
- 340 UNESCO (2019). New methodology shows that 258 million children, adolescents and youth are out of school (Fact Sheet UIS/2019/ED/FS/56), 16. UNESCO. <u>http://uis.unesco.org/sites/default/files/documents/new-methodology-</u> shows-258-million-children-adolescents-and-youth-are-out-school.pdf
- 341 UNESCO (2020). Gender report: A new generation: 25 years of efforts for gender equality in education. [Global Education Monitoring Report]. Paris, UNESCO. <u>https://unesdoc.unesco.org/ark:/48223/pf0000374514</u>
- 342 Imchen, A., & Ndem, F. (2020). Addressing the learning crisis: An urgent need to better finance education for the poorest children, 15. UNICEF. <u>https://www.unicef.org/reports/addressing-learning-crisis-2020</u>
- 343 Rutstein, S., & Winter, R. (2014). The Effects of fertility behavior on child survival and child nutritional status: Evidence from the Demographic and Health Surveys, 2006 to 2012 (DHS Analytical Studies 37), 77. USAID. <u>https://dhsprogram.com/pubs/pdf/AS37/AS37.pdf</u>
- 344 Starrs, A. (2015). A Lancet commission on sexual and reproductive health and rights: Going beyond the Sustainable Development Goals. *The Lancet, 386*(9999), 1111–1112. <u>https://doi.org/10.1016/S0140-6736(15)00250-0</u>
- 345 Sully, E., Biddlecom, A., Darroch, J. E., Riley, T., Ashford, L. S., Lince-Deroche, N., ... & Murro, R. (2020). Adding It up: Investing in sexual and reproductive health 2019. Guttmacher Institute. https://doi.org/10.1363/2020.31593
- 346 O'Neill, B. C., Liddle, B., Jiang, L., Smith, K. R., Pachauri, S., Dalton, M., & Fuchs, R. (2012). Demographic change and carbon dioxide emissions. *The Lancet, 380*(9837), 157–164. <u>https://doi.org/10.1016/S0140-6736(12)60958-1</u>
- 347 Dodson, J. C., Dérer, P., Cafaro, P., & Götmark, F. (2020). Population growth and climate change: Addressing the overlooked threat multiplier. Science of The Total Environment, 748(141346), 10. <u>https://doi.org/10.1016/j. scitotenv.2020.141346</u>
- 348 Liu, D. H., & Raftery, A. E. (2020). How do education and family planning accelerate fertility decline? *Population and Development Review*, 46(3), 409–441.
- 349 Myers, S., & Frumkin, H. (Eds.). (2020). *Planetary Health: Protecting Nature to Protect Ourselves*. Island Press. https://islandpress.org/books/planetary-health

- 350 Namasivayam, A., Graves, A., Kwauk, C., & Frischmann, C. (2022). Project Drawdown's health and education solution: The cascading benefits of access to universal education and voluntary family planning. Project Drawdown. <u>https://drawdown.org/publications/drawdown-solutions-policy-brief-health-and-education-solution</u>
- 351 Gakidou, E., Cowling, K., Lozano, R., & Murray, C. J. (2010). Increased educational attainment and its effect on child mortality in 175 countries between 1970 and 2009: A systematic analysis. *The Lancet, 376*(9745), 959–974. <u>https:// doi.org/10.1016/S0140-6736(10)61257-3</u>
- 352 Davidson, P. M., McGrath, S. J., Meleis, A. I., Stern, P., DiGiacomo, M., Dharmendra, T., ... & Covan, E. K. (2011). The health of women and girls determines the health and well-being of our modern world: A white paper from the International Council on Women's Health Issues. *Health Care for Women International, 32*(10), 870–886. <u>https://doi.org/10.1080/07399332.2011.603872</u>
- 353 Husain, I., Patierno, K., Zosa-Feranil, I., & Smith, R. (2016). Fostering Economic Growth, Equity, and Resilience in sub-Saharan Africa: The Role of Family Planning. PRB, 60. <u>https://www.prb.org/economic-growth-equity-sub-saharanafrica/</u>
- 354 Viner, R. M., Hargreaves, D. S., Ward, J., Bonell, C., Mokdad, A. H., & Patton, G. (2017). The health benefits of secondary education in adolescents and young adults: An international analysis in 186 low-, middle- and highincome countries from 1990 to 2013. SSM–Population Health, 3, 162–171. <u>https://doi.org/10.1016/j.ssmph.2016.12.004</u>
- 355 Starrs, A. M., Ezeh, A. C., Barker, G., Basu, A., Bertrand, J. T., Blum, R., ... & Ashford, L. S. (2018). Accelerate progress—sexual and reproductive health and rights for all: Report of the Guttmacher–Lancet Commission. *The Lancet*, 391(10140), 2642–2692. <u>https://doi.org/10.1016/S0140-6736(18)30293-9</u>
- 356 Hardee, K., Patterson, K. P., Schenck-Fontaine, A., Hess, S., Leisher, C., Mutunga, C., ... & Honzak, C. (2018). Family planning and resilience: Associations found in a population, health, and environment (PHE) project in western Tanzania. *Population and Environment*, 40(2), 204–238. <u>https://doi.org/10.1007/s11111-018-0310-x</u>
- 357 Greene, M. E., Berger, B. O., Hakobyan, L., Stiefvater, E., & Levtov, R. G. (2019). *Getting to Equal: Men, Gender Equality, and Sexual and Reproductive Health and Rights,* 116. Promundo-US. <u>https://promundoglobal.org/wp-content/uploads/2020/02/BLS19364_PRO_Men-and-SRH-report_018.1-WEB.pdf</u>
- 358 Psaki, S. R., Chuang, E. K., Melnikas, A. J., Wilson, D. B., & Mensch, B. S. (2019). Causal effects of education on sexual and reproductive health in low and middle-income countries: A systematic review and meta-analysis. SSM– Population Health, 8, 100386. <u>https://doi.org/10.1016/j.ssmph.2019.100386</u>
- 359 Cahill, N., Sonneveldt, E., Stover, J., Weinberger, M., Williamson, J., Wei, C., ... & Alkema, L. (2018). Modern contraceptive use, unmet need, and demand satisfied among women of reproductive age who are married or in a union in the focus countries of the Family Planning 2020 initiative: A systematic analysis using the Family Planning Estimation Tool. *The Lancet*, *391*(10123), 870–882. https://doi.org/10.1016/S0140-6736(17)33104-5
- 360 O'Neill, B. C., Dalton, M., Fuchs, R., Jiang, L., Pachauri, S., & Zigova, K. (2010). Global demographic trends and future carbon emissions. *Proceedings of the National Academy of Sciences*, 107(41), 17521–17526. <u>https://doi.org/10.1073/ pnas.1004581107</u>
- 361 Bongaarts, J., & Sitruk-Ware, R. (2019). Climate change and contraception. BMJ Sexual & Reproductive Health, 45(4), 233–235. <u>https://doi.org/10.1136/bmjsrh-2019-200399</u>
- 362 Pathfinder Initiative. (n.d.). *Pathfinder Initiative*. London School of Hygiene and Tropical Medicine. Retrieved March 6, 2022, from https://www.lshtm.ac.uk/research/centres-projects-groups/pathfinder-initiative
- 363 World Health Organization. (2015). Trends in maternal mortality: 1990–2015: estimates from WHO, UNICEF, UNFPA, World Bank Group and the United Nations Population Division, 92. (2015). World Health Organization. <u>https://www.who.int/reproductivehealth/publications/monitoring/maternal-mortality-2015/en/</u>
- 364 Naik, R., & Smith, R. (2015). Impacts of family planning on nutrition and food security, 36. [Policy Report]. Futures Group, Health Policy Project. <u>http://www.healthpolicyproject.com/index.cfm?id=publications&get=publD&publd=690</u>
- 365 Cleland, J. (2010). The benefits of educating women. *The Lancet, 376*(9745), 933–934. <u>https://doi.org/10.1016/S0140-6736(10)61417-1</u>
- 366 Alsan, M. M., & Cutler, D. M. (2013). Girls' education and HIV risk: Evidence from Uganda. Journal of Health Economics, 32(5), 863–872. <u>https://doi.org/10.1016/j.jhealeco.2013.06.002</u>
- 367 Agüero, J. M., & Bharadwaj, P. (2014). Do the more educated know more about health? Evidence from schooling and HIV knowledge in Zimbabwe. *Economic Development and Cultural Change*, 62(3), 489–517. JSTOR. <u>https://doi.org/10.1086/675398</u>
- 368 Durevall, D., Lindskog, A., & George, G. (2019). Education and HIV incidence among young women in KwaZulu-Natal: An association but no evidence of a causal protective effect. *PLOS ONE*, *14*(3), e0213056. <u>https://doi.org/10.1371/journal.pone.0213056</u>
- 369 Lutz, W., Muttarak, R., & Striessnig, E. (2014). Universal education is key to enhanced climate adaptation. Science, 346(6213), 1061–1062. <u>https://doi.org/10.1126/science.1257975</u>

- 370 Montez, J. K., & Friedman, E. M. (2015). Educational attainment and adult health: Under what conditions is the association causal? Social Science & Medicine (1982), 127, 1–7. https://doi.org/10.1016/j.socscimed.2014.12.029
- 371 Bongaarts, J. (2017). The effect of contraception on fertility: Is sub-Saharan Africa different? *Demographic Research*, 37(6), 129–146. https://doi.org/10.4054/DemRes.2017.37.6
- 372 Karnani, A. (2011). Reducing poverty through employment. *Innovations: Technology, Governance, Globalization,* 6(2), 73–97. https://doi.org/10.1162/INOV_a_00071
- 373 Montenegro, C. E., & Patrinos, H. A. (2014). Comparable estimates of returns to schooling Around the World. World Bank policy research working paper (7020). <u>https://doi.org/10.1596/1813-9450-7020</u>
- 374 Brunello, G., Fort, M., Schneeweis, N. E., & Winter-Ebmer, R. (2012). The causal effect of education on health: What is the role of health behaviors? (SSRN Scholarly Paper ID 2020147; 50). Social Science Research Network. <u>https://doi.org/10.2139/ssrn.2020147</u>
- 375 Garn, J. V., Sclar, G. D., Freeman, M. C., Penakalapati, G., Alexander, K. T., Brooks, P., ... & Clasen, T. F. (2017). The impact of sanitation interventions on latrine coverage and latrine use: A systematic review and meta-analysis. *International Journal of Hygiene and Environmental Health, 220*(2, Part B), 329–340. <u>https://doi.org/10.1016/j. ijheh.2016.10.001</u>
- 376 Gould, C. F., Urpelainen, J., & Hopkins Sais, J. (2020). The role of education and attitudes in cooking fuel choice: Evidence from two states in India. *Energy for Sustainable Development*, 54, 36–50. <u>https://doi.org/10.1016/j.esd.2019.09.003</u>
- 377 Kwauk, C., & Casey, O. (2021). A new green learning agenda: Approaches to quality education for climate action. Center for Universal Education at the Brookings Institution. <u>https://www.brookings.edu/research/a-new-green-learning-agenda-approaches-to-quality-education-for-climate-action/</u>
- 378 Striessnig, E., Lutz, W., & Patt, A. (2013). Effects of educational attainment on climate risk vulnerability. *Ecology and Society*, 18(1), 14. <u>https://doi.org/10.5751/ES-05252-180116</u>
- 379 Malala Fund. (2021). A greener, fairer future: Why leaders need to invest in climate and girls' education, 42. Malala Fund. https://malala.org/newsroom/archive/malala-fund-publishes-report-on-climate-change-and-girls-education
- 380 O'Neill, B. C., Jiang, L., Kc, S., Fuchs, R., Pachauri, S., Laidlaw, E. K., ... & Ren, X. (2020). The effect of education on determinants of climate change risks. *Nature Sustainability*, 3(7), 520–528. <u>https://doi.org/10.1038/s41893-020-0512-y</u>
- 381 UNICEF. (2020). Girls' education: Gender equality in education benefits every child. UNICEF. <u>https://www.unicef.org/</u>education/girls-education
- 382 UNESCO. (2020). A new generation: 25 years of efforts for gender equality in education [Global Education Monitoring Report]. Paris, UNESCO. https://unesdoc.unesco.org/ark:/48223/pf0000374514
- 383 Rao, N., Lawson, E. T., Raditloaneng, W. N., Solomon, D., & Angula, M. N. (2019). Gendered vulnerabilities to climate change: Insights from the semi-arid regions of Africa and Asia. *Climate and Development*, 11(1), 14–26. <u>https://doi.org</u> /10.1080/17565529.2017.1372266
- 384 UN Women and African Development Bank. (2021). Green jobs for women in Africa. UN Women and African Development Bank. <u>https://africa.unwomen.org/en/digital-library/publications/2021/11/un-wowen-green-jobs-policybriefs-and-reports-en-fr</u>
- 385 Joshi, S., & Schultz, T. P. (2007). Family planning as an investment in development: Evaluation of a program's consequences in Matlab, Bangladesh (February 2007). Yale University of Economic Growth Center Discussion Paper No. 951, IZA Discussion Paper No. 2639, iHEA 2007 6th World Congress: Explorations in Health Economics Paper. <u>https://papers.csm.com/sol3/papers.cfm?abstract_id=962938</u>
- 386 Canning, D., & Schultz, T. P. (2012). The economic consequences of reproductive health and family planning. *The Lancet, 380*(9837), 165–171. <u>https://doi.org/10.1016/S0140-6736(12)60827-7</u>
- 387 Drammeh, W., Hamid, N. A., & Rohana, A. J. (2019). Determinants of household food insecurity and Its association with child malnutrition in sub-Saharan Africa: A review of the literature. *Current Research in Nutrition and Food Science Journal*, 7(3), 610–623.
- 388 Madjdian, D. S., Azupogo, F., Osendarp, S. J. M., Bras, H., & Brouwer, I. D. (2018). Socio-cultural and economic determinants and consequences of adolescent undernutrition and micronutrient deficiencies in LMICs: A systematic narrative review. Annals of the New York Academy of Sciences, 1416(1), 117–139. <u>https://doi.org/10.1111/ nyas.13670</u>
- 389 Mishra, V. K., & Retherford, R. D. (2000). Women's education can improve child nutrition in India. National Family Health Survey Bulletin, 15, 4. <u>http://scholarspace.manoa.hawaii.edu/handle/10125/3468</u>
- 390 Das, S., & Rahman, R. M. (2011). Application of ordinal logistic regression analysis in determining risk factors of child malnutrition in Bangladesh. *Nutrition Journal*, 10(1), 124. <u>https://doi.org/10.1186/1475-2891-10-124</u>
- 391 Adeladza, A. (2009). The influence of socio-economic and nutritional characteristics on child growth in Kwale District of Kenya. African Journal of Food, Agriculture, *Nutrition and Development, 9*(7), 21. <u>https://doi.org/10.4314/</u> ajfand.v9i7.47686

- 392 Owoaje, E., Onifade, O., & Desmennu, A. (2014). Family and socioeconomic risk factors for undernutrition among children aged 6 to 23 months in Ibadan, Nigeria. *The Pan African Medical Journal*, 17, 161. <u>https://doi.org/10.11604/ pamj.2014.17.161.2389</u>
- 393 Jayachandran, S. (2015). The roots of gender inequality in developing countries. *Annual Review of Economics*, 7(1), 63–88. https://doi.org/10.1146/annurev-economics-080614-115404
- 394 Kwauk, C., & Braga, A. (2017). Three platforms for girls' education in climate strategies (No. 6; Brooke Shearer Series, p. 52). Brookings. <u>https://www.brookings.edu/research/3-platforms-for-girls-education-in-climate-strategies/</u>
- 395 Newman, K., & Feldman-Jacobs, C. (2015). Family planning and human rights–What's the connection and why is it important? 8. Population Reference Bureau. <u>https://www.igwg.org/wp-content/uploads/2017/06/family-planningrights-brief.pdf</u>
- 396 World Health Organization (2021). Abortion. World Health Organization. <u>https://www.who.int/news-room/fact-sheets/</u> <u>detail/abortion</u>
- 397 PRB (2021). Abortion facts & figures, 64. PRB. <u>https://scorecard.prb.org/wp-content/uploads/2021/03/2021-safe-engage-abortion-facts-and-figures-media-guide.pdf</u>
- 398 Lee-Rife, S., Malhotra, A., Warner, A., & Glinski, A. M. (2012). What works to prevent child marriage: A review of the evidence. *Studies in Family Planning*, *43*(4), 287–303. https://doi.org/10.1111/j.1728-4465.2012.00327.x
- 399 Girls Not Brides. A theory of change on child marriage–Girls not brides. (2014, July 9). Girls Not Brides. <u>https://www.girlsnotbrides.org/articles/theory-change-child-marriage-girls-brides/</u>
- 400 Nguyen, M. C., & Wodon, Q. (2015). Global and regional trends in child marriage. *The Review of Faith & International Affairs*, 13(3), 6–11. <u>https://doi.org/10.1080/15570274.2015.1075756</u>
- 401 Wodon, Q., Male, C., Nayihouba, A., Onagoruwa, A., Savadogo, A., Yedan, A., ... & Petroni, S. (2017). *Economic impacts* of child marriage: Global synthesis report (p. 99). <u>https://documents.worldbank.org/en/publication/documents-reports/</u> documentdetail/530891498511398503/economic-impacts-of-child-marriage-global-synthesis-report
- 402 Sayi, T. S., & Sibanda, A. (2018). Correlates of child marriage in Zimbabwe. *Journal of Family Issues*, 39(8), 2366–2388. https://doi.org/10.1177/0192513X18755198
- 403 UNICEF (2018). Child marriage: Latest trends and future prospects, 8. UNICEF. <u>https://data.unicef.org/resources/</u> child-marriage-latest-trends-and-future-prospects/
- 404 Kamal, S. M. M., Hassan, C. H., Alam, G. M., & Ying, Y. (2015). Child marriage in Bangladesh: trends and determinants. *Journal of Biosocial Science*, 47(1), 120–139. <u>https://doi.org/10.1017/S0021932013000746</u>
- 405 McCleary-Sills, J., Hanmer, L., Parsons, J., & Klugman, J. (2015). Child marriage: A critical barrier to girls' schooling and gender equality in education. *The Review of Faith & International Affairs*, 13(3), 69–80. <u>https://doi.org/10.1080/15</u> 570274.2015.1075755
- 406 Paul, P. (2019). Effects of education and poverty on the prevalence of girl child marriage in India: A district–level analysis. *Children and Youth Services Review, 100,* 16–21. https://doi.org/10.1016/j.childyouth.2019.02.033
- 407 Raj, A., McDougal, L., Silverman, J. G., & Rusch, M. L. A. (2014). Cross-sectional time series analysis of associations between education and girl child marriage in Bangladesh, India, Nepal and Pakistan, 1991–2011. PLOS ONE, 9(9), e106210. <u>https://doi.org/10.1371/journal.pone.0106210</u>
- 408 Silverman, J. G., Challa, S., Boyce, S. C., Averbach, S., & Raj, A. (2020). Associations of reproductive coercion and intimate partner violence with overt and covert family planning use among married adolescent girls in Niger. *eClinicalMedicine*, *22*, 100359. <u>https://doi.org/10.1016/j.eclinm.2020.100359</u>
- 409 Tomar, S., Dehingia, N., Dey, A. K., Chandurkar, D., Raj, A., & Silverman, J. G. (2020). Associations of intimate partner violence and reproductive coercion with contraceptive use in Uttar Pradesh, India: How associations differ across contraceptive methods. *PLOS ONE*, *15*(10), e0241008. <u>https://doi.org/10.1371/journal.pone.0241008</u>
- 410 The Nature Conservancy (n.d.). The Tuungane Project: Creating healthier families, fisheries, and forests in western Tanzania. (n.d.). The Nature Conservancy. <u>https://www.nature.org/en-us/about-us/where-we-work/africa/stories-in-africa/tuungane-project/</u>
- 411 Psacharopoulos, G., & Patrinos, H. A. (2018). Returns to investment in education: A decennial review of the global literature. *Education Economics*, *26*(5), 445–458. <u>https://doi.org/10.1080/09645292.2018.1484426</u>
- 412 Grant, C. (2017). *The contribution of education to economic growth*. Institute of Development Studies. <u>https://opendocs.ids.ac.uk/opendocs/handle/20.500.12413/13117</u>
- 413 Fink, G., & Peet, E. (2014). Returns to education in low and middle-income countries: Evidence from the living standards and measurement surveys (No. 12014). Program on the Global Demography of Aging at Harvard University. https://cdn1.sph.harvard.edu/wp-content/uploads/sites/1288/2015/06/PGDA_WP_120_Fink.pdf
- 414 Hahn, R. A., & Truman, B. I. (2015). Education improves public health and promotes health equity. International Journal of Health Services, 45(4), 657–678. <u>https://doi.org/10.1177/0020731415585986</u>
- 415 Boyle, M. H., Racine, Y., Georgiades, K., Snelling, D., Hong, S., Omariba, W., ... & Rao-Melacini, P. (2006). The influence of economic development level, household wealth and maternal education on child health in the developing world. *Social Science & Medicine*, 63(8), 2242–2254. <u>https://doi.org/10.1016/j.socscimed.2006.04.034</u>

- 416 Trudell, J. P., Burnet, M. L., Ziegler, B. R., & Luginaah, I. (2021). The impact of food insecurity on mental health in Africa: A systematic review. *Social Science & Medicine, 278,* 113953. https://doi.org/10.1016/j.socscimed.2021.113953
- 417 Behere, P. B., Mansharamani, H., Behere, A. P., & Yadav, R. (2020). Suicide and self-harms in rural setting: With special reference to farmers' suicide. In S. K. Chaturvedi (Ed.), Mental Health and Illness in the Rural World (pp. 151–167). Springer. <u>https://doi.org/10.1007/978-981-10-2345-3_21</u>
- 418 Smith, J. M., Olosky, L., & Fernández, J. G. (2021). The climate-gender-conflict nexus: Amplifying women's contributions at the grassroots (p. 38). Georgetown Institute for Women, Peace and Security. <u>https://giwps.georgetown.edu/resource/the-climate-gender-conflict-nexus/</u>
- 419 Blimpo, M. P., & Cosgrove-Davies, M. (2019). Electricity access in sub-Saharan Africa: Uptake, reliability, and complementary factors for economic impact. World Bank Publications. https://doi.org/10.1596/978-1-4648-1361-0
- 420 Girardeau, H., Oberholzer, A., & Pattanayak, S. K. (2021). The enabling environment for household solar adoption: A systematic review. *World Development Perspectives, 21*, 100290. <u>https://doi.org/10.1016/j.wdp.2021.100290</u>
- 421 Mperejekumana, P., Li, H., Wu, R., Lu, J., Tursunov, O., Elshareef, H., ... & Dong, R. (2021). Determinants of household energy choice for cooking in northern Sudan: A multinomial logit estimation. *International Journal of Environmental Research and Public Health, 18*(21), 11480. https://doi.org/10.3390/ijerph182111480
- 422 Karimu, A., Mensah, J. T., & Adu, G. (2016). Who adopts LPG as the main cooking fuel and why? Empirical evidence on Ghana based on national survey. *World Development*, 85, 43–57. <u>https://doi.org/10.1016/j.worlddev.2016.05.004</u>
- 423 Paudel, U., Khatri, U., & Pant, K. P. (2018). Understanding the determinants of household cooking fuel choice in Afghanistan: A multinomial logit estimation. *Energy*, *156*, 55–62. <u>https://doi.org/10.1016/j.energy.2018.05.085</u>
- 424 Tusting, L. S., Bisanzio, D., Alabaster, G., Cameron, E., Cibulskis, R., Davies, M., ... & Bhatt, S. (2019). Mapping changes in housing in sub-Saharan Africa from 2000 to 2015. *Nature, 568*(7752), 391–394. <u>https://doi.org/10.1038/</u> <u>s41586-019-1050-5</u>
- 425 Stickney, C. (2018). Building the Business Case for Housing Microfinance: Sub-Saharan Africa, 44. Habitat for Humanity: Terwilliger Center for Innovation in Shelter. <u>https://www.habitat.org/sites/default/files/documents/The-Business-Case-for-Housing-Microfinance-SSA.pdf</u>
- 426 Pay-As-You-Go (PAYGo). (n.d.). Lighting global. Retrieved January 31, 2022, from https://www.lightingglobal.org/paygo/
- 427 Verrinder, N. (2018). Building assets, unlocking access: KWFT housing microfinance impact evaluation final report, 56. Habitat for Humanity: Terwilliger Center for Innovation in Shelter. <u>https://www.habitat.org/sites/default/files/</u> <u>documents/KWFT-housing-microfinance-impact-2018.pdf</u>
- 428 West, D. M. (2015). Digital divide: Improving internet access in the developing world through affordable services and diverse content. Benton Foundation, March 3, 2015. <u>https://www.benton.org/headlines/3-ways-provide-internet-access-developing-world</u>
- 429 Karlsson, M., Penteniari, G., & Croxson, H. (2017). Accelerating affordable smartphone ownership in emerging markets. GSMA Connected Society & Connected Women Dalberg Global Development Advisors. <u>https://www.</u> gsma.com/mobilefordevelopment/resources/accelerating-affordable-smartphone-ownership-in-emerging-markets/
- 430 Pew Research. (2019). Smartphone ownership is growing rapidly around the world, but not always equally. Pew Research Center. 2019. <u>https://www.pewresearch.org/global/2019/02/05/smartphone-ownership-is-growing-rapidly-around-the-world-but-not-always-equally/</u>
- 431 Collier, P., Elliott, V. L., Hegre, H., Hoeffler, A., Reynal-Querol, M., & Sambanis, N. (2003). Breaking the conflict trap: Civil war and development policy (A World Bank Policy Research Report No. 56793, 242). World Bank. https://doi. org/10.1596/978-0-8213-5481-0
- 432 United Nations. (2021). Climate change "Biggest threat modern humans have ever faced," world-renowned naturalist tells Security Council, calls for greater global cooperation. (2021, February 23). United Nations: Meetings Coverage and Press Releases. <u>https://www.un.org/press/en/2021/sc14445.doc.htm</u>
- 433 Burke, M. B., Miguel, E., Satyanath, S., Dykema, J. A., & Lobell, D. B. (2009). Warming increases the risk of civil war in Africa. *Proceedings of the National Academy of Sciences*, 106(49), 20670–20674. <u>https://doi.org/10.1073/ pnas.0907998106</u>
- 434 Hsiang, S. M., Burke, M., & Miguel, E. (2013). Quantifying the influence of climate on human conflict. Science, 341(6151), 1235367. <u>https://doi.org/10.1126/science.1235367</u>
- 435 Mach, K. J., Kraan, C. M., Adger, W. N., Buhaug, H., Burke, M., Fearon, J. D., ... & von Uexkull, N. (2019). Climate as a risk factor for armed conflict. *Nature*, *571*(7764), 193–197. <u>https://doi.org/10.1038/s41586-019-1300-6</u>
- 436 Breisinger, C., Ecker, O., Maystadt, J.-F., Trinh Tan, J.-F., Al-Riffai, P., Bouzar, K., ... & Abdelgadir, M. (2014). *Building resilience to conflict through food-security policies and programs: An overview* (Conference Brief No. 3, 4). International Food Policy Research Institute (IFPRI). <u>https://ebrary.ifpri.org/digital/collection/p15738coll2/id/128157</u>
- 437 Ranganathan, S., Swain, R. B., & Sumpter, D. J. (2015). The demographic transition and economic growth: Implications for development policy. *Palgrave Communications*, 1(1), 1–8. https://doi.org/10.1057/palcomms.2015.33
- 438 Urdal, H. (2012). A Clash of generations? Youth bulges and political violence (Expert Paper No. 2012/1, 607–629). United Nations Department of Economic and Social Affairs. <u>https://academic.oup.com/isq/article-lookup/doi/10.1111/j.1468-2478.2006.00416.x</u>

- 439 Hirmer, S., & Guthrie, P. (2017). The benefits of energy appliances in the off-grid energy sector based on seven offgrid initiatives in rural Uganda. *Renewable and Sustainable Energy Reviews, 79,* 924–934. <u>https://doi.org/10.1016/j.</u> rser.2017.05.152
- 440 World Bank (2018). Off-grid solar market trends report 2018, 236. Dalberg Advisors and Lighting Global: The World Bank. https://www.gogla.org/resources/2018-off-grid-solar-market-trends-report
- 441 Meinzen-Dick, R., Quisumbing, A., Doss, C., & Theis, S. (2019). Women's land rights as a pathway to poverty reduction: Framework and review of available evidence. *Agricultural Systems, 172*, 72–82. <u>https://doi.org/10.1016/j.agsy.2017.10.009</u>
- 442 Chancel, L., Piketty, T., Saez, E., & Zucman, G. (2022). *World inequality report 2022*, 236. World Inequality Lab. https://wir2022.wid.world/www-site/uploads/2022/01/WIR_2022_FullReport.pdf
- 443 UNDESA. (2020). World social report 2020: Inequality in a rapidly changing world, 216. (2020). United Nations Department of Economic and Social Affairs. <u>https://doi.org/10.18356/7f5d0efc-en</u>
- 444 Roser, M., & Ortiz-Ospina, E. (2013). *Income inequality*. Published online at OurWorldInData.org. Retrieved from https://ourworldindata.org/income-inequality
- 445 Alvarez, F. N., & El-Sayed, A. M. (2017). National income inequality and ineffective health insurance in 35 low- and middle-income countries. *Health Policy and Planning*, *32*(4), 487–492. <u>https://doi.org/10.1093/heapol/czw156</u>
- 446 Ward, J. L., & Viner, R. M. (2017). The impact of income inequality and national wealth on child and adolescent mortality in low and middle-income countries. *BMC Public Health*, *17*(1), 429. <u>https://doi.org/10.1186/s12889-017-4310-z</u>
- 447 Diffenbaugh, N. S., & Burke, M. (2019). Global warming has increased global economic inequality. Proceedings of the National Academy of Sciences, 116(20), 9808–9813. <u>https://doi.org/10.1073/pnas.1816020116</u>
- 448 Islam, S. N., & Winkel, J. (2017). Climate change and social inequality (DESA Working Paper ST/ESA/2017/DWP/152)
 32. <u>https://www.un.org/development/desa/publications/working-paper/wp152</u>
- 449 Finkel, S. E. (2002). Civic education and the mobilization of political participation in developing democracies. *The Journal of Politics, 64*(4), 994–1020. <u>https://doi.org/10.1111/1468-2508.00160</u>
- 450 Kuenzi, M. T. (2006). Nonformal education, political participation, and democracy: Findings from Senegal. *Political Behavior, 28*(1), 1–31. <u>https://doi.org/10.1007/s11109-005-9000-3</u>
- 451 Lopes, J., Benton, T., & Cleaver, E. (2009). Young people's intended civic and political participation: Does education matter? *Journal of Youth Studies*, 12(1), 1–20. https://doi.org/10.1080/13676260802191920
- 452 Mayer, A. K. (2011). Does education increase political participation? *The Journal of Politics, 73*(3), 633–645. <u>https://doi.org/10.1017/S002238161100034X</u>
- 453 Wike, R., & Carle, J. (2014). Many in emerging and developing nations disconnected from politics: Participation highest in Middle East, 49. Pew Research Center. <u>https://www.pewresearch.org/global/2014/12/18/many-in-</u> emerging-and-developing-nations-disconnected-from-politics/
- 454 Croke, K., Grossman, G., Larreguy, H. A., & Marshall, J. (2014). The effect of education on political participation in electoral authoritarian regimes: Evidence from Zimbabwe, 59.
- 455 Devarajan, S., & Do, Q.-T. (2021). *Taxation, accountability, and cash transfers: Breaking the resource curse* (Policy Research Working Paper No. 9880, 27. The World Bank. <u>https://doi.org/10.1596/1813-9450-9880</u>
- 456 UN. (n.d.). Vanessa Nakate: Climate change is about the people | United Nations. Retrieved from <u>https://www.un.org/en/climatechange/vanessa-nakate-climate-change-is-about-people</u>

ABOUT PROJECT DRAWDOWN

The World's Leading Resource for Climate Solutions

Founded in 2014, Project Drawdown is a nonprofit organization that seeks to help the world reach drawdown—the point in the future when levels of greenhouse gases in the atmosphere stop climbing and start to steadily decline, thereby stopping catastrophic climate change—as quickly, safely, and equitably as possible.

Since the 2017 publication of the New York Times bestseller *Drawdown*, the organization has emerged as a leading resource for information and insight about climate solutions. Project Drawdown supports climate action and research by conducting rigorous review and assessment of climate solutions, creating compelling communication across media outlets, and partnering with efforts to accelerate climate solutions globally. Cities, universities, corporations, philanthropies, policymakers, communities, and more turn to Project Drawdown as they look to advance effective climate action. The Project Drawdown team supports a growing constellation of efforts to move climate solutions forward globally.

In 2020, Project Drawdown released The Drawdown Review, a report building on the organization's inaugural analysis and best-selling book that considers the rapidly evolving landscape of climate solutions available today. A 501(c)(3) nonprofit organization, Project Drawdown is funded by individual and institutional donations.

ABOUT DRAWDOWN LIFT

Launched in 2021, Drawdown Lift works to deepen collective understanding of the links between climate change solutions and poverty alleviation, particularly in low- and middleincome countries in sub-Saharan Africa and South Asia. The Lift team seeks to help address both extreme poverty and climate change by collaboratively identifying, promoting, and advancing solutions designed to catalyze positive, equitable change. Though research in interconnected, overlapping areas of concern—reducing poverty and addressing climate impacts—should go together, work often exists in silos. Drawdown Lift works to break down disciplinary walls and find solutions that can address climate change and extreme poverty and lead to enhanced human well-being around the world. Drawdown Lift works alongside academic researchers, practitioners, and change makers interested in achieving evidencebased, high-impact poverty alleviation and climate solutions worldwide.

Suggested citation

Jameel, Y., Patrone, C. M., Patterson, K. P., & West, P. C. (2022). Climate–poverty connections: Opportunities for synergistic solutions at the intersection of planetary and human well-being. Project Drawdown. https://doi.org/10.55789/y2c0k2p2



Follow @ProjectDrawdown



www.drawdown.org/drawdown-lift lift@drawdown.org